

Effect of Using Domain Ontologies to Facilitate Shared Understanding and Cross-Understanding in Groups

Henry Nelson Roa Marin

Bachelor of Science (Electronic and Telecommunications Engineering)

Master of Science (Information and Communications Technology)

A thesis submitted for the degree of Doctor of Philosophy at The University of Queensland in 2016

School of Information Technology and Electrical Engineering

<u>Abstract</u>

Shared cognition constructs, such as shared understanding and cross-understanding, are important factors for task performance in groups in organizations. Shared understanding defines the extent of common understanding of a topic among group members, whereas cross-understanding defines the extent to which group members accurately understand the mental model of another group member regarding a specific topic. Although research in different fields has focused on understanding the theoretical effects of these constructs, little emphasis has been placed on improving their development. In Information Systems and related fields, shared understanding of a domain is said to be facilitated using a domain ontology – a domain ontology is a description of concepts and their interrelationships in a domain, and are generally represented in a graphical way. However, there is a lack of empirical evidence to support the benefit of domain ontologies facilitating shared understanding. Accordingly, to address this knowledge deficit regarding the mechanisms to enhance shared understanding and cross-understanding, and the lack of empirical evidence regarding the effect of domain ontologies on shared understanding, this study aims to explore the effect of domain-ontology usage on shared understanding and crossunderstanding in novice group members. Accordingly, we propose a theoretical model that describes the effect of domain-ontology usage on shared understanding and crossunderstanding of domains in groups. To empirically test our model, we focus on a domain ontology developed specifically for providing shared understanding of the compliance management domain. We then tested our proposed model through an experimental research design. Our experimental research design considered a two-group, post-test only, randomized experimental design with a sample of student participants. The results of our experiments indicate that group members who use a domain ontology develop significantly higher levels of shared understanding than those who do not use a domain ontology. Secondly, group members who use a domain ontology have significantly better communication quality in terms of discussion efficiency and communication clarity than those who do not use a domain ontology. Our study also found that group members who use a domain ontology develop a significantly greater cross-understanding than those who do not use a domain ontology. The outcomes of this study are expected to contribute to both theory and practice. From a theoretical perspective, we contribute the model of domainontology effects in facilitating shared understanding and cross-understanding, as well as the operationalization and testing of the cross-understanding construct. From a practical perspective, we contribute by improving the understanding of the role domain ontologies can play as a tool to help improve performance in groups.

Declaration by author

This thesis is composed of my original work, and contains no material previously published or written by another person except where due reference has been made in the text. I have clearly stated the contribution by others to jointly-authored works that I have included in my thesis.

I have clearly stated the contribution of others to my thesis as a whole, including statistical assistance, survey design, data analysis, significant technical procedures, professional editorial advice, and any other original research work used or reported in my thesis. The content of my thesis is the result of work I have carried out since the commencement of my research higher degree candidature and does not include a substantial part of work that has been submitted to qualify for the award of any other degree or diploma in any university or other tertiary institution. I have clearly stated which parts of my thesis, if any, have been submitted to qualify for another award.

I acknowledge that an electronic copy of my thesis must be lodged with the University Library and, subject to the policy and procedures of The University of Queensland, the thesis be made available for research and study in accordance with the Copyright Act 1968 unless a period of embargo has been approved by the Dean of the Graduate School.

I acknowledge that copyright of all material contained in my thesis resides with the copyright holder(s) of that material. Where appropriate I have obtained copyright permission from the copyright holder to reproduce material in this thesis.

Publications during candidature

Peer-reviewed Papers

- Roa, H. N., Sadiq, S., and Indulska, M. 2014. "Ontology Usefulness in Human Tasks: Seeking Evidence," *The 25th Australasian Conference on Information Systems*, Auckland University of Technology, Auckland, New Zealand: ACIS.
- Roa, H. N., Indulska, M., and Sadiq, S. 2015. "Effectiveness of Domain Ontologies to Facilitate SU and CU," Proceedings of the 36th International Conference on Information Systems, Fort Worth, Texas: ICIS.

Publications included in this thesis

Publication citation – incorporated as Chapter 3.

 Roa, H. N., Sadiq, S., and Indulska, M. 2014. "Ontology Usefulness in Human Tasks: Seeking Evidence," *The 25th Australasian Conference on Information Systems*, Auckland University of Technology, Auckland, New Zealand: ACIS.

Partially incorporated as paragraphs in Chapter 2.

Contributor	Statement of contribution
Henry N. Roa (Candidate)	Wrote and developed the paper (100%)
Shazia Sadiq	Edited the paper (50%)
Marta Indulska	Edited the paper (50%)

 Roa, H. N., Indulska, M., and Sadiq, S. 2015. "Effectiveness of Domain Ontologies to Facilitate SU and CU," Proceedings of the 36th International Conference on Information Systems, Fort Worth, Texas: ICIS.

Partially incorporated as paragraphs in Chapters 1, 2, and 3.

Contributor	Statement of contribution
Henry N. Roa (Candidate)	Wrote and developed the paper (100%)
Shazia Sadiq	Edited the paper (50%)
Marta Indulska	Edited the paper (50%)

Contributions by others to the thesis

Professor Shazia Sadiq and A/Professor Malta Indulska provided advice on conceptualization and development of the theoretical model, as well as reviewing and editing the publications and thesis prior to submission.

Statement of parts of the thesis submitted to qualify for the award of another degree

None

Acknowledgements

Being a doctoral student has been an opportunity that allowed me to get valuable academic and personal experiences. This opportunity would not have been possible without the support of some institutions, inspiring colleagues and friends. I am delighted to have the opportunity to thank them for their support, advice and friendship.

First of all, I would like to express my deepest gratitude to my supervisor Prof Shazia Sadiq for giving me the opportunity to do my research under her supervision. Without her tremendous support, I would not have been able to conduct research at The University of Queensland. I also want to extend my appreciation to her for allowing me to be part of the teaching assistant team, allowing me to gain invaluable teaching experience, and support for research abroad.

Second, I also would like to voice my sincerest appreciation to A/Prof Marta Indulska for joining my supervisor team and providing me with such valuable direction and insightful feedback for my research. Her encouragement and optimism could have not been more appreciated in difficult moments. I also thank her for giving me the opportunity to be part of her research projects as a research assistant, and for the funds to support travelling research.

Third, I was very fortunate to have met Prof Andrew Burton-Jones. His course on Information Systems had a great impact on me and motivated me to continue doing research. Thanks to him, I came to realize that I did not make a wrong decision in leaving industry and joining academia. It was also a privilege to have him on my PhD Review Committee and receive support and guidance that I required. I will be always in indebted to Prof Burton-Jones.

I also would like to extend my thanks to Prof Ron Weber for his valuable comments and his time dedicated to check my research and provide input. It was my great honor to receive feedback from such a senior scholar and someone I greatly admire.

My sincere thanks also go to Dr Ben Matthews for his valuables comments during my confirmation and mid-term seminars and for support as a member of my PhD Review Committee.

My appreciation is also to the IS Round Table in the UQ Business School, Professor Iris Vessey, Dr Dongming Xu, Dr Michael Axelsen, and PhD students for their precious

comments for my ICIS paper. Special thanks to Professor Vessey and Dr Xu for their valuable recommendations on my experimental design.

I am grateful to Emmanuel College through Dr Stewart Gill and Mrs Liza Allen for the living scholarship that gave me the chance to do my research in Australia. Their help and support were very valuable.

I also would like to thank to the UQ Graduate School and the School of Information Technology and Electrical Engineering for the Top-Up Scholarship necessary to continue my work. A special thanks to all the people who work there and offered help and support during my studies.

A warm thank you the Data Knowledge Engineering (DKE) group for the good moments and laughs sharing cultural experiences. The group has been of great source of friendship and collaboration.

Another warm thank you also goes to all my Australian, Ecuadorian in Australia, and International friends for all the valuable and happy moments that we have had; and to my old friends in Ecuador for their encouragement and support they have provided me. I am particularly indebted to my dear friend Ms. Rose Gageler for her wonderful support and critical feedback on my papers.

Lastly, I would like to especially thank my family in Ecuador, particularly my mother, Isabel for all the endless support, encouragement, sacrifice and unconditional love. I dedicate this work to them.

Finally, I gratefully acknowledge the funding sources that made my PhD work possible; the Ecuadorian Government through the Secretariat of Higher Education, Science, Technology, and Innovation (SENESCYT).

<u>Keywords</u>

Conceptual modelling, domain ontology, shared understanding, cross-understanding, communication quality

Australian and New Zealand Standard Research Classifications (ANZSRC)

ANZSRC code: 080603, Conceptual Modelling, 60%

ANZSRC code: 080608, Information Systems Development Methodologies, 20%

ANZSRC code: 080609, Information Systems Management, 20%

Fields of Research (FoR) Classification

FoR code: 0806, Information Systems, 100%

TABLE OF CONTENTS

Lis	t of	Figure	S	xiv
Lis	t of	Tables	x	viii
Lis	t of	Abbre	viations	. xx
1	I	Introdu	iction	1
	11	Mot	ivation	1
		10100		
	1.2	Aim	of Research	3
	1.3	Thes	is Structure	4
2	I	Literati	ure Review	5
	2.1	Shai	ed Cognition Constructs	5
		2.1.1	Shared Understanding	5
		2.1.2	Cross-Understanding	11
		2.1.3	Other Constructs	12
	2.2	Onto	ology Foundations	13
		2.2.1	Ontology Definition	13
		2.2.2	Systematic Search of Ontology Use and Effective Use	15
		2.2.3	Ontology Use	17
	2.3	Con	ceptual Models and Domain Understanding	21
	2.4	Cha	pter Summary	24
3	I	Resear	ch Model and Hypotheses Development	.25
	3.1	Dorr	nain-Ontology Usage Process	25
		3.1.1	Individual Ontology Interpretation	26
		3.1.2	No-Group Interaction	27
	3	3.1.3	Team Communication	27
	3.2	Мос	lel Development and Hypotheses	28
		3.2.1	Impact of Domain-Ontology Usage on Developed Shared Understanding	29
		3.2.2	Impact of Domain-Ontology Usage on Communication Quality	29
	3	3.2.3	Impact of Communication Quality on Shared Understanding Improvement and Cross-Understanding	
	I	Develo	pment	31
	3.3	Cha	oter Summary	32
4	I	Empirio	al Validation	.34

	4.1	Com	pliance Management Domain	34
	4.2	The	Compliance Management Ontology	35
	4.	2.1	Description of the Ontology	35
	4.	2.2	Quality Evaluation of CoMOn	37
	4.3	Ехре	erimental Design	38
	4.4	Part	icipants and Recruitment Process	39
	4.5	Inter	rnal Validity	40
	4.6	Exte	rnal Validity	43
	4.7	Trea	Itment Materials	44
	4.	7.1	Treatment Group Materials	44
	4.	7.2	Control Group Materials	45
	4.8	Меа	isurements	46
	4.	8.1	Dependent Variables	47
	4.	8.2	Control Variables	50
	4.	8.3	Construct Validity	52
	4.9	Proc	edures	53
	4.10	E	thical Clearance	56
	4.11	Р	ilot studies	57
	4.12	С	hapter Summary	58
5	A	nalysi	s of Results	59
	F 1	Date	- Corporing	50
	5.1	Date	a screening	59
	5.2	Relia	ability of Measures	65
	5.3	Нур	otheses Testing	65
	5.	3.1	Test of Developed Shared Understanding	66
	5.	3.2	Test of Communication Quality	67
	5.	3.3	Testing of Shared Understanding Improvement	69
	5.	3.4	Test of Cross-Understanding	72
	5.4	Con	clusion Validity	75
	5.5	Cha	oter Summary	77
6	Co	onclu	sions	78
	6.1	Over	rview of Findings	78
	6.2	Cont	tributions and Implications	80

(6.2.1	Theoretical Contributions	80
	6.2.1	1 Theory of Domain Ontology Effect on Shared Understanding and Cross-Understanding	80
	6.2.1	2 Operationalization and Empirical Testing of Cross-Understanding	
	6.2.2	Practical Contributions	82
6.3	Limit	ations	82
6.4	Futur	e Research	84
6.5	Conci	uding Remarks	
Refere	ences		87
APPEN	IDIX A.	Recruitment Materials	98
A.1	Recruit	ment Slide and Leaflet	
A.2	Appoin	tment Mail Example	
A.3	Particip	ant Information and Consent Form	100
APPEN	IDIX B. I	Ethical Clearance Documents	105
APPEN	IDIX C. I	Experimental Training materials	108
C.1	Training	g Section for Treatment Group	108
С.2	Practice	2 Questions	111
С.3	Travel L	Domain Ontology	113
С.4	Travel L	Domain Description	114
С.5	Group T	Task Section	115
APPEN	IDIX D.	Data Collection Instrument	116
D.1	Instrum	ent for the Treatment Group	116
D.2	Instrum	ent for the Control Group	148
APPEN	IDIX E. A	Analysis of Outliers	176
APPEN	NDIX F. 1	est of Assumptions	178
F.1	Normal	ity Check	178
F.2	Linearit	y Analysis	180
APPEN	IDIX G.	Mediation Analysis	182
G.1	Mediat	ion of Shared Understanding Improvement: Four Factors	182

G.2 Mediation of Shared Understanding Improvement: Two Factors	184
G.3 Mediation of Developed Cross-Understanding: Four Factors	186
G.4 Mediation of Developed Cross-Understanding: Two Factors	188

LIST OF FIGURES

Figure 2-1. Ontology Representation (Guarino et al., 2009, p. 13)	15
Figure 2-2. Number of Publications per Year Related to Ontologies in the Scopus Databa	se16
Figure 3-1. Domain Ontology Usage Process to Facilitate Shared Understanding	and Cross-
Understanding	26
Figure 3-2. Domain-Ontology Usage Effect Model (F: Theoretical factor; O: Operation	alization of
factor)	
Figure 4-1. Graphical Representation CoMOn (Adapted from Syed Abdullah et al., 2012)	a)36
Figure 4-2. Conceptual Map produced by Leximancer	
Figure 4-3. Simplified Version of CoMOn	45
Figure 4-4. Experiment Stages	54
Figure 4-5. Practicing Stage	54
Figure 4-6. Individual Task Stage	55
Figure 5-1. Domain-Ontology Usage Effect Model (F: Theoretical factor; O: Operation	alization of
factor)	66
Figure 5-2. Diagram of a Mediation Model (Field, 2013)	71
Figure 6-1. Theory of Domain Ontology Usage on Shared Understanding and Cross-Und	derstanding
	81
Figure A-1. Recruitment Slide	
Figure A-2. Appointment Mail Example	99
Figure A-3. Participant Information and Consent form 1-5	
Figure A-4. Participant Information and Consent form 2-5	
Figure A-5. Participant Information and Consent form 3-5	
Figure A-6. Participant Information and Consent form 4-5	
Figure A-7. Participant Information and Consent form 5-5	
Figure B-1. Approved Ethical Clearance	
Figure B-2. Approval of Ethical Clearance Amendment	106
Figure B-3. Approval of Ethical Clearance Amendment	107
Figure C-1. Example of Student class	
Figure C-2. Properties that describe the student class	108
Figure C-3. Relationships	
Figure C-4. Student Subclass	109
Figure C-5. Library subscription domain	110

Figure C-6. Travel Domain Ontology	113
Figure D-1. Screenshot of the Experiment Introduction	116
Figure D-2. Screenshots of the Pre-Test Questions	117
Figure D-3. Screenshot of the Conceptual Modeling Knowledge Questions	118
Figure D-4. Screenshot of the Training Section	119
Figure D-5. Screenshot of the Practice Questions Section	120
Figure D-6. Screenshot of the Practice Question 1	121
Figure D-7. Screenshot of the Practice Question 2	122
Figure D-8. Screenshot of the Practice Question 3	123
Figure D-9. Screenshot of the Practice Question 4	124
Figure D-10. Screenshot of the Practice Question 5	125
Figure D-11. Screenshot of the Practice Question 6	126
Figure D-12. Screenshot of the Practice Question 7	127
Figure D-13. Screenshot of the Practice Question 8	128
Figure D-14. Screenshot of the Individual Task Section	129
Figure D-15. Screenshot of the Compliance Management Question 1	129
Figure D-16. Screenshot of the Compliance Management Question 2	130
Figure D-17. Screenshot of the Compliance Management Question 3	130
Figure D-18. Screenshot of the Compliance Management Question 4	131
Figure D-19. Screenshot of the Compliance Management Question 5	131
Figure D-20. Screenshot of the Compliance Management Question 6	132
Figure D-21. Screenshot of the Compliance Management Question 7	132
Figure D-22. Screenshot of the Compliance Management Question 8	133
Figure D-23. Screenshot of the Compliance Management Question 9	133
Figure D-24. Screenshot of the Compliance Management Question 10	134
Figure D-25. Screenshot of the Compliance Management Question 11	134
Figure D-26. Screenshot of the Compliance Management Question 12	135
Figure D-27. Screenshot of the Compliance Management Question 13	135
Figure D-28. Screenshot of the Compliance Management Question 14	136
Figure D-29. Screenshot of the Group Task Section	136
Figure D-30. Screenshot of the Final Question Section	137
Figure D-31. Screenshot of the Compliance Management Question 1	137
Figure D-32. Screenshot of the Compliance Management Question 2	138
Figure D-33. Screenshot of the Compliance Management Question 3	138
Figure D-34. Screenshot of the Compliance Management Question 4	139

Figure D-35. Screenshot of the Compliance Management Question 5	.139
Figure D-36. Screenshot of the Compliance Management Question 6	.140
Figure D-37. Screenshot of the Compliance Management Question 7	.140
Figure D-38. Screenshot of the Compliance Management Question 8	.141
Figure D-39. Screenshot of the Compliance Management Question 9	.141
Figure D-40. Screenshot of the Compliance Management Question 10	.142
Figure D-41. Screenshot of the Compliance Management Question 11	.142
Figure D-42. Screenshot of the Compliance Management Question 12	.143
Figure D-43. Screenshot of the Compliance Management Question 13	.143
Figure D-44. Screenshot of the Compliance Management Question 14	.144
Figure D-45. Screenshot of the Communication Openness Questions	.144
Figure D-46. Screenshot of the Discussion Efficiency Questions	.145
Figure D-47. Screenshot of the Discussion Effectiveness Questions	.146
Figure D-48. Screenshot of the Communication Clarity Questions	.147
Figure D-49. Screenshot of the End of the Experiment	.147
Figure D-50. Screenshot of the Experiment Introduction	.148
Figure D-51. Screenshots of the Pre-Test Questions	.149
Figure D-52. Screenshot of the Conceptual Modeling Knowledge Questions	.150
Figure D-53. Screenshot of the Compliance Management Knowledge Questions	.151
Figure D-54. Screenshot of the Practice Questions Section	.152
Figure D-55. Screenshot of the Practice Question 1	.152
Figure D-56. Screenshot of the Practice Question 2	.153
Figure D-57. Screenshot of the Practice Question 3	.153
Figure D-58. Screenshot of the Practice Question 4	.154
Figure D-59. Screenshot of the Practice Question 5	.154
Figure D-60. Screenshot of the Practice Question 6	.155
Figure D-61. Screenshot of the Practice Question 7	.155
Figure D-62. Screenshot of the Practice Question 8	.156
Figure D-63. Screenshot of the Individual Task Section	.156
Figure D-64. Screenshot of the Compliance Management Question 1	.157
Figure D-65. Screenshot of the Compliance Management Question 2	.157
Figure D-66. Screenshot of the Compliance Management Question 3	.158
Figure D-67. Screenshot of the Compliance Management Question 4	.158
Figure D-67. Screenshot of the Compliance Management Question 4 Figure D-68. Screenshot of the Compliance Management Question 5	.158 .159

Figure D-70. Screenshot of the Compliance Management Question 7	160
Figure D-71. Screenshot of the Compliance Management Question 8	160
Figure D-72. Screenshot of the Compliance Management Question 9	161
Figure D-73. Screenshot of the Compliance Management Question 10	161
Figure D-74. Screenshot of the Compliance Management Question 11	162
Figure D-75. Screenshot of the Compliance Management Question 12	162
Figure D-76. Screenshot of the Compliance Management Question 13	163
Figure D-77. Screenshot of the Compliance Management Question 14	163
Figure D-78. Screenshot of the Group Task Section	164
Figure D-79. Screenshot of the Final Question Section	164
Figure D-80. Screenshot of the Compliance Management Question 1	165
Figure D-81. Screenshot of the Compliance Management Question 2	165
Figure D-82. Screenshot of the Compliance Management Question 3	166
Figure D-83. Screenshot of the Compliance Management Question 4	166
Figure D-84. Screenshot of the Compliance Management Question 5	167
Figure D-85. Screenshot of the Compliance Management Question 6	167
Figure D-86. Screenshot of the Compliance Management Question 7	168
Figure D-87. Screenshot of the Compliance Management Question 8	168
Figure D-88. Screenshot of the Compliance Management Question 9	169
Figure D-89. Screenshot of the Compliance Management Question 10	169
Figure D-90. Screenshot of the Compliance Management Question 11	170
Figure D-91. Screenshot of the Compliance Management Question 12	170
Figure D-92. Screenshot of the Compliance Management Question 13	171
Figure D-93. Screenshot of the Compliance Management Question 14	171
Figure D-94. Screenshot of the Communication Openness Questions	172
Figure D-95. Screenshot of the Discussion Efficiency Questions	173
Figure D-96. Screenshot of the Discussion Effectiveness Questions	174
Figure D-97. Screenshot of the Communication Clarity Questions	175
Figure D-98. Screenshot of the End of the Experiment	175
Figure E-1. Dependent Variable Boxplot	177
Figure F-1. Linearity Analysis for Dependent Variables	

LIST OF TABLES

Table 1-1. Type of Users of a Domain Ontology (Adapted from Burton-Jones & Meso, 2008)	3
Table 2-1. Use of Ontologies vs. Types of Representation	18
Table 2-2. Empirical Evidence of Effective Use of Ontologies	20
Table 2-3. Ontology Development Methodologies	21
Table 2-4. Factors Affecting Understandability of Conceptual Models	23
Table 4-1. Industry Standards for Coverage Evaluation.	38
Table 4-2. Threats to Internal Validity and their Mitigation (Neuman, 2011)	41
Table 4-3. Simplified Version of the Standard ISO 19600:2014 Compliance Management System	ns –
Guidelines	46
Table 4-4. Compliance Management Questions to Measure Share Understanding	48
Table 4-5. Additional Question to Measure Cross-Understanding Development	49
Table 4-6. Communication Openness Measurement Instrument	49
Table 4-7. Discussion Efficiency Measurement Instrument	50
Table 4-8. Task Discussion Effectiveness Measurement Instrument	50
Table 4-9. Communication Clarity Measurement Instrument	50
Table 4-10. Control Variables to Check Homogeneity of Groups	51
Table 4-11. Conceptual Modelling Knowledge Measurement Instrument	51
Table 4-12. Compliance Management Knowledge Measurement Instrument	51
Table 4-13. Threats to Construct Validity and Their Mitigation (Trochim & Donnelly, 2008)	52
Table 5-1. Descriptive Statistics for the Control Variables	61
Table 5-2. Descriptive Statistics for the Dependent Variables	63
Table 5-3. Constructs Correlations	64
Table 5-4. Reliability of Dependent Variables	65
Table 5-5. Descriptive Statistics for Developed Shared Understanding	66
Table 5-6. Independent-Samples t-test Results for Developed Shared Understanding	67
Table 5-7. Descriptive Statistics for Communication Quality Factors	68
Table 5-8. Independent-Samples t-test Results for Communication Quality Factors	68
Table 5-9. Descriptive Statistics for Shared Understanding Improvement	69
Table 5-10. Independent-Samples t-test Results for SU Improvement	70
Table 5-11. Total, Direct and Indirect Effect on Shared Understanding Improvement (Four Fact	ors)
	71

Table 5-12. Total, Direct and Indirect Effect on Shared Understanding Improvement (Two Factors)
Table 5-13. Descriptive Statistics for Developed Cross-Understanding 73
Table 5-14. Independent-Samples t-test Results for Developed Cross-Understanding
Table 5-15. Total, Direct and Indirect Effect on Developed Cross-Understanding74
Table 5-16. Total, Direct and Indirect Effect on Developed Cross-Understanding
Table 5-17. Summary of Hypothesis Testing Results
Table 5-18. Threats to Conclusion Validity and their Mitigation (Trochim & Donnelly, 2008)76
Table 6-1. Summary of Results of Thesis 78
Table F-1. Shapiro-Wilk Test Results
Table F-2. Z-scores for Skewness and Kurtosis 179

LIST OF ABBREVIATIONS

CC	Communication Clarity	
CE	Communication Effectiveness	
СО	Communication Openness	
CoMOn	Compliance Management Ontology	
CQ	Communication Quality	
CU	Cross-Understanding	
SU	Shared Understanding	
TDE	Task Discussion Effectiveness	

1 INTRODUCTION

1.1 MOTIVATION

Practitioners and researchers consider *shared understanding* (SU) and *cross-understanding* (CU) as important constructs to enhance task performance in groups in organizations (Bittner & Leimeister, 2014; Briggs, 2014; Field, 2013; Huber & Lewis, 2010; Salas, Cooke, & Rosen, 2008). These groupunderstanding constructs are determined at the group level by considering the participation of all group members as contributors (Huber & Lewis, 2010). The first construct, SU, defines the extent of common or similar comprehension or understanding of a topic among group members (Ko, Kirsch, & King, 2005). Whereas, CU defines the extent to which group members accurately understand the mental model of another group member regarding a specific topic (Huber & Lewis, 2010). SU is considered of great importance in groups because differences between group member understandings can interfere with the productivity of collaborative work (Bittner & Leimeister, 2014; Langan-Fox, Anglim, & Wilson, 2004). Similarly, CU is considered of great importance in groups because, after it is achieved, it affects group processes and outcomes by: influencing the content and efficacy of member communication, elaborating or modifying member mental models, and affecting members individual and collaborative behaviors (Huber & Lewis, 2010). In sum, SU and CU are important enablers of a successful task performance in groups.

Given the importance of SU and CU, there is still a lack of studies that explore mechanisms of improvement for these group-understanding constructs (Bittner & Leimeister, 2014; Huber & Lewis, 2010). Most studies on SU and the few existing studies on CU are focused on defining and describing the effect of these constructs on groups, rather than providing alternatives to enhance these constructs – e.g. see the studies performed by Bierhals, Schuster, Kohler, and Badke-Schaub (2007), Cannon-Bowers and Salas (2001), Randolph-Seng and Norris (2011) and Wu and Keysar (2007). In brief, studies of the effectiveness of the existing mechanisms as well as the design of new mechanisms or artefacts to support and enhance SU and CU are still required.

Psychology studies indicate that group understanding constructs, such as SU and CU, can be achieved and enhanced through two different, but complementary approaches (Mathieu, Heffner, Goodwin, Salas, & Cannon-Bowers, 2000; Resnick, 1991; Stein, Wanstreet, & Glazer, 2011). The first approach requires the provision of a similar source of information to each group member about a topic in question (Resnick, 1991). The second approach requires that group members communicate with each other to exchange valuable information about the topic in quesion (Mathieu et al., 2000; Stein et al., 2011). Independent of the approach, when the information (provided or exchanged) allows for group members to develop a similar understanding of a particular topic, SU is enhanced. Similarly, when the information (provided or exchanged) allows group members to understand what the other group members know about a topic, CU is enhanced. Concisely, SU and CU can be improved by providing appropriate mechanisms to share information among group members and by an appropriate communication between group members. We consider that one of those mechanisms might be domain ontologies. A domain ontology can be explained as "a description of concepts and their interrelationships in a particular domain" (Bera, Burton-Jones, & Wand, 2011, p. 884), which are generally represented in a graphical way (Bera et al., 2011). These domain ontologies are sources of information of a particular domain, hence domain ontologies should facilitate SU.

Our literature review shows that a domain ontology can be used for the support of computer systems, as well as for the support of human tasks, such as: to benchmark and explore the quality and representation capability of conceptual modelling grammars, to improve conceptual modelling tasks, to provide individual understanding of a particular domain, to facilitate the construction of domain ontologies, and to provide SU among users. In Information Systems (IS) and related fields, the latter – the facilitation of SU of the domain represented in this ontology – is one of the oft-cited benefits of the use of a domain ontology to support human tasks (Al-Debei & Fitzgerald, 2010; Innab, Kayed, & Sajeev, 2012; Lin, Harding, & Shahbaz, 2004; Uschold, King, Moralee, & Zorgios, 1998). However, our literature review also shows that despite the popularity and high number of existing domain ontologies, this phenomenon is not well described in literature and even less commonly explored empirically. Moreover, our literature review also indicates that current ontology development methodologies lack recommendations for evaluating effectiveness of ontologies for the purpose for which they were developed, such as, for the facilitation of SU. It is therefore important to explore whether the claimed effectiveness of domain ontologies in regards to facilitating SU is evident in practice.

First, it is important to note that SU and CU are the result of the individual understanding that each group member has. This individual understanding is affected by prior knowledge of the domain and of conceptual modeling (Burton-Jones & Meso, 2008). The combination and extent of this previous knowledge in users might vary significantly, and will therefore distinguish different types of users, namely those ones who are knowledgeable in (a) conceptual modeling, (b) the domain, (c) both or (d) neither (Burton-Jones & Meso, 2008). Table 1-1 shows the different types of users that can be involved with the use of a domain ontology.

Because varying levels of knowledge will affect an individual's understanding, and hence SU and CU, we focus our efforts on a particular group of users. Our selected users range from those who have no knowledge to low knowledge levels in both domain knowledge and conceptual modeling i.e. novice practitioners (Group A in Table 1-1). We selected this group of users for two reasons. First, the purpose of this study is to understand the effect of a domain ontology on SU and CU. This enhancement of understanding at the group level is more important and significant in novice users than expert users. Second, most practitioners do not possess knowledge or experience in conceptual modeling. However, like other studies in conceptual modeling, we recognize that perhaps the most interesting cases are those in which individuals have varying levels of knowledge, however such cases are also complex and we believe that it is first necessary to understand a context involving novices prior to introducing the effect of more complex knowledge (Burton-Jones & Meso, 2008).

Table 1-1. Type of Users of a Domain Ontology (Adapted from Burton-Jones & Meso, 2008)

		Domain Knowledge		
		None - Low	Medium -High	
Conceptual	None - Low	A. Novice practitioners	B. Senior management practitioners	
Modelling	Medium -High	C. Novice practitioners with experience	D. Senior practitioners with formation in	
Knowledge		in IT or IS	IT or IS	

Finally, it is important to consider that group size might play an important role in our study. However, we consider groups of only two people. We recognize that it would be desirable to use larger groups, however such cases are also very complex and much more difficult to study. This difficulty is mainly down to two reasons. First, there is not an ideal number of members in a group. The best number of people in a group is defined by the task to be performed (Lim & Klein, 2006). And second, a higher number of participants affects the performance in groups (Mueller, 2012). Thus, it will be difficult to study our construct in such settings. Consequently, to make our goal achievable, we have deemed it necessary to define boundaries regarding group size and the type of members in our selected groups.

1.2 AIM OF RESEARCH

While many domain ontologies have been developed to facilitate the development of SU, there is a lack of empirical evidence to support such claims. Accordingly, to address this gap in the body of knowledge in the IS field, the aim of this research is to theoretically and empirically understand the effect of a domain ontology on SU and CU in novice groups. Specifically, our main research question in this study is *what is the effect of using a domain ontology on shared understanding and cross-understanding*?

To investigate our research question, we developed a theoretical model as a basis for the empirical testing of the impact of domain ontology use on SU and CU. The theoretical model is based on the

usage process of a domain ontology to facilitate SU and CU. This process uses theories of conceptual modelling (Burton-Jones, Wand, & Weber, 2009; Gemino, 2004) to describe the use of ontologies at the individual level, and theories of shared cognition to extend the use of ontologies at the group level (Cannon-Bowers & Salas, 2001; Resnick, 1991; Salas et al., 2008). Accordingly, the theoretical model and its operationalization theorizes the influence that a domain ontology has on SU and CU. Finally, through a laboratory experiment, we evaluate our operationalized model to explore the effect of a domain ontology in facilitating SU and CU. To do this, our study focuses on one instance of a domain ontology – the Compliance Management Ontology (CoMOn) (Syed Abdullah, Sadiq, & Indulska, 2012b) chosen specifically due to its relevance for the facilitation of shared understanding.

1.3 THESIS STRUCTURE

This thesis is composed of eight chapters, as follows.

Chapter 1 provides an overview of and the motivation for performing this study. This chapter presents the aim of the research, our main research question, and the research approach to perform this study.

Chapter 2 defines the main constructs involved in this research and provides a review of related research, exposing definitions, current research into domain ontologies and the group understanding constructs of *shared understanding* and *cross-understanding*, and existing gaps in literature.

Chapter 3 describes and operationalizes the theoretical model that we propose to explain the effects of a *domain ontology* on *shared understanding* and *cross-understanding*. Based on the theoretical model, we propose a set of six hypotheses, which are tested in the next chapter.

Chapter 4 presents the method used for the empirical validation of the research model. We proposed an experimental design to test our hypotheses. Specifically, details of the experiment design, participants, measures, procedures, and data processing are described in this chapter. Moreover, we present the validity evaluation of our empirical validation in terms of internal, external, and construct validity.

Chapter 5 presents the analysis and discussion of the results of the empirical validation. In this chapter, the results of each of our hypotheses is presented and discussed. Furthermore, we evaluate the conclusion validity of our results.

Chapter 6 presents the conclusions of the research. Expected contributions, limitations of the study and future research opportunities are also considered in this chapter.

2 LITERATURE REVIEW

To clarify relevant concepts, understand prior work related with this study, and facilitate the development of our theoretical model, we performed several independent literature review rounds with a systematic approach that includes a backward and forward search (Tamm, Seddon, Shanks, & Reynolds, 2011). First, we reviewed literature related to shared cognition to develop a comprehensive view of *shared understanding* (SU) and *cross-understanding* (CU) and how these can be enhanced. Second, a review of ontology literature contributed to defining what a *domain ontology* is, identifying various notions of domain ontology usage, and finding evidence of ontology use to support SU. Finally, we reviewed literature related to conceptual modelling to understand how domain ontologies facilitate domain understanding and what factors may affect this facilitation.

2.1 SHARED COGNITION CONSTRUCTS

To define SU and CU, understand their importance and find current mechanisms to improve these constructs, we used the *Scopus* database to search for highly cited publications that included the terms *"shared understanding", "shared cognition", "shared mental models", "common understanding", "common mental models", and "cross-understanding".* We present the results of the systematic literature review in the following.

2.1.1 Shared Understanding

The first construct, *shared understanding* (SU), is frequently referred to as shared knowledge, shared cognition or a shared mental model (Cooke, Salas, Cannon-Bowers, & Stout, 2000; Mathieu et al., 2000). This construct is defined as "a knowledge structure held by members of a group that enables them to form accurate explanations and expectations for the task, and in turn, to coordinate their actions and adapt their behavior to demands of the task and other group members" (Levesque, Wilson, & Wholey, 2001). That is, shared understanding is a group members' shared, organized understanding and mental representation of knowledge/beliefs about key elements of the group's relevant environment (Mohammed, Klimoski, & Rentsch, 2000). In addition, at a specific time, group members can hold multiple mental models to actively conceptualize and process information about their group, group members, equipment, the environment, and/or the task (Levesque et al., 2001). Moreover, what is shared among group members can be diverse. The various types of knowledge that contribute to shared understanding include (Johnson et al., 2007):

- *Team knowledge:* knowledge pertaining to their individual responsibilities and required actions, and knowledge of the skills, abilities, behavioral tendencies, and knowledge of other group members.
- *Team task:* knowledge of task procedures, strategies, contingencies and scenarios, and environmental constraints.
- *Team interaction:* patterns of interaction and communication, and interdependencies among members, roles/responsibilities, information sources, interaction patterns, communication channels, role interdependencies, information flow.
- *Technology and equipment knowledge:* knowledge regarding equipment functioning and limitations, operating procedures, and likely failures.
- Collective understanding of the current situation.

Similarly, according to Cannon-Bowers and Salas (2001), knowledge to be shared falls into one of four categories:

- *Task-specific knowledge:* this knowledge allows group members to act in a coordinated manner without the need to communicate overtly.
- *Task-related knowledge:* common knowledge about task-related processes, but not necessarily to a single task. For example, what is group work, how it operates, and its importance.
- *Knowledge of groups:* group members need to understand each other to maximize performance.
- *Attitudes and beliefs:* when team members are similar in terms of their attitudes and beliefs, it will cause them to have compatible perceptions about the task/environment and ultimately reach effective decisions.

Prior research has also considered the exposition of shared understanding through a shared mental model for a group task. Fiore and Schooler (2004) provides the following as elements of a shared mental model:

- *Shared problem structure:* an overlapping organized knowledge held by team members such as declarative or procedural knowledge concerning the problem and decision rules associated with the problem.
- *Understanding of each team member's roles and skills:* the roles and skills of the group as they pertain to the problem and the shared awareness that each member of the group possesses this knowledge.

In summary, shared understanding (SU) refers to different types of knowledge being shared between members. Therefore, it is important to specify what type of knowledge is being shared among group members when the construct is being studied. In this study, following the above definitions, we consider SU as a type of shared mental model, that is, the extent of common knowledge of a topic among group members (Ko et al., 2005). That knowledge topic can refer to different aspects, such as task-specific knowledge, task-related knowledge, knowledge of groups, or attitudes and beliefs. Our study is focused on a specific kind of knowledge - domain knowledge - that will be explained later.

Shared understanding has been recognized to be of great importance because it contributes to better organizational outcomes, better task performance and more positive attitudes in groups (Cannon-Bowers & Salas, 2001) despite the motivational states and behavioral processes in groups (Wildman, Salas, & Scott, 2014). Indeed, through SU, members can interpret information in the same way, make compatible decisions, and take correct actions (Cannon-Bowers & Salas, 2001). For example, in software engineering, SU is essential for efficient development when the risk of unsatisfactory outcomes necessitating the rework of project results must be kept as low as possible (Glinz & Fricker, 2014). Similarly, achieving SU between business and IT is of great importance to succeed in the effective use of IT in organizations (Jentsch, Beimborn, Jungnickl, & Renner, 2014).

Regarding mechanisms to improve SU, in theory, there are two approaches than can contribute to enhancing SU in groups:

In the first approach, SU can be improved by *providing timely and accurate information to the group* (Langan-Fox et al., 2004). This approach is more appropriate when group members are not familiar with each other (Resnick, 1991). According to Resnick (1991), providing information to group members can facilitate the process of referential anchoring. This process ensures that members understand terminology or phrases similarly, aiding any future communication between them. Thus, future communication can be improved by initially providing a good quality source of information to group members, which will then enhance SU as we explain later in this section. Not only this, but we also consider that SU might be achieved before group members communicate with each other. For example, if group members individually achieve a correct understanding of the domain, each will have a similar understanding about the domain, and by definition, a domain SU.

The second approach to enhancing SU requires social interaction, which can be achieved through *communication among group members* (M. Tan, 1994; Van den Bossche, Gijselaers, Segers, & Kirschner, 2006). For example, a field study of twenty-eight pairs, comprising a practicing systems analyst and their clients from state government departments in Australia shows that communication

is an important factor to enhance shared understanding (M. Tan, 1994). According to M. Tan (1994), communication in groups to *manage transaction* and *establish rapport* has a positive relationship with shared understanding. *Managing transactions* is defined as the ability to handle the procedural aspects of structuring, controlling, and maintaining a conversation. While establishing rapport "is an aspect of an interpersonal relationships that may be defined as creating a state of harmony, accordance, and congruity developed in a relationship" (M. Tan, 1994, p. 165). Also, the results of an observational field study of an IS development project shows that shared understanding is achieved as a result of the communication process in requirements development (Corvera Charaf, Rosenkranz, & Holten, 2013). However, communication by itself cannot be sufficient to enhance SU (Resnick, 1991) and communication quality among group members is needed to allow them to achieve common comprehension of a particular topic. That is, group members must exchange information related to the topic in which SU is desired. However, over-communication or improper communication may inhibit SU. For example, during IS development projects, purely informal communication may not be effective when dealing with a large number of stakeholders and vast amount of information (Corvera Charaf et al., 2013). According to Langan-Fox et al. (2004) such communication problems can be improved by providing timely and accurate information to the group.

In summary, we consider that SU among groups can be developed by providing a good quality information – information related to the topic to be improved – to group members before they communicate with each other. Also, we consider that SU can be improved through a good communication between group members. The quality of the communication between group members can be improved through by providing good quality information. Thus, it is important to know the best method of delivering this information to group members, i.e. how the information should be presented.

Although some general recommendations have been suggested to improve SU, our literature review indicates a lack of empirically tested mechanisms for improving SU. In fact, most existing studies are focused on one of three aspects related to SU, namely: to define the construct properly, to understand how SU develops, and to explore the impact of SU on groups.

First, some authors focus on improving the definition of the construct. For example, Cannon-Bowers and Salas (2001) present some reflections on the definition of SU. According to their study, it is necessary to address four fundamental questions to improve the definition of SU, namely (1) what must be shared? (2) what does 'shared' mean? (3) how should 'shared' be measured? and (4) what outcomes do we expect shared cognition to affect?

Second, other authors focus on understanding the development of SU in groups. For example, Corvera Charaf et al. (2013) studied how SU develops from a linguistic communication perspective. That is, through an observational field study of an IS development project that developed an internet-based application, they investigated and analyzed the semantic alignment process by which stakeholders achieve shared understanding in requirements development. Similarly, He, Butler, and King (2007) analyzed the emergence and evolution of team cognition in software project groups, and examined how communication activity and group diversity impacted the formation of group cognition. In this study, through a longitudinal study of 51 database development groups, they show that group cognition in terms of both shared understanding task and awareness of expertise location are affected by some forms of communication and group diversity.

Finally, other studies are focused on the impact of SU in groups and the mechanisms to measure SU. For example, as the result of the analysis of group publications in a period of 50 years, Salas et al. (2008) highlight that there is an important impact of SU on team performance and better mechanisms to measure SU are required. Also, Cooke et al. (2000) after a literature review described several mechanisms to measure group knowledge in the context of mapping specific methods onto features of targeted group knowledge. They also present some empirical results concerning the relationship between group knowledge and performance. Another example is DeChurch and Mesmer-Magnus (2010b) who through a meta-analysis of 23 independent studies have empirically examined shared understanding in relation to group process and performance, and test three aspects of measurement as potential moderators, namely elicitation method, structure representation, and representation of emergence. Their results show that a shared mental model impacts the observed relationship between SU and group process. Also, their results show that SU is positively related to group performance regardless of the manner of operationalization. However, Jentsch et al. (2014) propose that more cognitive methods are needed to develop measures for complex constructs like shared understanding. Moreover, the authors develop a content validity proven survey instrument that measures the degree of shared business/IT understanding. Their instrument was validated through an experiment with students and a pilot study with practitioners. Similarly, Johnson et al. (2007) described the development process of an instrument to measure SU and the conceptual framework for factors associated with SU. Their final instrument consists of 42 items that are linked to five factors of SU, namely general task and team knowledge, general task and communication skills, attitude toward teammates and task, team dynamics and interactions, and team resources and working environment. Bierhals et al. (2007) presents two studies to gain insights into the cognitive processes of designers working together in a group, and to clarify the impact of SU on team performance. The authors applied process-oriented research strategies to groups of mechanical engineering students and to multidisciplinary project groups in the automobile industry. Some of their results show that the SU of groups is related to group performance. Moreover, SU is operationalized in terms of group members' skill and the process of interaction. Similarly, Mathieu et al. (2000) test the influence of group members shared mental models on group process and performance. They use 56 undergraduate pairs who flew a series of missions on a personal-computer-based flight-combat simulation. The authors both conceptually and empirically distinguished between group members' task and group-based mental models, and indexed their SU by comparing paired-comparisons matrices. Their results illustrated that both shared group- and task-based mental models related positively to subsequent group process and performance. In the same way, Nelson and Cooprider (1996) explored the relation of shared knowledge between IS groups and their line customers as a contributor to IS performance. Accordingly, the authors tested the relationship of mutual trust, influence, and shared knowledge with IS performance using path analysis in a study of 86 IS departments. Their results show that shared knowledge mediates the relationship between IS performance and both mutual trust and mutual influence.

Throughout the literature review we only found one study that focused on providing and empirically testing a specific mechanism to improve shared understanding of a task. Bittner and Leimeister (2014) derived a validated collaboration process module to systematically support heterogeneous work groups in building shared understanding of their task. Moreover, Bittner and Leimeister (2014) conducted an action research study at a German car manufacturing company to test their approach. Their results show that with the use of the collaboration process module team learning behaviors occur, and SU of the task in a complex work process increases for experienced diverse tool and dye makers.

In summary, our review of the literature shows, most research in shared understanding is focused on three main aspects namely, to define the construct properly, to understand the development of shared understanding in teams, and the impact of SU on teams and mechanism to measure it. All of these studies measured shared understanding in a particular context or for a particular task. However, to the best of our knowledge, none have tried to improve SU through a particular mechanism. In fact, the only efforts undertaken to improve SU that we found were focused on the SU of the task rather than the domain. While there are some general recommendations to improve SU, we could not find much evidence of how those recommendations can be put into practice or any empirical evidence of their effectiveness. Thus, we consider it necessary to continue exploring specific mechanisms able to enhance shared understanding of domains.

2.1.2 Cross-Understanding

According to Huber and Lewis (2011), it is difficult to draw clear conclusions about group outcomes due to variations in group members' knowledge, beliefs, and perspectives. This difficulty arises from studies with contradictory results. For example, some studies mention that high levels of diversity have a positive effect on group performance while other studies conclude that low levels of diversity have a positive effect. To explain some of these inconsistences in literature about social cognition, Huber and Lewis (2011) recently defined cross-understanding (CU), another type of shared cognition construct. Huber and Lewis define CU as the extent to which group members have an accurate understanding of what other group members know about a particular phenomenon. Huber and Lewis (2011) argue that the CU construct explains certain inconsistencies in the literature and we further provide support for specific group outcomes and processes beyond the explanations currently existing in literature.

In a similar way to SU, CU is important in improving performance in groups (Randolph-Seng & Norris, 2011). The benefits of CU are improved communication between members, better elaboration or modification of members' mental models, and better individual and collaborative behaviors in members (Huber & Lewis, 2010). Each of these benefits, can improve group performance and outcomes with CU being principally useful to groups with diverse knowledge, beliefs or perspectives (Huber & Lewis, 2010). However, it is important to note that CU can also have a negative effect on groups when it is based on shared beliefs, preferences, and sensitives (Randolph-Seng & Norris, 2011). According to Randolph-Seng and Norris (2011), only an accurate, knowledge-based CU can have a positive effect on groups. Moreover, according to Huber and Lewis (2010), the benefits of CU are independent of shared knowledge. Thus, achieving a high level of CU does not imply that a high level of SU has been achieved. Accordingly, enhancing CU is also required regardless of SU achievement.

According to Huber and Lewis (2011), CU can be enhanced through team communication or interactive experiences, observations of members' communications or behaviors, and access to member's biographical information. Each of these approaches allows group members to understand the mental models of other members and, hence, to achieve CU. Because CU is a recently defined construct, there is a lack of studies that focus on establishing specific mechanisms to enhance CU and empirical evidence to support such enhancement. We have identified therefore, a need for studies that focus on mechanisms to enhance CU. Specifically, it is important to provide evidence of the effect that the mechanisms proposed by Huber and Lewis have on improving the construct. For example, to elucidate what effect group communication has on CU. Also, it would be of great interest to see what

effect providing high quality information to team members has on CU and compare this with the results for SU to see if CU can be similarly improved.

2.1.3 Other Constructs

There are several shared cognition constructs that have been studied extensively in psychology and management research other than SU and CU. Other examples of shared cognition constructs are *team mental models, team cognition, transactive memory systems,* and *group learning* (Cooke et al., 2000; Huber & Lewis, 2010; Mohammed, Ferzandi, & Hamilton, 2010; Mulder, Swaak, & Kessels, 2002; Van den Bossche et al., 2006). In this section, we define each one of these constructs and differentiate them from the constructs central to this study, i.e. SU and CU.

Team mental models are generally referred to as shared mental models by some authors – e.g. see Cooke et al. (2000), DeChurch and Mesmer-Magnus (2010b), Mohammed and Dumville (2001). However, it is important to also note that several other authors do not consider these two constructs to be equivalent. For example, some authors consider t*eam mental models* to refer to multiple levels or sets of shared mental models and to a synergistic functional aggregation of the teams mental functioning representing similarity, overlap, and complementarity (Langan-Fox et al., 2004; Langan-Fox, Wirth, Code, Langfield-Smith, & Wirth, 2001). In this study, we consider *team mental models* to be shared mental models, covering any study or performed effort that is related with shared understanding. However, although the *team mental model* construct can refer to SU, it differs completely from CU.

Transactive memory systems are defined as shared systems for encoding, storing, and retrieving knowledge from different but complementary areas of expertise (Huber & Lewis, 2010; Thompson & Fine, 1999). Moreover, a *transactive memory system* includes the knowledge held by a particular group member with a group awareness of who knows what (DeChurch & Mesmer-Magnus, 2010a). Therefore, if we compare the *transactive memory system* definition with our definitions of SU and CU, we can assert that a *transactive memory system* differs from both SU and CU.

Team cognition defines the result of the interplay between the individual cognition of each team member and team process behaviors. These team process behaviors, such as communication and coordination, transform a collection of an individual's knowledge to team knowledge that ultimately guides action (Fiore & Schooler, 2004). To illustrate, *team cognition* is an activity, not a property or a product (Cooke, Gorman, Myers, & Duran, 2013). In fact, team cognition can have many components such as shared understanding of the domain, shared understanding of the task, awareness

of expertise location among others (He et al., 2007). As a result, we can assert that *SU* is a component of *team cognition*; however, CU in unrelated to *team cognition*.

Group learning refers to a process of construction of new knowledge in a group and is defined in terms of both processes and outcomes of group interaction. (Mulder et al., 2002) (Mohammed & Dumville, 2001). As a process, group learning is characterized by an ongoing process of reflection and action, which involves asking questions, seeking feedback, experimenting, reflecting on results, and discussing errors (Mohammed & Dumville, 2001). As an outcome, group learning refers to changes in the knowledge and performance of an interdependent set of individuals associated with experience (Mohammed & Dumville, 2001). Thus, we can assert that group learning is different from both SU and CU.

In summary, both SU and CU are outcome constructs. SU is defined as a type of shared mental model, that is, the extent of common comprehension of a topic among group members (Ko et al., 2005) and CU is defined as the extent to which group members have an accurate understanding of what other group members know about a particular topic (Huber & Lewis, 2010). Consequently, if we compare the definitions of SU and CU with the other construct definitions, we can conclude that SU and CU are distinct constructs to *transactive memory systems, team cognition*, and *group learning*. However, *team mental model* can be considered as a synonym of SU. Indeed, SU and CU are a unique result, while either *transactive memory systems, team cognition*, and *group learning* are more complex constructs that not only includes an outcome, but also consider group processes and/or individual behaviors.

2.2 ONTOLOGY FOUNDATIONS

2.2.1 Ontology Definition

The Oxford English Dictionary defines ontology from three different perspectives, namely as a science of philosophy, as a theory relating to the nature of being, and as a logic system (Oxford University Press, 2015). Due to such varied uses of the term 'ontology', it is often difficult to understand what the term refers to in a particular research area – the term 'ontology' makes reference to different concepts depending of both the type and area of application. Thus, in the interest of clarity, we first provide a review of ontology definitions and a clarification of the concept as it is understood in this study.

In general, the term 'ontology' defines the branch of philosophy that studies the nature and structure of reality (Wand, Monarchi, Parsons, & Woo, 1995). However, this term is also used in other research

fields in science and engineering to express a range of diverse meanings. The proliferation of varying ontology definitions stems from the high number of ontology variants and how they are classified. Ontologies have been classified in different ways by researchers depending on their: level of generality, type of structure and conceptualization, appearance, scope, functionality and degree of formality. Although the classifications vary from one author to another, the two most commonly-used classifications are those based on the level of generality and the degree of formality.

According to the level of generality, ontologies can be top-level ontologies or domain ontologies. Top-level ontologies describe abstract and general concepts (Grimm, Abecker, Völker, & Studer, 2011) such as space, time, matter, objects, events, or actions. Because these ontologies are independent of any domain (Guarino, 1998), top-level ontologies can be shared across different domains and applications, and reused as a basis for developing domain ontologies (Grimm et al., 2011). Some examples of top-level ontologies are Cyc's Upper Ontology (Lenat & Guha, 1989), Standard Upper Merged Ontology (SUMO) (Milton & Smith, 2004), and the Basic Formal Ontology (Grenon & Smith, 2004). The second generality level, domain ontologies, capture the knowledge or describe the vocabulary related to a generic domain, such as medicine, geography or project management (Grimm et al., 2011). The number of ontologies that exist in this classification is high – these ontologies can be found in different fields of science and engineering, and are explained in the following sections. Normally, these ontologies should specialize the terms of a top-level ontology; however most ignore top-level ontologies contribute to the creation of better domain ontologies. This can be problematic because top-level ontologies contribute to the creation of better domain ontologies (Smith, 2008).

The representation of ontologies varies from the informal to the formal. The formality of an ontology determines to what extent it is acclimatized by means of logical statements about the domain (Grimm et al., 2011). In function of its formality, an ontology can be represented as a thesaurus, a concept schema, a taxonomy, a conceptual data model, a rule base, or a general logical theory (Sharman, Kishore, & Ramesh, 2004) (see Figure 2-1). For example, ontologies can be represented as a semantic network composed of interlinked concepts. While such a representation may be easy to interpret because of the simplified representation, it may lack important information, such as particular kinds of axioms (Grimm et al., 2011). Ontologies can also be represented through formal languages that allow the capture of all the knowledge related with the domain in question (Gómez-Pérez, Fernandez-Lopez, & Corcho, 2004). One example of such a language is the Web Ontology Language (OWL), which is commonly used in the formalization of ontologies. It is important to note that although ontologies can be represented in many informal ways, representations that lack an organized semantic structure cannot be considered ontologies (Guarino, Oberle, & Staab, 2009, p. 13). For example, a set

of terms without any kind of organization or structure that facilitate an understanding of the domain is just a set of terms and not an ontology.



Figure 2-1. Ontology Representation (Guarino et al., 2009, p. 13)

This situation has made it difficult to achieve consensus about what ontologies are. To clarify this issue, efforts have been undertaken to define ontologies. For example, Almeida (2013), Fonseca (2007), Giaretta and Guarino (1995), Hepp (2008), Kishore and Sharman (2004), and Weber (2002) all contribute to the discussion of ontologies by providing their interpretation of what an ontology is. However, a lack of consensus still remains and, as a result, each field has adopted its own definition. Nonetheless, considering the general purpose of ontologies, ontologies can be broadly defined as a shared conceptualization of reality, which, according to Hepp (2008), can be represented by formal means or informal means. Therefore, in this study, we define an ontology as a specification of a shared conceptualization of reality, which has varying levels of generality and representation formality. In this definition, 'conceptualization' refers to an abstract model of reality through concepts and relations relevant to that reality (Studer, Benjamins, & Fensel, 1998); 'shared' refers to an ontology capturing consensual knowledge of reality by a group of experts in that reality (Studer et al., 1998); 'levels of generality' refers to the notion that ontologies can be very general (the whole reality) to very specific (a domain in particular); and 'formality' refers to how an ontology can be specified through formal means such as OWL (Bera et al., 2011), to informal means such as a graphical representation of concepts and their relationships (Grimm et al., 2011). Consequently, we refer to a domain ontology as the specification of a shared conceptualization of a domain, which can be established through formal or informal means.

2.2.2 Systematic Search of Ontology Use and Effective Use

To understand popular use of ontologies, as well as insights of effective use, we conducted a systematic search of literature to identify the most cited and relevant papers for our analysis (Webster & Watson, 2002). To identify relevant publications, we considered journals, papers, books, and book

chapters since 1985. The selected timeframe was chosen because the references to ontologies rose to prominence in the late 1990s and early 2000s (Figure 2-2). We therefore chose to commence our literature review in 1985 in order to encapsulate any early, but significant papers.



Figure 2-2. Number of Publications per Year Related to Ontologies in the Scopus Database

After establishing the time frame and publication sources, we used the terms 'ontology', 'ontologies', and 'ontological' to search publications in the Scopus database, which returned over 69,000 hits. To obtain a more focused result, as well as for analytical feasibility, we limited our search to titles only, which resulted in 23,618 publications. Given the volume of relevant work, we focused further on identifying the most impactful publications – i.e. to proxy a reasonable set of publications that can be analyzed. We considered two metrics: the age-weighted citation rate and the number of citations since publication. The age-weighted citation rate (AWCR) is calculated for each publication considering the paper age and the number of citations. Using a threshold of AWCR greater than 115 and the number of citations greater than 23, a set of 135 publications was selected for the final analysis. This threshold was considered because one of the most important publications in the ontology-engineering-field theory (Gruber, 1993) had a AWCR = 229 (We defined this value as AWCRg). Hence we wanted a set of publications with similar relevance, that is, at least 50% of the AWCRg or a number of citations greater than the 10% of the AWCRg if the previous condition does not hold.

Although the systematic review allowed us to find a set of relevant and highly impactful publications, this approach may have eliminated significant references because of the high threshold used. Therefore, we also used an exploratory search to find additional relevant publications related with the use and effective use of ontologies. This kind of search did not follow a strict approach, instead it involved using related keywords and phrases in online search tools, and doing backwards and forwards searches on the citations of references found in the first step (Tamm et al., 2011, p. 144).

Using Google Scholar, we searched for 'ontology usefulness', 'ontology use', 'ontology usability', and 'ontology applicability'. The results of these queries allowed us to identify 26 publications, which were added to the previously identified set (resulting in a total of 161 papers). The 161 publications were reviewed in full by one researcher to identify relevant insights about the use and effective use
of ontologies. To do this, each paper was coded through three different phases in order to reduce bias (Neuman, 2011). First, some preliminary codes were established and the publications codified. Then, these codes were organized in categories, new codes were considered, and some discharged. Finally, only six main codes, namely, field of application, nature of use, agent (system or human), type of ontology (domain ontology or top-level ontology), type of ontology representation (formal language or informal language), and type of effective-use evaluation (empirical, theoretical, none) were considered in the findings. The codes formed the criteria for our analysis, the results of which are presented in the next section.

Finally, to further reduce the risk of missing relevant papers, we performed an additional Scopus search using the terms 'ontology', 'ontologies' or 'ontological', and 'shared understanding', 'crossunderstanding', 'common understanding', 'similar understanding' or 'joint understanding' in the title, abstract or keywords of publications since 1985. The query returned 319 results from which titles were compared with our initial 161-paper set and then reduced to 244 results after eliminating duplicates papers. Consequently, this set of papers was analyzed through the use of the NVivo¹. First, we filtered papers that did not have stemmed words of the terms 'empirical', 'existential', 'experimental', 'objective' or 'observational'. We did so to find papers that provided empirical evidence. Second, in the filtered set of papers, we searched for stemmed words of 'person', 'member', 'individual', 'participant', 'subject' or 'group' to find papers that conducted empirical research with participants. This filtering process results in 216 additional papers that were analyzed in full text in order to identify empirical evidence. The results are presented in the next section.

2.2.3 Ontology Use

By definition, and as demonstrated by studies on ontology use (Guarino, 1998; Studer et al., 1998; Uschold & Jasper, 1999), ontologies can be applied to any area in which a representation of reality is required. Ontologies can also be applied to different situations and scenarios, such as the interoperability of computer systems, the development of better conceptual models for IS, or for improving domain understanding, among others.

To identify the types of uses that are common in practice we looked for the application of such ontologies in the set of publications that we selected. The results show that ontologies are used in many fields within science and engineering. For example, in IS, ontologies can be used to facilitate conceptual modelling (Wand, Storey, & Weber, 1999), as a basis for evaluation and improvement of process modelling notations (J. C. Recker, Rosemann, Indulska, & Green, 2009), as components of

¹ NVivo is a software package for qualitative data analysis: <u>http://www.qsrinternational.com</u>.

computer systems (Lutz & Klien, 2006), as repositories of information (Neches et al., 1991) and facilitators of interoperability of the semantic web (Berners-Lee, Hendler, & Lassila, 2001), among others. Similarly, in biology, ontologies can be used as repositories of information that contain vocabularies and classifications of genes, as components of computer systems for the retrieval and extraction of biological information, or for the integration and extension of gene databases (Horrocks & Patel-Schneider, 2011). Ontologies have also been used in industry in operations research and management science. In these settings, they are used to provide a SU of manufacturing-related terms, the reuse of knowledge resources within globally extended manufacturing teams (Lin et al., 2004), as part of the architecture of decision support systems (Niaraki & Kim, 2009), or to model knowledge related to product configuration (Yang, Miao, Wu, & Zhou, 2009).

When the different types of applications are considered, it is clear that there are two general uses of ontology: one relating to the use of ontologies for the support of computer systems and the other relating to the support of human tasks (Table 2-1). These findings are significant because the use of ontologies should not be generalized. Indeed, differences exist between the ontologies used to support systems and those used to support human tasks, and hence clarification is required.

Turnes of Lies	Types of Representation		
Types of Use	Informal	Formal	
Computer Systems Support	0.0%	100.0%	
Information retrieval	0.0%	100.0%	
Interoperability	0.0%	100.0%	
Knowledge representation	0.0%	100.0%	
Systems based on ontology	0.0%	100.0%	
Other	0.0%	100.0%	
Human Task Support	73.1%	26.9%	
Analysis of Conceptual Modelling Grammars	100.0%	0.0%	
Conceptual modelling	92.9%	7.1%	
Domain Understanding	50.0%	50.0%	
Ontology construction	0.0%	100.0%	
Shared Understanding	60.0%	40.0%	

Table 2-1. Use of Ontologies vs. Types of Representation

Although the use of ontologies can be extensive in the two general types of use, for the support of computer systems, we found that ontologies can be used to facilitate information retrieval, to enable the interoperability or communication between computer systems (e.g. semantic web), to represent and store knowledge, as components of operation of computer systems (e.g. support decision systems based on ontology) among other uses. When it comes to ontology use for the support of human tasks, we found that ontologies are used to benchmark and explore the quality and representational capability of conceptual modelling grammars, to improve conceptual modelling tasks, to provide the individual understanding of a particular domain, to facilitate the construction of domain ontologies, and to provide shared understanding among users.

Table 2-1 also shows that both formal and informal representations of ontologies are used to support human tasks. For example, visual ontologies (Bera et al., 2011) and ontologies presented through the use of a computer system (Kim, 2012) have been found to provide support for knowledge identification (a task related with domain understanding). When ontologies are used to support systems, they, by necessity, must be represented in a formal language. For example, OWL is the common language that is used to implement ontologies that will work with the semantic web (Horrocks & Patel-Schneider, 2011). Although some studies show that formal ontologies are used to support human tasks, the ontologies in those studies are embedded in computer systems. Thus, in these studies, it is difficult to establish under what circumstances the ontology itself supports the human task, rather than the system supporting the task. In other words, it is unclear in these studies whether it is the ontology or the system, or both, that supports the human task.

While the benefits of ontologies in systems can be validated through prototypes and their functionality, the benefits of ontologies for support human tasks are more difficult to show, with only empirical evidence able to indicate their effective use. To clarify this aspect, we searched for empirical evidence that supports ontology effective use in supporting human tasks. Table 2-2 shows that there is considerable evidence that indicates that ontologies do indeed support tasks related to conceptual modelling, domain understanding, and ontology construction. For example, in the context of conceptual modelling, J. Recker, Rosemann, Green, and Indulska (2011) demonstrated how ontological deficiencies in modelling grammars affects conceptual modelling, Gašević, Kaviani, and Milanović (2009) assessed how ontologies can support some activities in the software development life cycle, and Sugumaran and Storey (2006) tested the use of ontologies as support for the design of databases. Despite these contributions, there is a lack of empirical evidence to support the claimed effective use of ontologies developed with the purpose of facilitating SU, even though many ontologies were developed with this aim in mind -e.g. the enterprise ontology (Uschold et al., 1998), manufacturing system engineering ontology (Lin et al., 2004), V4 service business model ontology (Al-Debei & Fitzgerald, 2010), the ontology for software requirements modelling (Innab et al., 2012), to name just a few. In fact, after the full text analysis of the additional 216 papers that we collected, we could not find any empirical evidence to justify the effectiveness of ontologies to facilitate SU.

Types of Use	Empirical Evidence of Effective Use		
Types of Ose	Yes	No	
Systems Support	64.0%	36.0%	
Information retrieval	75.0%	25.0%	
Interoperability	40.9%	59.1%	
Knowledge representation	67.3%	32.7%	
Systems based on ontology	100.0%	0.0%	
Other	50.0%	50.0%	
Human Task Support	30.8%	69.2%	
Analysis of Conceptual Modelling Grammars	50.0%	50.0%	
Conceptual modelling	28.6%	71.4%	
Domain Understanding	100.0%	0.0%	
Ontology construction	33.3%	66.7%	
Shared Understanding	0.0%	100.0%	
Total	56.5%	43.5%	

Table 2-2. Empirical Evidence of Effective Use of Ontologies

Because of the lack of empirical evidence for the effective use of ontologies in establishing shared domain understanding, we analyzed whether highly-cited methodologies for building and evaluating ontologies include any approach for evaluation of effective use in such settings. In this analysis, we considered that ontology evaluation may occur at two distinct stages. The first is related to the evaluation of the ontology before it is put into use, that is, the *desired effective use* of the ontology. Whereas the second scenario is related to the evaluation of the ontology after it is put into use, that is, the *resulting effective use* of the ontology. The results of the analysis show that most ontology development methodologies lack the evaluation of ontology's resulting effective use, i.e. evaluation of the ontology post-implementation. Table 2-3 summarizes selected characteristics of these methodologies. Although some methodologies consider an evaluation of the ontology, these approaches are more focused on only assessing the quality of the ontology content, and do not pay particular attention to understanding whether the ontology itself is useful. For example, the methodology used for developing the enterprise ontology (Uschold & King, 1995) has four phases of development: to identify the purpose of the ontology, to build the ontology, to evaluate the ontology, and to document the ontology. Despite this multi-stage process, in the evaluation phase the methodology prescribes evaluation from a knowledge-representation point of view, which is more likely related to a desired quality parameter rather than actual effective use. Similarly, OntoClean (Guarino & Welty, 2009) is an evaluation methodology that allows validation of the ontological adequacy and logical consistency of taxonomic relationships, but it lacks recommendations to evaluate other aspects, such as the effective use of the ontology.

Of those methodologies that propose to evaluate ontology effective use, the evaluation purpose is focused on the perceived effective use of the ontology and/or effective use through application systems instead of the ontology's effective use for human tasks. One example of this is the methodology used in the development of the DO4MG ontology (Delir Haghighi, Burstein, Zaslavsky,

& Arbon, 2013), which evaluates the effective use of the ontology through a decision support system that uses the ontology; however, empirical evidence of its application by users is not considered. Also, although the NeON methodology (Suárez-Figueroa, Gómez-Pérez, & Fernández-López, 2012) suggests the evaluation of effective use, it is orientated towards the evaluation of the application in which the ontology is applied. In addition, in the methodology used to develop the Compliance Management Ontology (CoMOn) (Syed Abdullah, Sadiq, & Indulska, 2013), effective use is evaluated as perceived by users; however, they do not attempt to explore actual ontology effective use. Thus, although there are many methodologies for the construction and evaluation of ontologies, they lack a well-structured prescription of how to evaluate the effective use of ontologies for the specific purpose of shared domain understanding.

Methodology Name	Development Process	'Desired' Effectiveness Evaluation	'Resulting' Effectiveness Evaluation
Cyc (Guha & Lenat, 1990)	\checkmark		
N/A- (Gruber, 1995)			
Based in Enterprise Ontology (Uschold & King, 1995)	\checkmark	\checkmark	
Based in TOVE (Grüninger & Fox, 1995)	\checkmark	\checkmark	
KACTUS (Gómez-Pérez et al., 2004)	\checkmark		
METHONTOLOGY (Fernández-López, Gómez-Pérez, & Juristo, 1997)	\checkmark	\checkmark	
Based in SENSUS (Swartout, Patil, Knight, & Russ, 1996)	\checkmark		
Ontology Development 101 (Noy & McGuinness, 2001)	\checkmark		
Based in On-To-Knowledge (Staab, Studer, Schnurr, & Sure, 2001)	\checkmark	\checkmark	
DILIGENT (Tempich, Pinto, Sure, & Staab, 2005)	\checkmark	\checkmark	
HCOME (Kotis & Vouros, 2006)	\checkmark	\checkmark	
OntoClean (Guarino & Welty, 2009)		\checkmark	
UPON (De Nicola, Missikoff, & Navigli, 2009)	\checkmark	\checkmark	
DOGMA (Jarrar & Meersman, 2009)	\checkmark		
Based on DO4MG (Delir Haghighi et al., 2013)	\checkmark	\checkmark	\checkmark
NeOn (Suárez-Figueroa et al., 2012)	\checkmark	\checkmark	\checkmark
Based on CoMOn (Syed Abdullah et al., 2013)	\checkmark	\checkmark	\checkmark

Table 2-3. Ontology Development Methodologies

As our literature review shows, there is a lack of theoretical and empirical evidence of the real benefits of domain ontologies to support and enhance shared understanding. Thus, it is necessary to explore such benefits, especially considering the high number of ontologies that have been developed for such benefits.

2.3 CONCEPTUAL MODELS AND DOMAIN UNDERSTANDING

Recall we define an ontology as a specification of a shared conceptualization of reality, which has varying levels of generality and representation formality. According to our definition, domain

ontologies are similar to traditional conceptual models used in IS and related fields. Conceptual models – also named as conceptual schemas or conceptual scripts – are representations of a real-world domain² (Allen & March, 2006; Wand & Weber, 2002). In IS and related fields, conceptual models are used generally to document the common understanding that stakeholders have about a domain that is intended to be supported by an information system (J. Recker et al., 2011). Thus, by considering a domain ontology as a particular type of conceptual model, we conducted a review of conceptual modelling literature to understand what factors can affect the understanding of a conceptual model.

To represent conceptual models, modelers (analysts and designers) use a modelling method and a modelling grammar. The modelling method specifies the procedure for constructing the conceptual models (J. Recker et al., 2011). While the modelling grammar defines the constructs and rules to combine those constructs (Allen & March, 2006; Wand & Weber, 2002). For example, by using entity-relationship grammar, database designers develop entity-relationship diagrams that can represent data requirements of a particular domain.

As aforementioned, conceptual models have been used extensively to facilitate domain understanding by users. However, domain understanding can be considerably affected by external and internal elements to the conceptual model. For example, the way in which those conceptual models are represented can affect domain understanding (Burton-Jones & Meso, 2008). As a result, graphical representations are most often used to provide understandability of the conceptual model (Burton-Jones & Weber, 2014; Mayer, 2009).

Conceptual modelling research has dealt with understanding this issue from two perspectives: how to improve the development of conceptual models and how to improve the understanding of conceptual models (Vijay Khatri, Vessey, Ramesh, Clay, & Park, 2006). An example of these studies is: how to improve the semantics of conceptual modelling grammars to facilitate users developing a better understanding of the domain the models represent (Clarke, Burton-Jones, & Weber, 2013). This research indicates that conceptual modelling grammars should not only follow the principles of ontological clarity (J. Recker et al., 2011; Saghafi & Wand, 2014), but also those of logical clarity. In addition to the effect of the conceptual model itself, there are also external factors that may affect the understandability of the domain. For example, prior knowledge of both conceptual modelling and of the application domain will affect the understandability of the domain depending of the cognitive fit with the kind of task (Vijay Khatri et al., 2006). Similarly, prior research shows that not only the

² From here on out, we use the term "domain" to refer to the expression: "real-world domain".

semantics of conceptual models is important, but also the pragmatics in which the model is created and used (Bera, Burton-Jones, & Wand, 2014). Based on these perspectives, we summarize the different types of factors that can affect the understandability of conceptual models (See Table 2-4).

Factor Category	Factor	Supporting Theory or Literature	Operationalization
Conceptualization	Level of captured knowledge	Theory of knowledge representation (Sowa, 2000)	Taxonomy (low-level knowledge) Taxonomy plus semantic relations (medium- level knowledge) Taxonomy plus semantic relations, plus axioms (High-level knowledge)
of the domain	Quality of captured knowledge (Conceptual semantic quality)	Theory of knowledge representation (Sowa, 2000)	Quality of captured knowledge
	Formality of representation grammar	Multimedia learning theory (Mayer, 2005)	Natural language (Informal) Visual language (Semiformal) OWL language (Formal)
Representation of	Ontological completeness	Theory of ontological expressiveness (Wand & Weber, 1995)	Construct deficit
the conceptual model	Ontological clarity	Theory of ontological expressiveness (Wand & Weber, 1995)	Construct redundancy Construct overload Construct excess
	Quality of represented knowledge (Representation semantic quality)	Theory of knowledge representation (Sowa, 2000)	Quality of represented knowledge
	Semiotic clarity	D. Moody (2007)	Symbol redundancy Symbol overload Symbol excess Symbol deficit
Visualization of the represented	Perceptual discriminability	D. Moody (2007)	Visual distance Primacy of shape
conceptual model	Semantic transparency	D. Moody (2007)	Icons Semantically transparent relations
	Graphic complexity	D. Moody (2007) Cognitive load theory (D. L. Moody, 2004)	Modularization Hierarchy
	Type of understanding task	Cognitive fit theory (Vessey & Galletta, 1991)	Level of understanding questions
Involved task	Cognitive load	Cognitive load theory (D. L. Moody, 2004)	Comprehension time Comprehension accuracy Verification time Verification accuracy
Previous	Previous domain knowledge	Cognitive fit theory (Vessey & Galletta, 1991)	People with different domain knowledge levels
knowledge	Previous conceptual modelling knowledge	Cognitive fit theory (Vessey & Galletta, 1991)	Individuals with and without conceptual modelling knowledge

Table 2-4. Factors Affecting Understandability of Conceptual Models

As shown in Table 2-4, there are several factors affecting the understandability or interpretation of conceptual models (e.g., domain ontologies). First, the interpretation of a conceptual model is affected by the conceptualization of the domain. Examples of factors involved in the conceptualization are the level of knowledge captured and the quality of captured knowledge for the modeler (Sowa 2000). Second, the representation of the conceptual model also affects the interpretation of the conceptual model. Examples of these affecting factors include the formality of the representation grammar used to represent the conceptual model (Mayer 2005), the ontological completeness and the ontological clarity of the grammar used to represent the conceptual model (Wand 1995), and the quality of represented knowledge (Sowa 2000). Third, the visualization of the represented conceptual model is another factor affecting interpretation. Examples of these factors are semiotic clarity perceptual discriminability, semantic transparency, and graphic complexity (Moody 2007; Moody 2004). Fourth, the interpretation of a conceptual model is also affected by involved task factors. For example, the type of interpretative or understanding task given to users (Vessey and Galletta 1991) or the cognitive load that the task exerts on the users (Moody 2004). Finally, previous knowledge of the user also affects the understandability or interpretation of a conceptual model. For example, factors involved in conceptual model understandability are previous domain knowledge and conceptual modeling knowledge (Vessey and Galletta 1991).

In summary, as our literature review shows, conceptual models (e.g., domain ontologies) can facilitate domain understanding. However, it is important to note that many factors may also affect such benefits.

2.4 CHAPTER SUMMARY

In this chapter, we presented the results of our literature reviews. Through a systematic approach, we found literature that allowed us the clarify important concepts, which are used later in this study. First, we introduced two shared cognition constructs of interest - shared understanding (SU) and crossunderstanding (CU). We also provided definitions of these constructs, their importance, and ways of enhancing them, in addition to identifying current gaps in literature. Second, we introduced ontology foundations by defining what ontologies are, providing types of use, and exploring evidence of effective ontology use. Finally, we provided insights into conceptual models and their use in facilitating domain understanding. The literature analysis allowed us to establish the required foundations to develop our theoretical model, which is presented in the next chapter.

3 RESEARCH MODEL AND HYPOTHESES DEVELOPMENT

Our main research question in this study is *what is the effect of using a domain ontology on shared understanding (SU) and cross-understanding (CU)?* To address this question, we describe a domain ontology usage process to facilitate SU and CU. This process describes the different stages and activities that team members perform to achieve SU and CU through the use of domain ontologies. Then, considering the aforementioned process, we develop a theoretical model that describes the effect of domain ontology on SU and CU through the mediator of *communication quality* (CQ). According to our theoretical model, domain ontologies also affect CQ, which is an enabler of SU improvement and CU development.

3.1 DOMAIN-ONTOLOGY USAGE PROCESS

As we concluded in our literature review, the facilitation of group understanding constructs, such as SU and CU, through the use of conceptual models is not well described in existing conceptual modelling research. Most of research in this field is focused on the development of conceptual models and facilitating domain understanding between individuals through the use of those conceptual models (V. Khatri, Vessey, Ram, & Ramesh, 2006). Hence, there is still a lack of understanding of how SU and CU develop through the use of conceptual models. To address this problem, first, we describe how using a domain ontology can enhance SU and CU. Then, we will theorize the effects of domain ontologies in facilitating SU and CU based on this description.

Accordingly, in this section, we describe a process through which a domain ontology facilitates SU and CU (Figure 3-1). Based on literature and theory related to conceptual modelling, group communication and shared cognition, we describe and support our usage process. Our process specifies that the facilitation of SU and CU via a domain ontology implies three main phases: *Individual Ontology Interpretation, No Team Interaction*, and *Team Communication* (see Figure 3-1). In the next sections, we describe in detail each of the phases of our domain ontology usage process.





3.1.1 Individual Ontology Interpretation

The first phase of our domain ontology usage process model is the *Individual Ontology Interpretation* phase. During this phase, each group member is required to interpret the domain ontology. Previous studies in conceptual modelling have already established that users can develop an understanding of a domain through interpreting a representation of said domain (Burton-Jones et al., 2009; Gemino, 2004). For example, visual (graphically represented) domain ontologies can help users to learn relevant concepts and relationships about the represented domain (Bera et al., 2011). As result of this ontology interpretation, group members develop an individual mental model of the domain represented in the ontology.

The *Individual Ontology Interpretation* phase is important for successful development of SU and CU. SU and CU are group constructs that require a domain understanding (Huber & Lewis, 2010; Ko et al., 2005). According to the definition of these constructs, SU and CU cannot be established if team members do not individually develop an understanding of the domain. Thus, it is important to consider external and internal factors that could affect the ontology interpretation.

As we have explained in our literature review, representation formality, representation quality, visualization approaches, ontology quality, user knowledge, and user experience, among others, could affect ontology interpretation. In this study however, we are not focused on understanding how these factors affect ontology interpretation. Rather, we are focused on understanding how well an ontology can facilitate SU and CU. Previous studies, including Bera et al. (2011) and Burton-Jones et al. (2009), show that a well-represented graphical ontology improves domain understanding to a greater extent than ontologies without graphical representations. Accordingly, our first assumption is that

graphical ontologies will be better at facilitating SU and CU of a domain than textual descriptions of that domain.

3.1.2 No-Group Interaction

The *No-group Interaction* phase occurs immediately after each group member develops a shared understanding of the domain through the use of the ontology but before group members communication between one another. In this phase, we theorize that SU development can be achieved.

In our literature review, we theorized that a common comprehension of a domain can be achieved when an appropriate source of information is provided to group members. We based this assumption on the definition of SU. Shared cognition research defines and measures SU as the similarity among the domain understandings between individual team members (Johnson & O'Connor, 2008; Salas et al., 2008; F. B. Tan & Hunter, 2002). Accordingly, if, through the use of a source of information, group members correctly understand the information, they will have a common comprehension of the domain.

Based on the previous assumption, we theorize that by providing a graphical domain ontology as a source of information to group members, they will develop SU before they communicate with each other. That is, group members will develop an individual mental model of the domain represented in the ontology as a result of the *Individual Ontology Interpretation*. If we compare those mental models in the group once they are formed, we can establish whether SU was developed or not in the group. Thus, if a domain ontology is provided to group members to help understand a domain, they will develop an initial level of SU.

It is important to note that during this phase – different of the previous one – external or internal factors to the ontology do not affect the development of shared understanding because this is just a measure of the mental models achieved during the first phase. Thus, the individual ontology interpretation phase is one of the most important in which to achieve SU.

3.1.3 Team Communication

Finally, during the Team Communication phase, group members must communicate with each other to improve SU and develop CU. However, normal communication will not result in achieving these goals. As we mentioned in the literature review, high quality communication in terms of exchanging information related to the domain is desired (Resnick 1991) – e.g., points of view about the domain (Mathieu et al. 2000; Stein et al. 2011). Literature shows that team communication can be improved when team members have some common knowledge about the discussion topic (Wu and Keysar

2007). Thus, if a domain ontology allows team members to develop some shared understanding about the domain, communication will be improved in that team members will exchange more information about the domain, i.e. a higher communication quality.

Hence, if the exchanged information allows team members to achieve a more similar point of view of the represented domain, SU will be improved. Additionally, if the information exchanged allows team members to understand the other's point of view about the domain, CU will be developed.

3.2 MODEL DEVELOPMENT AND HYPOTHESES

The key purpose of this study is to understand what the effect of a domain ontology as support tool on SU and CU is. Accordingly, we propose a theoretical model that uses the main assumption of domain ontologies following the ontology-usage process described in the previous section.

The model we propose (Figure 3-2) is grounded in theory and literature regarding conceptual modelling, multimedia learning, team communication, and shared cognition, as described later in this section. These theoretical principles lead us to propose that the usage of a domain ontology as an information tool in group teams facilitates SU development (H1) and enhances Communication Quality (H2). The model also proposes that Communication Quality mediates the effect of domain ontology usage on SU (H3 and H4) and CU (H5 and H6). The impact and influence of Communication Quality (CQ) is further described in depth in subsequent sections. In the following subsections, we also provide a more detailed description of our theoretical model and hypotheses.



Figure 3-2. Domain-Ontology Usage Effect Model (F: Theoretical factor; O: Operationalization of factor)

3.2.1 Impact of Domain-Ontology Usage on Developed Shared Understanding

In this section, we assert that domain ontology usage will contribute to the development of SU before team members communicate with each other. We further explain our prediction of the effect of the ontology use on SU development through theory of conceptual modelling, shared cognition, and multimedia learning.

As previously established, conceptual models facilitate the development of domain understanding in team members (Bera et al., 2011; Burton-Jones & Meso, 2008). Also, group-level understanding constructs such as SU can be achieved without team interaction by providing similar sources of information to team members (Resnick, 1991). Thus, when the same domain ontology is used as an information tool by team members without interaction, they are likely to gain a similar level of domain understanding, which leads to SU development.

Although other sources of information also might enhance the development of SU, we assert that a graphical domain ontology is more effective than a written description of a domain. We based our prediction on the Multimedia Learning Theory which states that "users understand much better from graphical representations that a set of words" (Mayer, 2009). In this study, we are considering graphical representations of domain ontologies. Specifically, we are using a graphical representation of OWL grammar similarly to a previous study by Bera et al. (2011). We theorize that through the use of a domain ontology rather than a written description of a domain, team members will develop a better understanding of the domain. Through a better understanding of the domain, team members can reduce the likelihood of a differential understanding of the domain and hence gain similar facts about the domain, leading to SU development. Therefore, we propose the following hypothesis:

H1. Group members who use a domain ontology develop higher levels of shared understanding than those who do not use a domain ontology.

As previously discussed, SU is a direct result of the similarity in users' domain understanding. Thus, this study establishes SU through comparing the individual understandings of group members (F. B. Tan & Hunter, 2002), and operationalizes SU through establishing the extent of similarity of answers to comprehension questions between group members.

3.2.2 Impact of Domain-Ontology Usage on Communication Quality

As specified in our domain ontology usage process, group members require good communication to enhance SU and create CU. In this section, we assert that domain ontology usage enhances Communication Quality (CQ) during team communication. We further explain our prediction based on communication theory and conceptual modelling.

Previous studies of computer mediated communication have defined CQ as an important factor that affects the performance of group members in particular tasks (Kahai & Cooper, 2003; Kayworth & Leidner, 2002; Lowry, Romano, Jenkins, & Guthrie, 2009; O'Reilly & Roberts, 1977). In these studies, CQ is defined in terms of Communication Openness, Discussion Efficiency, Task Discussion Effectiveness, and Communication Clarity. Accordingly, we follow previous studies and define CQ using the same lenses. This definition allows us to consider whether relevant information is exchanged between group members during group communication, in addition to allowing us to understand other factors that are relevant to communication and can be affected by the ontology use. Communication Openness is defined as "the receptiveness of a group member to the communication of others" (Lowry et al., 2009; O'Reilly & Roberts, 1977). Discussion Efficiency reflects the extent to which group members are oriented towards results, the effective use of time in interactions, and how thoroughly issues are discussed (Lowry et al., 2009). Task Discussion Effectiveness reflects the extent to which group members participate in the discussion, develop discussion content, exchange information, and examine issues and ideas effectively and critically (Lowry et al., 2009). Communication Clarity defines to which extent exchanged messages between group members are clear (Kahai & Cooper, 2003; Kayworth & Leidner, 2002).

Based on these definitions, we predict the effect of domain ontology usage on each of these constructs. Recall that domain ontologies facilitate domain understanding of the represented domain (Burton-Jones et al., 2009; Gemino, 2004). Therefore, by having a good understanding of the domain, group members will likely have more confidence to talk about what they understand about the domain in order to solve a particular issue. This increased confidence implies that *Communication Openness* is enhanced. Group members will also likely be more focused on solving the issue rather than trying to further understand the domain. Similarly, they will likely reduce the time spent on interactions explaining the domain and be more willing to further discuss the problem; consequently, *Discussion Efficiency* is enhanced. Furthermore, group members will likely be more participative because they are confident about what they know of the domain. This will likely lead to group members being more open to develop and discuss content, increase the exchange of ideas about domain issues, examine the issues more effectively and critically; thus enhancing *Task Discussion Effectiveness*. Finally, group members are also likely to convey clearer messages about the domain, improving communication clarity.

Accordingly, to enhance Communication Quality, group members must thoroughly understand the domain. Moreover, if group members use a source of information that does not allow them to understand the domain clearly, Communication Quality will be compromised. We thus propose the following hypothesis:

H2. Group members who use a domain ontology engage in better communication quality in terms of communication openness, discussion efficiency, task discussion effectiveness, and communication clarity than those who do not use a domain ontology.

3.2.3 Impact of Communication Quality on Shared Understanding Improvement and Cross-Understanding Development

In this section, we assert that Communication Quality (CQ) enhances SU improvement and CU development. We base our arguments on theory related to team communication and shared cognition.

As per our domain ontology usage process, SU and CU can be developed and improved through good team communication (Mathieu et al., 2000; Resnick, 1991; Stein et al., 2011). Based on communication theory (Kahai & Cooper, 2003; Lowry et al., 2009; O'Reilly & Roberts, 1977), we defined CQ in terms of four dimensions – *Communication Openness, Discussion Efficiency, Task-Discussion Effectiveness,* and *Communication Clarity*. Therefore, SU improvement and CU development are a function of the degree to which users have good communication in terms of these four dimensions.

By having better *CQ*, which is improved by domain ontology usage, first, group members will be more open to exchanging information about the domain; second, have a more efficient discussion about the domain; third, have a more effective discussion of tasks in which they are involved; and communicate more clearly with each other. This leads to the exchange of more precise information about the domain between group members, which can subsequently lead to achieving agreement about the said domain among group members and SU improvement (He et al., 2007; Kacmar, 2003; Kayworth & Leidner, 2002; Ko et al., 2005; Lowry et al., 2009). Additionally, this exchange of information allows group members to understand what other group members know or think about the domain, creating CU.

To summarize, with a higher quality communication enabled by domain ontology usage, users can focus on relevant information represented about the domain and are therefore able to better exchange their beliefs and personal knowledge with one another, thus improving SU and leading to the achievement of CU. Contrarily, during a less effective communication, users are distracted and need

to bring to bear prior knowledge to interpret the representation, the focus of the discussion may be disrupted and lead to a low exchange of information about the domain between users. Thus, we hypothesize:

H3. Group members who use a domain ontology have greater shared understanding improvement than those who do not use a domain ontology.

H4. Communication quality in terms of communication openness, discussion efficiency, task discussion effectiveness, and communication clarity mediates the effect of domain ontology usage on the improvement of share understanding.

H5. Group members who use a domain ontology develop greater cross-understanding than those who do not use a domain ontology.

H6. Communication quality in terms of communication openness, discussion efficiency, taskdiscussion effectiveness, and communication clarity mediates the effect of domain ontology usage on the development of cross-understanding.

CU can be operationalized in two main ways, *viz*. Perceptual Approach and Behavioral Manifestation Approach (Huber & Lewis, 2010). Nevertheless, each approach has some shortcomings. On the one hand, the Perceptual Approach is easy to execute, but is based on the perception of participants. On the other hand, the Behavioral Manifestation Approach is based on the perception of an external observer and is difficult to implement and quantify in terms of levels of understanding. This study adopts the Perceptual Approach because it is less subjective to code for analysis. Accordingly, we operationalize CU by asking each participant their perception of answers that the other participant will give to comprehension questions.

Details of the measurement instruments used to measure our different operationalized variables will be covered in the next chapter on empirical validation.

3.3 CHAPTER SUMMARY

In this chapter, we developed a theoretical model that allows us to understand our research question. To develop the model, we first described a domain ontology usage process to facilitate shared understanding (SU) and cross-understanding (CU). Based on our usage process, we then developed the theoretical model that describes the effect of domain ontology usage on SU and CU. Specifically, our model describes that *communication quality* (CQ) mediates the effect of domain ontology usage on SU and CU are supported by

literature and theory related to conceptual modelling, multimedia learning, group communication and shared cognition.

4 EMPIRICAL VALIDATION

To the extent of our knowledge, we believe that our study is the first to test the effect of domain ontology on shared understanding (SU) and cross-understanding (CU) in IS research. Therefore, we selected an experimental design to maximize internal validity (Cook & Campbell, 1979; Trochim & Donnelly, 2008). To extend the external validity of our study, we selected an appropriate domain and thus, domain ontology, to simulate a more practical situation necessary to provide SU and CU. This chapter explains details of the selected domain, the used ontology, and the experimental design used in our study.

4.1 COMPLIANCE MANAGEMENT DOMAIN

Compliance management is defined as "a process of meeting or adhering to the organization's obligations, such as those required by laws, industry and organizational standards and codes, principles of good governance and, accepted community and ethical standards" (Standards Australia, 2006, p. 5). Compliance management has been discussed within academia and industry for two main reasons; the high cost of compliance and the economic impact of non-compliance. Firstly, the operational cost of compliance management is very high. For instance, Protiviti (2012, p. 3) in their report of the Sarbanes-Oxley Compliance Survey found that the expenditures for compliance with this legislation in any size of company is between USD 100,000 to USD 500,000 annually. Secondly, the consequences of failing to align with regulations or standards can be drastic. For example, the Australian company Cotton On Kids was required to pay AUD \$1 million for non-compliance with a safety standard (Australasian Compliance Institute, 2013b). Thus, compliance management can cause significant economic issues, making it a relevant issue for both industry and academia.

Significant efforts have been made in academic research for ways to improve compliance management in organizations. Some effort has been focused on the mechanisms of compliance by design, that is, to embed the compliance requirements in business processes (Sadiq & Governatori, 2010; Sadiq, Governatori, & Namiri, 2007). Other researchers paid attention to improving the methods of checking compliance in business process (El Kharbili, 2012; Elgammal, Turetken, Heuvel, & Papazoglou, 2011; Ramezani, Fahland, & Aalst, 2012). Additionally, others emphasized establishing new technologies and prescriptive frameworks to achieve organizational compliance (COMPAS, 2011; Foorthuis & Bos, 2011; Schäfer, Fettke, & Loos, 2012), or focused on both extracting and translating the compliance requirements into process rules (Sapkota, Aldea, Younas, Duce, & Banares-Alcantara, 2012). Finally, some studies have concentrated on developing IT tools

or artefacts that help organizations manage and understand compliance knowledge (Boella, Humphreys, Martin, Rossi, & Torre, 2012; Syed Abdullah et al., 2012b). Despite these efforts, more recent studies show that the maturity level of compliance in organizations is still low (Australasian Compliance Institute, 2013a, p. 7).

Many of the IS tools for compliance management that exist in the market have achieved a high level of maturity in the features offered (Gartner, 2012). However, recent industry surveys show that organizations still do not understand the importance of compliance management. For instance, the Australasian Compliance Institute (2013a, p. 7) concluded that organizations do not understand the real value-add of compliance functions to business operations. Eighty-seven per cent of respondents answered that the level of competency and maturity in proficiency and training needs are at a low level. Similarly, the Open Compliance & Ethics Group (2012) reported that in over 50 per cent of companies surveyed, the activities of compliance did not provide a clear view of compliance effectiveness and performance. Indeed, 45.2 percent of respondents indicated that compliance activities are not integrated at all. These numbers show that although many organizations are using different tools that help control compliance, there is still a lack of understanding of how compliance management is related to or impacts the organizations operations, in addition to a lack of common understanding of compliance in general within different functions of the same organization.

In summary, compliance management has been an important topic discussed within academia and industry for three main reasons: high-compliance management cost (Protiviti, 2012), drastic consequences of failing to comply (Australasian Compliance Institute, 2013b), and a weak and conflicting understanding of compliance management concepts in organizations (Australasian Compliance Institute, 2013a; Open Compliance & Ethics Group, 2012). Thus, we conclude that compliance management is a suitable setting in which to study the effectiveness of domain ontology in facilitation of SU and CU.

4.2 THE COMPLIANCE MANAGEMENT ONTOLOGY

4.2.1 Description of the Ontology

The *compliance management ontology* (CoMOn) is an informal ontology developed with the specific purpose to facilitate the development of a shared understanding of compliance management (Syed Abdullah et al., 2013). It was constructed based on a five-step approach composed of: identification of purpose, scope and relevant sources, ontology building, ontology evaluation, ontology refinement,

and ontology documentation (Syed Abdullah et al., 2013). CoMOn uses three kinds of sources: scholarly articles, industry experts, and industry practitioners (Syed Abdullah et al., 2013).

The resulting ontology (CoMOn) is composed of 81 concepts, which have varying relationships among them and are structured into four levels of details (Syed Abdullah, Sadiq, & Indulska, 2012a). Figure 4-1 shows a top-level graphical representation of CoMOn.



Figure 4-1. Graphical Representation CoMOn (Adapted from Syed Abdullah et al., 2012a)

4.2.2 Quality Evaluation of CoMOn

Although some aspects of CoMOn, such as clarity, interpretability, accuracy, comprehensiveness, and relevance have been evaluated (Syed Abdullah et al. 2013), there are a number of ontology evaluation methodologies that propose additional aspects for evaluation (Delir Haghighi et al. 2012; Sabou and Fernandez 2012; Suárez-Figueroa 2012). To ensure that CoMOn is of a high enough quality for this study, we performed a validation of two additional aspects: coverage of the ontology and completeness of definitions. First, the coverage evaluation allowed us to verify whether all important terms of the relevant domain were incorporated (Delir Haghighi et al. 2012). Even though CoMOn was constructed based on industry reports, industry practitioner input and Australasian and industry standards (Syed Abdullah et al. 2013), the original evaluation did not evaluate the coverage of the ontology. Coverage defines the completeness and coverage of terms and concepts to represent an information domain (Delir Haghighi et al. 2013). Instead the evaluation focused on the quality of concepts and usability of the ontology. Second, the completeness evaluation of definitions allowed us to determine whether the ontology definitions are sufficiently complete to satisfy the requirements for which the ontology was created, that is, to facilitate SU of the compliance management domain (Grüninger and Fox 1995; Yu et al. 2005).

Based on the evaluation approach developed by Delir Haghighi et al. (2012), we evaluated the coverage of the ontology through the use of Leximancer. Coverage is one of the eight evaluation criteria promoted by this approach, namely 1) clarity, 2) consistency/coherence, 3) conciseness, 4) expandability/expandability, 5) correctness, 6) completeness, 7) minimal ontological commitment, and 8) coverage. Accordingly, the sources of data for this evaluation were the Compliance Program Standard and its constituent base of standards. To carry out the evaluation, we considered seven industry standards in addition to the two standards considered during the development of the ontology (See Table 4-1). We did so because CoMOn was developed based on only AS 3806 Compliance Programs and AS/NZS ISO 31000:2009 Risk Management-Principles and Guidelines – see Syed Abdullah et al. (2012b).

The results of content analysis facilitated through Leximancer show that most of the concepts of the 9 standards are included in CoMOn, and those few that are not included are deemed to be not relevant (see Figure 4-2). For example, a relevant term included in CoMOn is "process". The "process" term is part of the concept of Business Process Management. An example of a term that is not relevant is "form". Although this term is in the standards, it does not represent an important element in the domain (based on our experience). Thus, we conclude that CoMOn is of high enough quality to be used in our study to investigate the effect of domain ontology on CU and SU.

Standard Code	Standard Name	Considered During CoMOn Development
AS/NZS ISO 190011:2003	Guidelines for quality and/or environmental management systems auditing	No
AS/NZS 4801:2001	Occupational health and safety management systems	No
AS 8000-2003	Good governance principles	No
AS ISO 10002-2006	Customer satisfaction-guidelines for complaints handling in organizations (ISO 10002:2004, MOD)	No
AS/NZS 14001:2004	Environmental management systems- requirements with guidance for use	No
HB 436:2004	Risk management guidelines. Companion to AS/NZS 4360:2004	No
AS/NZS ISO 31000:2009	Risk Management-Principles and Guidelines	Yes
AS/NZS ISO 9001:2008	Quality Management Systems-Requirements	No
AS 3806-2006	Compliance programs	Yes





Figure 4-2. Conceptual Map produced by Leximancer

4.3 EXPERIMENTAL DESIGN

Once we had determined a suitable domain ontology to test our hypotheses, we had to establish an appropriate experimental design. We considered a two-group, post-test only, randomized experimental design (Shanks, 2008; Trochim & Donnelly, 2008). Although this design is the simplest design of all experimental designs (Trochim & Donnelly, 2008), it is the most appropriate to determine whether two experimental groups are different. In our theoretical model, we needed to test if a group that used a domain ontology performed differently to one that did not use the ontology.

Accordingly, our treatment group would use CoMOn and our control group would use a portion of the standard ISO 19600:2014 (British Standards Institution, 2014), which provides guidelines for compliance management systems. By using a portion of the standard, we provided more external validity to the experiment design as it is source of information highly used in the industry to achieve compliance. In practice, team members are not able to answer questions or solve a particular problem related to a domain if they do not have a source of information that allows them to first understand the domain (Bera et al., 2011; Burton-Jones & Meso, 2008). More details about the use of CoMOn and the standard are in the subsequent treatment materials section.

Our experiment design has an important novel difference to previous experimental designs in conceptual modelling. Previous studies in conceptual modelling are focused on an individual's domain understanding – procedures and measures and are applied to only one participant e.g. see Vijay Khatri et al. (2006) – whereas our study is focused on group understanding – procedures and measures are applied to groups of two participants. Therefore, in our study, each experiment requires the participation of two-student pairs, as is further described in the measurement and procedure section.

4.4 PARTICIPANTS AND RECRUITMENT PROCESS

In Chapter 3, we mentioned that domain understanding development is important for success in developing SU and CU. When domain ontologies are used to provide domain understanding in group members, such an understanding can be affected by prior knowledge; domain or conceptual modelling (Burton-Jones & Meso, 2008). Specifically, in our study, domain understanding can be affected by either prior compliance management knowledge or the interpretation of conceptual models. In this study, we are interested in users that have zero to low knowledge in both compliance management and conceptual modelling – i.e. novice practitioners. We selected this group of users because the aim of this study is to understand the effect of an ontology on SU and CU. This change in understanding is most important and significant in novice users rather than expert users (Burton-Jones, Clarke, Lazarenko, & Weber, 2012), as aforementioned in section 1.1.

To represent this population of domain ontology users, we chose a student sample. We invited students from a course of database fundamentals offered in the School of Information Technology and Electrical Engineering at the University of Queensland to participate. Students from this course are first-year students from different backgrounds that have a requirement to learn fundamental concepts of relational databases. This cohort of students, who have zero or low level of conceptual modelling and compliance management knowledge helps us to recreate the group of users that we

were targeting. Thus, we believe that our student sample has an acceptable similarity to the target population (Compeau, Marcolin, Kelley, & Higgin, 2012). In fact, previous studies consider that students can be a proxy for novice and junior practitioners if the task is designed properly – i.e. see Burton-Jones and Meso (2008), Bera et al. (2011), Bera et al. (2014), Vijay Khatri et al. (2006). To conclude, we believe that the participants in our study adequately represent novice and practitioners in organizations, which are the target of our research.

Suitable students were invited to participate through an invitation presented at the end of a lecture by the coordinator of the aforementioned course at the beginning of semester 1 2016. Figure A-1 in Appendix A.1 shows the invitation that was presented and distributed among the students. Students who were interested in participating had the option of getting information through a Facebook page or by directly contacting the main researcher. If students decided to participate, they were asked to register in a Doodle³ poll by providing their name, contact email, and the times when they would be available.

Participants who were willing to participate and had a similar availability were paired randomly and notified by email about their participation. The email contained an appointment which specified the date and time of students' participation (See example on the notification on Figure A-2 in Appendix A.2). The notification also attached a digital copy of the *Participant Information* and *Consent Form* (See Figures in Appendix A.3).

4.5 INTERNAL VALIDITY

Internal Validity is the approximate truth about inferences regarding cause-effect or causal relationships (Trochim & Donnelly, 2008). That is, we have not made internal errors to the design of a research project that might produce false conclusions (Neuman, 2011). There are twelve threats to internal validity that needs to be controlled (Cook & Campbell, 1979; Neuman, 2011). Table 4-2 presents the different internal validity threats, their definitions, and the general controls that we undertook to mitigate such threats. These controls were considered during our experimental design (See section 4.3).

According to Table 4-2, there are 12 likely threats that could impact the internal validity of our experimental design. For each one of these threats we adopted a mitigation strategy to decrease the likely of occurrence of them. To follow we describe these threats and the mitigations adopted:

³ Doodle is a web tool that facilitates the process of scheduling events: <u>http://doodle.com</u>

	Threat	Description		Mitigation Measures
1.	Selection bias	Experiment has more than one group, which differs or does not differ from equivalent groups.	•	Participants were assigned randomly to the control and treatment group.
2.	History	An event unrelated to the treatment will occur during the experiment and influence the dependent variable.	•	Experimental sessions were short in time (less than 2 hours). No unrelated event was present during each experiment session.
3.	Maturation	Result of a threat that a biological, psychological, or emotional process within participants other than the treatment occurs during the experiment and influences the experimental variable.	•	Experiments were short in time (less than hours), which decrease the risk of maturation. Also, we considered a control and treatment group, which helped to show similar changes over time.
4.	Testing	The effect of a pretest measure in the experiment.	•	Our experiment design did not include a pretest.
5.	Instrumentation	The instrument or dependent variable measure changes during the experiment.	•	Both our instrument and dependent variables did not change during the experiment.
6.	Experiment mortality	Some research participants do not continue through the entire experiment.	•	Samples in which participants left the experimental session were not considered during the analysis.
7.	Statistical regression effect	 a) Participants are unusual regarding the dependent variable. b) Measure is such that most people score very high or very low on a variable, random chance alone will produce a change between the pretest and the posttest. 	a) b)	Participants had a practice section in the experiment familiarize them with the questions related to the dependent variables. Our experiment design did not consider a pretest.
8.	Diffusion of treatment or contamination	Participants in different groups will communicate with each other and learn about the other's treatment.	•	Each experimental session had a particular schedule. A session included participants from only one group – control group or treatment group.
9.	Compensatory behaviour	Something of value is provided to one group of participants but not to other and the difference becomes known.	•	Both control group and treatment group received the same economic incentive for their participation.
10.	Experimental expectancy	The experimenter indirectly communicates a desired outcome to participants.	•	Experimenter involvement during experiments was only to facilitate the experimental session. Experimenter had minimal communication and involvement with participants during the experiment.
11.	Demand characteristics	Participants pick up clues about the hypothesis or an experiment's purpose and they modify their behaviors to what they think the research demands of them.	•	There were many questions during the experiment, which would had made it difficult for participants to realize our hypothesis.
12.	Placebo effect	Some participants receive a placebo, but they respond as if they have received the real treatment.	•	We do not used a placebo in our experiment. In fact, participants did not know if they were in a control or treatment group.

 Table 4-2. Threats to Internal Validity and their Mitigation (Neuman, 2011)

1. *Selection bias* threat is the effect produced when an experiment has more than one group, which differ or do not from equivalent groups (Neuman, 2011). To mitigate this threat, participants were assigned randomly to the control and treatment group.

- 2. *History* threat occurs when an event unrelated to the treatment will occur during the experiment and influence the dependent variable (Neuman, 2011). To avoid this problem, experimental sessions were short in time (less than 2 hours). Also, no unrelated events were present during each experiment session.
- 3. *Maturation* threat is the result of a threat that a biological, psychological, or emotional process within participants other than the treatment occurs during the experiment and influences the experimental variable (Neuman, 2011). To avoid this problem, experiments were short in time (less than hours), which decreased the risk of maturation. Also, we considered a control and treatment group, which helped to show similar changes over time.
- 4. *Testing* threat is the effect of a pretest measure in the experiment (Neuman, 2011). To avoid this problem, our experiment design did not include a pretest.
- 5. *Instrumentation* threat occurs when the instrument or dependent variable measure changes during the experiment (Neuman, 2011). To avoid this problem, both the instrument and dependent variables did not change during the experiment.
- 6. *Experiment mortality* occurs when some research participants do not continue through the entire experiment (Neuman, 2011). To avoid this threat, samples in which participants left the experimental session were not considered during the analysis.
- 7. *Statistical regression effect* threat can occur in 2 situations. First, when participants are unusual regarding the dependent variable (Neuman, 2011). It also occurs when measure is such that most people score very high or very low on a variable, random chance alone will produce a change between the pretest and the posttest (Neuman, 2011). To avoid this threat, first, participants had a practice section in the experiment to familiarize themselves with the questions related to the dependent variables. Finally, our experiment design does not consider a pretest.
- 8. *Diffusion of treatment or contamination* threat occurs when participants in different groups communicate with each other and learn about the other's treatment (Neuman, 2011). To avoid this threat, each experimental session had a particular schedule. Also, an experimental session included participants from only one group control group or treatment group.
- 9. *Compensatory behavior* threat occurs when something of value is provided to one group of participants but not to other and the difference becomes known (Neuman, 2011). To avoid this threat, both control group and treatment group received the same economic incentive for their participation.
- 10. *Experimental expectancy* threat occurs when the experimenter indirectly communicates a desired outcome to participants (Neuman, 2011). To avoid this threat, experimenter involvement during experiments was only to facilitate the experimental session. In fact, the

experimenter only had minimal communication and involvement with participants during the experiment.

- 11. *Demand characteristics* threat occurs when participants pick up clues about the hypothesis or an experiment's purpose and they modify their behaviors to what they think the research demands of them (Neuman, 2011). To avoid this threat, there was a large set of questions during the experiment, which would had made difficult to participants to determine purpose.
- 12. *Placebo effect* threat occurs when some participants receive a placebo, but they respond as if they have received the real treatment (Neuman, 2011). To avoid this threat, we do not used a placebo in our experiment. In fact, participants were tested blindly and did not know if they were in a control or treatment group.

4.6 EXTERNAL VALIDITY

External validity is the degree to which the conclusions in a study would hold for other persons in other places and at other times (Trochim & Donnelly, 2008). In experimental research, external validity is concerned with the generalization of the results outside the laboratory. Although experimental research has strong internal validity, it has some limitations in terms of external validity (Trochim & Donnelly, 2008). The following control was undertaken to increase external validity of this study:

- The participants that we recruited had similar characteristics to our identified requirements i.e. low level of knowledge of the domain and low level of knowledge in conceptual modelling.
- 2. We selected an appropriate domain and thus, domain ontology, to simulate a more practical situation necessary to provide SU and CU. Specifically, we selected the compliance management domain and CoMOn. Compliance management has been an important topic discussed within academia and industry for three main reasons: high-compliance management cost (Protiviti, 2012), drastic consequences of failing to comply, and a weak and conflicting understanding of compliance management concepts in organizations (Australasian Compliance Institute, 2013a, 2013b; Open Compliance & Ethics Group, 2012). Thus, compliance management is a suitable setting in which to study the effect of domain ontology in facilitation of SU and CU.

4.7 TREATMENT MATERIALS

Recall that our experiment design considers a treatment group (usage of CoMOn) and a control group (usage of ISO 19600:2014 standard). In this section, we provide more details of the materials used during our experiment in each one of these groups.

4.7.1 Treatment Group Materials

The treatment group received a domain ontology (CoMOn) to complete the experimental task. Thus, they were required to be familiar with how to interpret the ontology and how to answer the questions. To make participants familiar with these two issues in the treatment group, we developed a simple practice section before they received CoMOn to complete the experimental task.

We adapted both the materials and questionnaires used by Bera et al. (2011) to develop this practice section. The practice section consisted of a training section and a practice questions section. The training section gave an introduction to the main constructs used to represent CoMOn (See Appendix C.1). To represent the ontology, we used a graphical representation of Web Ontology Language (OWL). OWL is a formal language developed for the semantic web and highly used for the representation of ontologies in general (Bera et al., 2011; Gómez-Pérez et al., 2004). This formal language contains several constructs and some of these constructs (i.e. classes, external properties, and internal properties) can be represented graphically as is the case of the provided visualizations in Protégé, OWLGrEd (Cerans, Liepins, Sprogis, Ovcinnikova, & Barzdins, 2015) or visual ontologies (Bera et al., 2011). We extended upon the study of Bera et al. (2011) in order to understand the effect of graphically-represented ontologies on domain understanding at the group level.

The practice question section consisted of eight questions related to a travel domain (See Appendix C.2), which had to be answered according to information represented in a *travel domain ontology* (See Appendix C.3). The questions included true/false answers and an option "I don't know" in the case that participants did not know the answer. We implemented this option to check that participants were engaged with the task.

Often, because of the large number of terms and relations involved, ontologies are very large. Thus, if we wanted to use the entire CoMOn ontology in measuring SU and CU, the participants may be subjected to cognitive load, making it difficult to measure the desired variables (D. L. Moody, 2004). Cognitive load refers to the total amount of mental effort being used in the working memory. To avoid this problem, only a subsection of the ontology was used in the experiment. We choose the most representative concepts and relations in order to measure how the knowledge captured in this

ontology can be transferred to the users of the ontology. Also, we only considered concepts and relations that were part of the standard ISO 19600:2014 (British Standards Institution, 2014) used in the control group in order to maintain information equivalence among the materials (Aguirre-Urreta, 2008; Burton-Jones & Meso, 2006; Burton-Jones et al., 2009; Parsons, 2011; Shanks, 2008). Figure 4-3 shows the resulting version of CoMOn that was used during the experiments.



Figure 4-3. Simplified Version of CoMOn

4.7.2 Control Group Materials

Similar to the treatment group, the control group received practice section materials and the final materials necessary to complete the experimental task. Different to the treatment group, the practice section for the control group did not include a training section and only a set of questions were presented. We did so because the control group only received a domain described textually. The practice section consisted of eight questions related to a travel domain (See Appendix C.2), which had to be answered based using the textual description of a travel domain (See Appendix C.4). We adapted both the materials and questionnaires used by Bera et al. (2011) to develop this practice section. Similar to the treatment group, the questions included true/false answers and an option "I don't know" in the case that participants did not know the answer. We implemented this option to check that participants were engaged with the task at hand.

As we mentioned before, we used a simplified version of the ISO 19600:2014 Compliance Management Systems – Guidelines (British Standards Institution, 2014). The simplification was performed with consideration to two issues. First, the standard is very extensive and we wanted to reduce cognitive load on participants (Sweller, 1988). Secondly, we wanted to maintain information equivalence with the materials used in the treatment group (those using CoMOn) (Aguirre-Urreta, 2008; Burton-Jones & Meso, 2006; Burton-Jones et al., 2009; Parsons, 2011; Shanks, 2008). Table 4-3 shows the sections considered in the resulting simplified version of the standard. For copyright reasons, we have been unable to include a complete copy of the simplified version of the standard.

Section	Section Title	Section	Section Title
	Introduction	4.5.2	Maintenance of compliance obligations
3	Terms and definition	4.6	Identification, analysis and evaluation of compliance risks
3.1	Organization	7	Support
3.2	Interested party (preferred term)	7.1	Resources
	Stakeholder (admitted term)	7.2	Competence and training
3.3	Top management	7.2.1	Competence
3.4	Governing body	7.2.2	Training
3.5	Employee	7.3	Awareness
3.6	Compliance function	7.3.1	General
3.7	Management system	7.3.2	Behavior
3.8	Policy	7.5	Documented information
3.9	Objective	7.5.1	General
3.1	Process	7.5.2	Creating and updating
3.11	Risk	7.5.3	Control of documented information
3.12	Compliance risk	8	Operation
3.13	Requirement	8.1	Operational planning and control
3.14	Compliance requirement	8.2	Establishing controls and procedures
3.15	Compliance commitment	8.3	Outsourced processes
3.16	Compliance obligation	9	Performance evaluation
3.17	Compliance	9.1	Monitoring, measurement, analysis and evaluation
3.18	Noncompliance	9.1.1	General
3.19	Compliance culture	9.1.2	Monitoring
3.2	Code	9.1.3	Sources of feedback on compliance performance
3.21	Organizational and industry standards	9.1.4	Methods of information collection
3.22	Regulatory authority	9.1.5	Information analysis and classification
3.23	Competence	9.1.6	Development of indicators
3.24	Documented information	9.1.7	Compliance reporting
3.36	Risk treatment	9.1.8	Content of compliance reports
4.5	Compliance obligations	9.1.9	Record-keeping
4.5.1	Identification of compliance obligations	6	

4.8 MEASUREMENTS

Most measures that we used to quantify our variables at the individual level were based on existing studies. However, our group level variables, SU and CU, had to be defined after the individual

components were established. Subsequently we describe the different measures used to quantify our dependent and control variables.

4.8.1 Dependent Variables

Based on our theoretical model, there are several dependent variables to be measured: *Shared Understanding Development, Shared Understanding Improvement, Cross-Understanding Development, Communication Openness, Discussion Efficiency, Task Discussion Effectiveness* and *Communication Clarity.*

Recall we defined shared understanding as the extent of common comprehension of a topic among team members (Ko et al., 2005; Mathieu et al., 2000). Therefore, this study establishes SU through comparing the individual understandings of team members (F. B. Tan & Hunter, 2002), and operationalizes SU through establishing the extent of similarity of answers in comprehension questions between team members. Accordingly, to measure SU we developed a set of 14 comprehension questions to evaluate the understanding of the compliance management domain (Shaft & Vessey, 2006; Shanks, 2008), and to allow us to establish the similarity of answers in each group. This set of questions allowed us to measure initial *Shared Understanding Development* i.e. before group members interact with each other during the group task.

To measure *Shared Understanding Improvement*, participants received a re-phrased version of the comprehension questions used to measure *Shared Understanding Development*. We followed this approach to maintain consistency of question complexity. We chose not to use the exact set of question twice because participants may have learnt or memorized the questions. Consequently, participants would be likely to associate the question with a previous one already answered.

The set of questions that we developed were extracted from the domain information represented in CoMOn and the ISO 19600:2014 standard. From such development, the questions are able to be answered by an understanding of the ontology (CoMOn) or the standard (ISO 19600:2014). Also, to ensure that we could rule out potentially conflicting explanations, we asked participants to identify "Which part of the documentation or what information allowed you to come to your answer". Table 4-4 shows the questions used to measure SU development and SU improvement.

To quantify SU, both development and improvement, this study considered the answers of the two participants in each group and uses the following rules to score points in the measure:

- When group members have the same answer, the measure receives a value of "1".
- When group members have a different answer, the measure receives a value of "0".

• When a group member does not know the answer to a question, the measure receives a value of "0".

No	Shared Understanding Development Questions	Improved Shared Understanding Questions
1.	Organizations implement a compliance	To enable compliance management, organizations
	management system to enable compliance	implement a compliance management system.
	management.	
2.	Organizations engage in compliance management to	To aid risk management, organizations engage in
	aid risk management.	compliance management.
3.	Organizations need to provide sufficient resources	To enable compliance management, organizations
	to enable compliance management.	need to provide sufficient resources.
4.	Organizations engage in compliance management to	To ensure organizations comply with obligations,
	ensure they comply with obligations.	they engage in compliance management.
5.	A compliance requirement (or mandatory	An example of an obligation type is a compliance
	obligation) of an organization is an example of an	requirement (or mandatory requirement).
	obligation type.	
6.	A compliance commitment (or voluntary	An example of an obligation type is a compliance
	obligation) of an organization is an example of an	commitment (or voluntary obligation).
7	obligation type.	To nonform a rich account on an institute should
7.	Organizations should identify, analyze and evaluate	identify analysis and evaluate risks
0	risks to perform a risk assessment.	Identify, analyze and evaluate risks.
8.	Monetary resources (or financial resources) and	Compliance management in organizations requires
	numan resources (non-monetary resources) are	human resources (non monetary resources) and
	organizations	numan resources (non-monetary resources).
9	Organizations through their compliance	The competency and training of employees are
	management systems ensure the competency and	ensured by the organization through their
	training of organization staff.	compliance management systems.
10.	Compliance management systems register and	Documented information of organizations is
	report documented information of organizations.	registered and reported by compliance management
		systems.
11.	Compliance management systems establish all	To check compliance in organizations, compliance
	controls necessary to check compliance in	management systems establish all the necessary
	organizations.	controls.
12.	A process (or business process) has related risks.	A process (or business process) includes related
		risks.
13.	The compliance culture of an organization specifies	The compliance culture of an organization describes
	behavioral norms.	behavioral norms.
14.	The treatment of a risk does not include a	The treatment of a risk does not consider a
	description of the applied treatment nor the risk.	description of the applied treatment nor the risk.

Table 4-4. Compliance Management Questions to Measure Share Understanding

Cross-understanding is defined as "the extent to which group members have an accurate understanding of another's mental model" (Huber & Lewis, 2010, p. 7). Accordingly, to measure CU development, we asked each participant their perception of answers that the other participant will give to the comprehension questions. The accuracy of those perceptions was then evaluated to establish CU. Table 4-5 shows the additional question that we added to each part in the second set of comprehension questions to help evaluate CU.

To quantify CU, this study considered the answers of the two participants in each group and used the following rules to score points in the measure:

- When a team member correctly recognizes the answer of the other team member, the question receives a value of "1". In this category, we also include the occurrence of one participant correctly answering that other person does not know the answer to a particular question.
- When a team member incorrectly recognizes the answer of the other team member, the question receives a value of "0".
- When a team member does not know what the other would respond, the measure receives a value of "0".

Question Statement	What do you think would be the answer of your partner?
Possible Answers	My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer

 Table 4-5. Additional Question to Measure Cross-Understanding Development

Recall *Communication Openness* is defined as the receptiveness of a group member to the communication of others (Lowry et al., 2009; O'Reilly & Roberts, 1977). Therefore, we adapted the measurement developed by Lowry et al. (2009) to measure this dependent variable. We re-stated questions 1 and 3 from Lowry et al. (2009)'s instrument in such a way that these questions make reference to a group of 2 people rather than larger groups. We did not consider question 5 from the original instrument because the communication was one-to-one instead of through a computer mediated communication (CMC). Table 4-6 shows our resulting measurement instrument for *Communication Openness*. Each question in the instrument was quantified using a seven-point Likert scale.

Table 4-6. Communication Openness Measurement Instrument

Question Statement
It was easy to communicate openly with my partner.
Communication in this group was very open.
When we communicated with each other in this group, there was a great deal of understanding.
It was easy to ask advice from my partner.

Recall that *Discussion Efficiency* reflects to what extent group members are oriented towards results, the effective use of time on interactions, and how thoroughly the issues are discussed (Lowry et al., 2009). We used the similar reflective measurement used in the studies of Lowry et al. (2009) and Davison (1997) to measure *Discussion Efficiency*. Table 4-7 shows the resulting measurement instrument for *Discussion Efficiency*. Each question in the instrument was quantified using a seven-point Likert scale.

Table 4-7. Discussion Efficiency Measurement Instrument

Question Statement To what extent would you agree that this group discussion was result oriented? The time spent in the group discussion was efficiently used. Issues raised in the group interaction were discussed thoroughly.

Task Discussion Effectiveness reflects to what extent group members participate in the discussion, develop discussion content, exchange information, and examine issues and ideas effectively and critically (Lowry et al., 2009). We modified slightly the formative measure used by Lowry et al. (2009) in order to make the questions clearer to the reader. Table 4-8 shows our resulting measurement instrument for *Task Discussion Effectiveness*. Similar to *Discussion Efficiency*, each question in the instrument was quantified using a seven-point Likert scale.

Table 4-8. Task Discussion Effectiveness Measurement Instrument

Question Statement
The discussions were effective.
The context of the discussions was carefully developed.
Issues were not examined effectively (reverse coded).
Participation in the discussions was consistently distributed.
Ideas in the discussions were critically examined.
The amount of information exchanged was sufficient.

Communication Clarity defines to what extent exchanged messages between group members are clear (Kahai & Cooper, 2003; Kayworth & Leidner, 2002). We adapted the reflective measurement used by Kahai and Cooper (2003) in order to make the questions clearer to the reader. Table 4-9 shows our resulting measurement instrument for Communication Clarity. Again, each question in the instrument was quantified using a seven-point Likert scale.

Question Statement
How clear were the messages of your partner?
How often were the messages of your partner unclear?

4.8.2 Control Variables

Our study includes a number of control variables for two main purposes. First, to check that our groups are formed homogenously, we obtained data on several control variables: *participant age, participant gender, type of student, area of study, year of study in current degree,* and *native language.* These variables were measured through multiple choice questions and is not able to be used to identify any specific participants. Table 4-10 shows the questions used to measure these variables.

Finally, to provide additional evidence that differences between groups stemmed from our treatment rather than from confounding factors (Bera et al., 2011), we obtained additional data on prior

conceptual modelling knowledge, and *compliance management knowledge*. Prior knowledge is used as a control variable because participants might try to use their prior knowledge to perform the task rather than using the provided information (Bera et al., 2011). Also, we considered these control variables because we wanted to check that our participants match those required in our experimental design; i.e. participants with a low level of compliance management and conceptual modelling. To measure *conceptual modelling knowledge*, we adapted the measurement instrument used by Bera et al. (2011) (See Table 4-11). *Conceptual modelling knowledge* was quantified using a seven-point Likert scale. To measure *compliance management knowledge*, we expanded the measurement instrument used by Bera et al. (2011) in order to ask more specific questions related to the compliance management domain (See Table 4-12). Similarly, each question in the instruments was quantified using a seven-point Likert scale.

	Possible Answers		
Particinant age	16-24		
i unicipant age	25 - 35		
	Above 35		
Participant gender	Male		
1 0	Female		
Type of student	Undergraduate		
••	Coursework postgraduate		
	Research postgraduate		
Area of study	Agribusiness, Agriculture, Environment and Science		
	Business, Economics and Law		
	Engineering, Architecture and Planning, and Information Technology, Health		
	Humanities, Education, Psychology and Music		
Year of study in	First year		
current degree	Second year		
	Third year		
	Fourth year		
	Fifth year		
Native language	English		
	Other		

Table 4-10. Control Variables to Check Homogeneity of Groups

Table 4-11. Conceptual Modelling Knowledge Measurement Instrument

Question Statement

What is the extent of your knowledge of data modelling concepts such as entities, classes, relationships, and properties? Over the last two years, to what extent do you have experience in using data modelling concepts such as entities, classes, relationships, and properties?

Table 4-12. Compliance Management Knowledge Measurement Instrument

Question Statement
What is the extent of your knowledge in compliance management concepts?
Over the last two years, to what extent do you have experience with applying compliance management concepts?
What is the extent of your knowledge of risk management concepts?
Over the last two years, to what extent do you have experience with applying risk management concepts?
What is the extent of your knowledge of Business Process Management concepts?

Over the last two years, to what extent do you have experience with applying Business Process Management concepts?

4.8.3 Construct Validity

Construct validity refers to the degree to which inferences can legitimately be made from the operationalizations in a study to the theoretical constructs on which those operationalizations were based (Trochim & Donnelly, 2008). Moreover, construct validity refers to generalizing from our measures to the concept of our measures (Trochim & Donnelly, 2008). Table 4-13 presents the threats to construct validity, their description and summarizes the mitigation measures that we undertook to prevent or reduce such threats.

	Threats	Description		Mitigation
1.	Inadequate preoperational explication of constructs	Inadequate operational definition of the construct.	•	We defined and operationalized all the latent constructs precisely – i.e. SU, CU, communication openness, discussion efficiency, task discussion effectiveness, communication clarity – including the specificity, nature, and the measurement scale of the constructs.
2.	Mono-operation bias	Using a single version of a treatment in a single place at a single point in time.	•	We used two treatments: a control group and a treatment group.
3.	Mono-method bias	Bias produced by measures or observations.	•	We implemented multiple measures of our constructs. Also, our pilot tests showed that measures we used behaved as we theoretically expect them to.
4.	Interaction of testing and treatment	Testing or measurement itself make the groups more sensitive or receptive to the treatment.	•	We had a control group and treatment group to reduce this threat. If there is such effect, both groups will be affected.
5.	Restricted generalizability across constructs	It occurs when observed effects (treatment X is effective) are generalized to other outcomes. Some negative outcomes could not be anticipated.	•	We did not generalize our results further than the definition of our constructs.
6.	Construct confounding and levels of constructs	The proposed construct does not represent the intended construct.	•	Communication quality constructs were adopted from prior studies. SU and CU are based on comprehension questions. This measurement instrument is commonly used in previous studies to measure understanding. Statistical analysis was conducted to examine whether the latent constructs achieve an adequate level of validity and reliability.

Table 4-13. Threats to Construct Validity and Their Mitigation (Trochim & Donnelly, 2008)

According to Table 4-13, there are 6 likely threats that could impact the construct validity of our experimental design. For each one of these threats we adopted a particular strategy to decrease the likely of occurrence of them. To follow we describe these threats and the mitigations adopted:

1. *Inadequate preoperational explication of constructs* threat occurs by an inadequate operational definition of the construct (Trochim & Donnelly, 2008). To avoid this threat, we defined and operationalized all the latent constructs precisely – i.e. SU, CU, communication
openness, discussion efficiency, task discussion effectiveness, communication clarity – including the specificity, nature, and the measurement scale of the constructs.

- 2. *Mono-operation bias* threat occurs for using a single version of a treatment in a single place at a single point in time (Trochim & Donnelly, 2008). To avoid this threat, we used two treatments: a control group and a treatment group.
- 3. *Mono-method bias* threat is produced by measures or observations. To avoid this threat, we implemented multiple measures of our constructs (Trochim & Donnelly, 2008). Also, our pilot studies showed that the measures we used behaved as we theoretically expect them to.
- 4. *Interaction of testing and treatment* threat occurs by testing or measurement itself making the groups more sensitive or receptive to the treatment (Trochim & Donnelly, 2008). To avoid this threat, we had a control group and treatment group. If there is such effect, both groups will be affected.
- Restricted generalizability across constructs threat occurs when observed effects (treatment X is effective) are generalized to other outcomes; hence some negative outcomes could not be anticipated (Trochim & Donnelly, 2008). To avoid this threat, we did not generalize our results further than the definition of our constructs.
- 6. *Construct confounding and levels of constructs* threat occurs when the proposed construct does not represent the intended construct (Trochim & Donnelly, 2008). To avoid this threat, *communication quality* constructs were adopted from prior studies. Also, SU and CU are based on comprehension questions. This measurement instrument is commonly used in previous studies to measure understanding. Additionally, a statistical analysis was conducted to examine whether the latent constructs achieved an adequate level of validity and reliability.

4.9 PROCEDURES

The different experimental sessions were conducted in a quiet computer lab in the School of Information and Electrical Engineering. We used a computer lab because our experimental materials and measurement instrument were implemented using Qualtrics application software⁴ (Appendix D shows print screens of the data collection instrument that we used). Accordingly, this software allows us to collect and store the data more easily compared to collection using physical documents. Also, it is important to mention that the computers we used had 21-inch screens to facilitate adequate visualization for participants.

⁴ Qualtrics is a web-based research application that allows users to design online surveys and experiments: <u>https://www.qualtrics.com</u>

Before the experiment began, the moderator asked participants to read the participation information sheet, and sign the consent form (to confirm willingness to participate) (See Appendix A.2). All the participants that attended the scheduled experiments signed the consent form without any hesitation. Once participants signed the consent form, the moderator provided a brief explanation of how the experiment is performed – this information was also provided on the participation information sheet. Then, unbeknownst to the participants the moderator assigned both to either the control or both to the treatment group, and allowed participants o access Qualtrics.

The experiment included five stages, namely *Control Variables, Training, Individual Task, Group Task, and Final Individual Task* (see Figure 4-4). First, during the *Demographic Data Collection*, participants had to answer a set of questions to measure the control variables described in the previous section.





In the second stage (Figure 4-5), *Practicing*, participants had a simple practice activity to familiarize themselves with the types of questions they had to answer later. We adapted both the materials and questionnaires used by Bera et al. (2011) to develop this practicing section. However, procedures were a little different for each treatment group. In the treatment group, participants had access to a training section that allowed them to understand the graphical representation of ontologies (See Appendix C.1). Then, participants received a travel domain ontology (See Appendix C.3) to facilitate the answering of the practice questions (See Appendix C.2). On the other hand, in the control group, participants did not have access to the training section. Participants received a textual description of the travel domain (See Appendix C.4) to facilitate the answering of the practice questions (See Appendix C.2). Appendix D.1 and D.2 show the data collection instruments used in the treatment group and control group respectively.



Figure 4-5. Practicing Stage

In the third stage (Figure 4-6), *Individual Task*, procedures for the treatment group and control group were the same, but each group received different materials. In the treatment group, participants received a digital and physical copy of the graphical representation of CoMOn. On the other hand, in the control group, participants received a digital and physical copy of the simplified version of the standard ISO 19600:2014. The hard copy was supplied to make it easier for the participant to refer back to the provided information. Then, participants in both the treatment and control group, were quizzed on their domain understanding through a set of questions defined in the section 4.6.1. All the questions were managed through Qualtrics. Appendix D.1 and D.2 show the data collection instruments used in the treatment group and control group respectively.



Figure 4-6. Individual Task Stage

In the fourth stage, *Group Task*, participants were asked to collaborate with their partner to perform a group task. As with the individual questions, the treatment group had the use of the ontology as a tool to aid their task progression, whilst the control group only had the standard of compliance management systems. The discussion process was audio recorded with permission for further analysis in the future. We designed a group task that allowed participants to maximize the exchange of information. As previously mentioned, in order to enhance group understanding constructs, it is necessary to improve the exchange of valuable information (Mathieu et al., 2000; Resnick, 1991; Stein et al., 2011). Therefore, we adapted our group task to suit the *MindMerger* process to improve this exchange of information between team members. The *MindMerger* process is a collaborative process that helps individuals in teams to develop SU (Bittner & Leimeister, 2014). Accordingly, during the group task, participants had to initially solve the problem individually using a source of information (the ontology or the simplified ISO 19600:2014 standard) and then they showed their partner their answers. During this process, participants had to see if there was any disagreement and discuss the disagreement in order to achieve an agreement. Appendix C.5 shows the instruction used during the group task. The information collected in this task was not formally measured or considered in our final variables; however, it will be used for future research.

In the final stage, *Final Individual Task*, participants were quizzed again, but this time questions about their partner's perception of the problem and solution were added. As with the individual questions, the treatment group had the use of the ontology as a tool to aid their task progression, whilst the control group only had the simplified standard of compliance management systems. The purpose of this approach is to gauge the participant's perception of the task from their own and from their partner's point of view to measure SU improvement and CU.

The required time for the whole experimental session was around 1.5 hours. At the end of the session each participant received AUD 50 in exchange for their participation.

4.10 ETHICAL CLEARANCE

This study includes human behavior as data to assess our theoretical model and our hypotheses. Most of the instrumental measures that we used considered self-reported measures. For example, questions related to what participants understand about compliance management (domain understanding) or the perceptions that participants have with regards to the understanding of other participants (CU) are associated with human behavior. Therefore, our study requires a risk and ethical assessment before any data collection is performed (Saunders, Lewis, & Thornhill, 2009). Additionally, ethical clearance is required to ensure our study complies with the University of Queensland's Ethical Conduct guidelines, including the characteristics of the required participants, experimental procedures, and data management.

Prior to conducting any data collection, including the pilot tests and main test, this study complied with the ethical guidelines of the University of Queensland. Ethical clearance was assessed and approved by the Ethical Committee of the School of Information Technology and Electrical Engineering on 26/08/2014. This ethical clearance was amended on 24/07/2015 and on 01/03/2016 to include some changes in recruitment process (See Appendix B to see the different approvals). We also had to increase the participation incentives to gain more participants.

Our study poses no feasible risk, and no risk that would exceed that which people experience as part of everyday life. Once participants read the study information sheet and agreed to participate, they had to sign the provided consent form before the experiment began. Participants, as noted in the information sheet, may withdraw at any time from the experiment without any consequence. Data was anonymously collected and the unidentified data can only be accessed by the research team. Any personal data relative to participants in this study is not disclosed in or relevant to the results of the study.

4.11 PILOT STUDIES

Prior to conducting the experiment, we conducted 4 pilots to test our experimental procedure, materials, required time for the experiment, and results on the dependent variables. In the first pilot, we recruited a sample of 10 undergraduate and postgraduate university students from a course of database fundamental offered in the School of Information Technology and Electrical Engineering at the University of Queensland. Participants were distributed in two different treatment groups. Each treatment evaluated a different representation of CoMOn, i.e. high quality representation and low quality representation. One group had 3 pairs of participants and the other group had 2 pairs of participants. We proceeded in this way because in the early stages of this research, we also wanted to explore the influence that representation quality has on SU and CU. This pilot specifically helped us to analyze the comprehension questions that we developed to test the understanding of the compliance management domain. Similarly, the pilot helped as to test our experimental procedures. As one of the results of this pilot, we noticed that our initial interaction procedure did not work at all. Thus, we needed to improve this procedure.

The second pilot allowed us to test a new set of questions and a new experimental procedure on the group task. For this pilot, we recruited 10 participants distributed randomly: 2 pair of participants for control group and 3 pairs of participants for the treatment group. Participants were also students from a course of database fundamental offered in the School of Information Technology and Electrical Engineering at the University of Queensland. The results of this pilot allow us to improve the quality of the experimental procedure and the group task. However, at this stage we decided not to analyze the effect of representation quality on SU and CU. We had to leave this analysis for future research because of time restrictions.

The third pilot that we ran, allowed us to test our new control group. We recruited 4 pairs of people: 2 pairs for the control group and 2 pairs for the treatment group. Participants were also students from a course of database fundamental offered in the School of Information Technology and Electrical Engineering at the University of Queensland. The results of this experiment allowed us to confirm that the control group had a good design. Also, we observed the main differences among the scores of the different variables between the control and the treatment, which showed good treatment manipulation.

On the fourth pilot, we only recruited 2 pair of people: one pair for the treatment and one pair for the control. The objective of this pilot was to evaluate whether written or verbalized responses are more appropriate to collect qualitative data regarding comprehension questions. We could observe that verbalization was very difficult to transcribed because not all participants had English as their first language. Thus, we decided only to consider the written responses.

At the end of the four pilots, we tuned our experimental design and procedures to the final ones.

4.12 CHAPTER SUMMARY

In this chapter, we explained the details of our research design used to address our research questions and to test our proposed hypotheses. To achieve this, first, we introduced and justified the use of the compliance management domain in our study. Secondly, we provided details of the ontology of the compliance management domain used during the experiments and its respective evaluation to guarantee its eligibility. Thirdly, we provided details and justification of the experimental design that were used in this study. Then, we specified the participants considered for the experiment and how they were recruited. After that, we detailed the different treatment materials that were used during the experiment. Once we defined our experimental settings, we provided details of the measurement instruments to measure dependent and control variables, and we explained how the experiment was performed. Finally, we provided details of the ethical clearances that were required before our study was conducted.

5 ANALYSIS OF RESULTS

In this chapter, we present the analysis of the results of the hypothesis testing. Accordingly, we perform a data screening of our results, we analyze the reliability of the measures, we test our hypotheses and discuss the results. Finally, we present an analysis of conclusion validity.

5.1 DATA SCREENING

Once we fine-tuned our experimental design, materials, and procedures, we performed our final data collection. For our final experiment, we recruited 39 participants. All the participants were paired before the experiment with the exception of our first participant, whose assigned partner did not arrive. Consequently, 19 groups were formed and tested. Assignment of participants to the treatment and control group was performed randomly based on availability the participants provided during the study recruitment process. Accordingly, we had eight pairs of participants in the control group and ten pairs of participants in the treatment group.

In addition to the one excluded participant we had to exclude data from a further two groups from the study. They were excluded for the following reasons; participants being unable to complete the task within the allocated time, lack of adherence to the designed tasks, and working individually rather than collaborating on the group tasks. Therefore, at the end of the experiments, we took into account data from 16 groups (8 for control and 8 for treatment), totaling 32 participant samples.

Before we tested our hypotheses, we checked that our data complied with the assumptions required to perform independent-sample t-test and linear regression analyses. This included checking for existence of outliers, normality, and linearity. First, we checked that our data did not have any significant outliers. We proceeded in this way because our sample is small and outliers may have affected our results in the independent-sample t-tests. Figure E-1 in Appendix E, shows the results of our outlier analysis. According to the results, there are two outliers in the scores of *Developed Shared Understanding* belonging to the treatment group. As we can observe on the boxplot of *Developed Shared Understanding*, those outliers are more than 1.5 box-lengths away from the edge of their box. Thus, we proceeded to eliminate those samples to avoid influences in our results. Therefore, our resulting data was 30 samples, 16 samples for no ontology use and 14 for ontology use.

Our second check was normality or normal distribution of the data. To check normality, we conducted a Shapiro-Wilk test (see Table F-1 in Appendix F). According to the results of the Shapiro-Wilk's test (Table E-1), *Communication Openness, Discussion Efficiency, Task Discussion Effectiveness* and

Communication Clarity were normally distributed for both non-ontology usage and ontology usage (p > 0.05). Contrarily, Developed Shared Understanding, Improved Shared Understanding, and Developed Cross-Understanding were not normally distributed in some of the experimental treatments. To determine if we still can consider this data without any normalization procedures, we analyzed the skewness and kurtosis of the data. We chose to further examine the skewness and kurtosis of the data as the data did not have scores close to zero, which could affect normality, and because the data may still be valid if the skewness and kurtosis is acceptable (Hayes, 2013). Table F-2 on Appendix F shows the results of the Z-scores for skewness and kurtosis. According to the data presented in Table E-2, all the dependent variables can be considered to be normally distributed for non-ontology and ontology groups (Z-scores were inside the range of - 2.58 to 2.58).

Finally, we performed a linearity check on our data. One of the requirements of linear regression is the existence of a linear relationship between the dependent and independent variable. To check this assumption, we visually inspected a scatterplot of the dependent variable plotted against the independent variable to see if a linear relationship exists. The scatterplots of the dependent variables used in this study are shown on Appendix E.2 and appear to meet the requirements for linearity.

After we verified that our data met the basic assumptions for our planned statistical tests, we collated our results. Table 5-1 presents the descriptive statistics for our control variables. According to Table 5-1, in terms of *age*, the control group (non-ontology use) had 12 participants with an age from 16 to 24 years old, and 4 participants with an age from 25 to 35 years old. While the treatment group had 13 participants with an age from 16 to 24 years old, and 1 participants with an age from 25 to 35 years old. Thus, both the control group and treatment group were homogenous in terms of *age*.

In terms of *gender* of participants, Table 5-1 shows that the control group had 10 male participants and 6 female participants. While the treatment group had 8 male participants and 6 female participants. Thus, both the control group and the treatment group were homogenous in terms of *participant gender* of participants.

In terms of *type of study* of participants, Table 5-1 shows that most participants were undergraduate students with 10 participants in the control group and 11 participants in the treatment group. The rest of participants were coursework postgraduate students: 6 participants for the control group and 3 participants for the treatment group. Thus, both the control group and the treatment group were homogenous in terms of *type of study* of participants.

		Non-ontology Usage			Ontology Usage		
	Scale	Ν	Mean	Std. Dev	Ν	Mean	Std. Dev.
	1 (16 – 24)	12			13		
Age	2 (25 - 35)	4	1.25	0.45	1	1.07	0.27
	3 (Above 35)	0			0		
Condor	1 (Male)	10	1 29	0.5	8	1 42	0.51
Gender	2 (Female)	6	1.58	0.5	6	1.45	0.51
	1 (Undergraduate)	10			11		
Type of Study	2 (Coursework postgraduate)	6	1.38	0.5	3	1.21	0.43
	3 (Research postgraduate)	0			0		
	1 (Agribusiness, Agriculture, Environment and Science)	4			4		3 1.02
Area of Study	2 (Business, Economics and Law)	1			1	2.43	
	3 (Engineering, Architecture and Planning, and Information Technology)	10	2.56	1.09	8		
	4 (Health)	0			1		
	5 (Humanities, Education, Psychology and Music)	1			0		Std. Dev. 0.27 0.51 0.43 1.02 1.16 0.5 0.91 0.91
	1 (First year)	14			12		
	2 (Second year)	2			0		
Year of Study	3 (Third year)	0	1.13	0.34	1	1.43	1.16
	4 (Fourth year)	0			0		
	5 (Fifth year)	0			1		
Native	1 (English)	8	15	0.52	9	1 26	0.5
Language	2 (Other)	8	1.5	0.52	5	1.50	0.5
Conceptual Model Knowledge	1 – 7		2.59	0.84		3.32	0.91
Compliance Management Knowledge	1-7		2.48	1.15		2.32	0.91

Table 5-1. Descriptive Statistics for the Control Variables

In terms of *area of study* of participants, Table 5-1 shows that most of the students were enrolled in the *Engineering, Architecture and Planning, and Information Technology* faculty. Specifically, 10 students were allocated in the control group and eight students were allocated in the treatment group. From the faculty of *Agribusiness, Agriculture, Environment and Science,* we had 4 students for both the control and the treatment group. We also had one student from the Business, Economics and Law faculty in the control group and one in the treatment group. Finally, we had one participant from the *Health* faculty for the treatment group. Thus, both the control group and the treatment group were homogenous in terms of *area of study* of participants.

In terms of *year of study* of participants, most participants were first year students. Fourteen participants were for the control group and 12 participants were for the treatment group. A group of two participants (second year) were part of the control group. While for the treatment group, we had

one participant in the third year and one participant in the fifth year. Thus, both the control group and the treatment group were homogenous in terms of *years of study* of participants.

In terms of *conceptual model knowledge* of participants, Table 5-1 shows that both the control group and the treatment group had similar means, 2.59 and 3.32 respectively. This knowledge was measured in a scale from 1 to 7. Thus, both the control group and the treatment group were homogenous in terms of *conceptual model knowledge* of participants.

In terms of *compliance management knowledge* of participants, Table 5-1 shows that both the control group and the treatment group had similar means, 2.48 and 2.32 respectively. This knowledge was measured in a scale from 1 to 7. Thus, both the control group and the treatment group were homogenous in their *compliance management knowledge*.

In summary, according to Table 5-1, scores for each of the control variables do not vary substantially between the experimental groups. This suggests that our randomization was effective and both groups are homogenous. In other words, we have similar groups for both experimental groups.

Table 5-2 presents the descriptive statistics for dependent variables. According to Table 5-2, means and scores for each of the dependent variables vary substantially between the experimental groups. Moreover, we see better results in the treatment group (ontology usage) than the control group (nonontology usage). For example, the means for *developed shared understanding* for the control group and treatment group were 6.63 and 9.57 respectively on a scale from 0 to 14. The means for *improved* shared understanding for the control group and treatment group were 8.13 and 12 respectively on a scale from 0 to 14. The means for *developed cross-understanding* for the control group and treatment group were 7.94 and 10.64 respectively on a scale from 0 to 14. Also, the means for *communication* openness were 4.88 and 5.43 for the control group and the treatment group respectively on a scale from 1 to 7. For discussion efficiency, the means were 4.58 and 5.48 for the control group and treatment group respectively on a scale from 1 to 7. Similarly, the means for *communication clarity* were 4.88 and 5.71 on a scale from 1 to 7. For task discussion effectiveness, the means were 4.31 and 5.04 for the control group and treatment group respectively on a scale from 1 to 7. Similarly, the means for communication clarity were 4.88 and 5.71 on a scale from 1 to 7. Finally, for shared understanding improvement, the means were 1.50 and 2.43 for the control group and treatment group respectively on a scale from 1 to 7.

In summary, considering that the control variable scores do not vary substantially (See Table 5-1), but the dependent variables do (See Table 5-2), we can preliminarily consider that manipulation of

the ontology use was successful. Also, means in the treatment group were higher than means in the control group, which is at least partly in line with our hypothesis.

		Non-Onto	ology Usage	Ontolo	ogy Usage
	Scale	Mean	Std. Dev	Mean	Std. Dev
Developed Shared Understanding	0 - 14	6.63	3.42	9.57	1.65
Improved Shared Understanding	0 - 14	8.13	2.99	12.00	1.36
Developed Cross-Understanding	0 - 14	7.94	3.43	10.64	2.21
Communication Openness	1 - 7	4.88	1.06	5.43	1.13
Discussion Efficiency	1 - 7	4.58	1.05	5.48	1.17
Task Discussion Effectiveness	1 - 7	4.31	1.12	5.04	1.17
Communication Clarity	1 - 7	4.88	0.94	5.71	1.09
Shared Understanding Improvement	0 - 14	1.50	1.93	2.43	1.65

 Table 5-2. Descriptive Statistics for the Dependent Variables

Table 5-3 presents our correlation matrix. There are some important points to note in this table. First, *Developed Shared Understanding, Developed Cross-Understanding, Discussion Efficiency,* and *Communication Clarity* are correlated significantly with the *Domain-Ontology Usage*, as per our predictions. However, we can see that *Shared Understanding Improvement, Communication Openness* and *Task Discussion Effectiveness* are not correlated significantly with the *Domain-Ontology Usage*, contrary to our predictions. Finally, we also note that *Shared Understanding Improvement* is correlated with most of the Communication Quality factors with exception of *Communication Clarity*; and *Developed Cross-Understanding* is correlated significantly with all the Communication Quality factors. This also is at least partly in line with our hypotheses. We examine these results further below.

		1	2	2	4	F	6	7	0	0	10	11	10	12	14	15	16	17
		1	2	3	4	3	6	1	8	9	10	11	12	13	14	15	16	1/
1	Domain Ontology Usage	1																
2	Age	-0.24	1															
3	Gender	0.05	0	1														
4	Type of Study	-0.17	0.49**	0.06	1													
5	Area of Study	-0.07	-0.22	0.13	-0.25	1												
6	Year of Study	0.19	-0.15	-0.02	-0.13	-0.2	1											
7	Native Language	-0.14	-0.03	0.11	0.46*	0.16	-0.04	1										
8	Conceptual Model Knowledge	0.39*	-0.16	0.24	-0.07	0.23	0.05	-0.01	1									
9	Compliance Management Knowledge	-0.08	0.37*	0.03	0.04	0.45*	-0.29	-0.22	0.3	1								
10	Developed Shared Understanding	0.48**	-0.27	-0.02	-0.58**	0	-0.12	-0.60**	0.14	0.13	1							
11	Improved Shared Understanding	0.64**	-0.14	0.02	-0.47**	-0.08	0.01	-0.47**	0.27	0.06	0.82**	1						
12	Shared Understanding Improvement	0.26	0.21	0.07	0.19	-0.13	0.22	0.22	0.21	-0.12	-0.32	0.28	1					
13	Developed Cross-Understanding	0.43*	0.11	-0.16	-0.23	-0.12	-0.1	-0.42*	0.21	0.11	0.53**	0.76**	0.37*	1				
14	Communication Openness	0.25	0.11	0.18	0.25	0	0.13	0.06	0.32	0.14	0.07	0.32	0.42*	0.49**	1			
15	Discussion Efficiency	0.38*	0.03	0.27	0.25	-0.01	0.21	0.02	0.42*	0.09	0.08	0.36	0.46*	0.45*	0.87**	1		
16	Task Discussion Effectiveness	0.31	-0.11	0.27	0.21	-0.22	0.12	0.11	0.28	-0.14	0.11	0.40*	0.48**	0.52**	0.82**	0.85**	1	
17	Communication Clarity	0.39*	-0.24	0.18	-0.06	-0.12	0.11	-0.03	0.17	-0.22	0.17	0.34	0.28	0.38*	0.59**	0.58**	0.64**	1

 Table 5-3. Constructs Correlations

* Correlation is significant at the 0.05 level (2-tailed).

** Correlation is significant at the 0.01 level (2-tailed).

5.2 RELIABILITY OF MEASURES

As we mentioned before, the pilots that we performed allowed us to validate the procedures and to observe whether *shared understanding* (SU) and *cross-understanding* (CU) can be achieved and measured appropriately. However, because of the changes made to the experimental design resulting from the first pilot studies, we re-worded the set of comprehension questions to measure domain understanding in our final pilot study. We proceeded in this way to collect both 'True/False' answers and qualitative answers instead of just qualitative answers as our previous designs. The modifications of these statements in the comprehension question could introduce problems of understanding during question reading in our domain understanding measurement instrument. Unfortunately, the low number of participants in our last pilot (4 participants) did not allow us to unequivocally check the reliability of our instrument.

The rest of measurement instruments were taken from previous studies (Davison, 1997; Kahai & Cooper, 2003; Lowry et al., 2009), in which reliability and validity were already evaluated. Nevertheless, we analyzed if the reliability is retained for this study.

We examined average Cronbach's alpha values for all the multi-item constructs used in the experiment to assess their reliability. Table 5-4 show our results. According to Table 5-4, all the measures are over 0.7, which are regarded as satisfactory (Bland & Altman, 1997). We therefore concluded that the instruments are reliable and began testing our hypothesis.

	Cronbach's Alpha	No of Items
Domain Understanding	0.76	14
Improved Domain Understanding	0.71	14
Communication Openness	0.85	4
Discussion Efficiency	0.74	3
Task Discussion Effectiveness	0.89	6
Communication Clarity	0.76	2

Table 5-4. Reliability of Dependent Variables

5.3 HYPOTHESES TESTING

In the following sub-sections, we present the tests that we performed to test each of the dependent variables and discuss how the results support or refute our hypotheses. Figure 5-1 re-iterates our theoretical model and the hypotheses to be tested.





5.3.1 Test of Developed Shared Understanding

Recall our first hypothesis is related to the impact of domain ontology use on developed shared understanding:

H1. Group members who use a domain ontology develop higher levels of shared understanding than those who do not use a domain ontology.

Accordingly, the aim of our first test is to examine if team members who use a domain ontology develop higher levels of SU than those who do not use a domain ontology. To test this, we conducted an independent-samples t-test with *Developed Shared Understanding* as the dependent measure and *Domain-Ontology Usage* as independent variable. Table 5-5 shows the mean scores (*M*), the number of samples (N), and the standard errors (*SE*) for *Developed Shared Understanding*. Table 5-6 shows the results of the independent-samples t-test for this variable.

1 able 5-5. De	escriptive Statist	lics for Develo	ped Snared Unde	erstanding

Experimental Group		Ν	М	Std. Error Mean
Developed Shared Understanding	Ontology	14	9.57	0.44
	Non-Ontology	16	6.63	0.86

Table 5-6. Independent-Samples	t-test Results for Developed	Shared Understanding
--------------------------------	------------------------------	----------------------

						95% Confidence	e Interval
	t	df	Sig. (2-tailed)	Mean Difference	Std. Error Difference	Lower	Upper
Developed Shared Understanding	3.06	22.22	0.01	2.95	0.96	0.95	4.94

According to the results on Table 5-5 and Table 5-6, on average, participants given a domain ontology developed a better *shared understanding* (M = 9.57, SE = 0.44), than those not given a domain ontology (M = 6.63, SE= 0.86). This difference of 0.96 (95% CI, 0.95 to 4.94), was significant t (22.2) = 3.06, p = 0.01. According to this result, H1 is supported fully and participants who used a domain ontology developed higher level of *shared understanding* than those who did not. In other words, we can use a domain ontology – providing it has acceptable quality – to facilitate the development of shared understanding among group members before they have an interaction or communication between them. As we theorized, this effect is produced because conceptual models facilitate the development of domain understanding (Bera et al., 2011; Burton-Jones & Meso, 2008). Thus, when the same domain ontology is used as an information tool by group members without interaction, they are likely to gain a similar level of domain understanding, which leads to SU development.

5.3.2 Test of Communication Quality

Our second hypothesis is related to the impact of domain-ontology usage on communication quality:

H2. Group members who use a domain ontology engage in better communication quality in terms of communication openness, discussion efficiency, task discussion effectiveness, and communication clarity than those who do not use a domain ontology.

Accordingly, the objective of this test is to measure whether the use of a domain ontology facilitates better *Communication Quality* in terms of *Communication Openness, Discussion Efficiency, Task Discussion Effectiveness*, and *Communication Clarity* as compared to situations in which a domain ontology in not used. To test this, we conducted an independent-samples t-test having *Communication Openness, Discussion Efficiency, Task Discussion Effectiveness, and Communication Effectiveness, and Communication Clarity* as dependent measures and *Domain-Ontology Usage* as independent variable. Table 5-7 shows the mean scores (*M*), the number of samples, and the standard errors (*SE*) for the dependent variables. Table 5-8 shows the results of the independent-samples t-test for these variables.

	Experimental Group	Ν	Mean	Std. Error Mean
Communication Openness	Ontology	14	5.43	0.30
	Non-Ontology	16	4.88	0.26
Discussion Efficiency	Ontology	14	5.48	0.31
	Non-Ontology	16	4.58	0.26
Task Discussion Effectiveness	Ontology	14	5.04	0.31
	Non-Ontology	16	4.31	0.28
Communication Clarity	Ontology	14	5.71	0.29
	Non-Ontology	16	4.88	0.23

Table 5-7. Descriptive Statistics for Communication Quality Factors

 Table 5-8. Independent-Samples t-test Results for Communication Quality Factors

						95% Confidence Interv of the Difference	
	t	df	Sig. (2-tailed)	Mean Difference	Std. Error Difference	Lower	Upper
Communication Openness	1.39	26.89	0.18	0.55	0.40	-0.26	1.37
Discussion Efficiency	2.21	26.44	0.04	0.89	0.40	0.06	1.72
Task Discussion Effectiveness	1.73	27.14	0.10	0.72	0.42	-0.13	1.58
Communication Clarity	2.27	25.95	0.03	0.84	0.37	0.08	1.60

According to the results in Table 5-7 and Table 5-8, on average, participants given a domain ontology developed better *Communication Openness* (M = 5.43, SE = 0.30), than those not given a domain ontology (M = 4.88, SE= 0.26) however this difference, 0.55 (95% CI, -0.26 to 1.37), was not statistically significant *t* (26.89) = 1.39, *p* = 0.18. This result means that the use of a domain ontology does not enhance *Communication Openness* during the communication between group members. In other words, group members are not more open in exchanging information or more receptive to the communication of others purely by using a domain ontology during their task and communication.

Second, on average, participants given a domain ontology developed better *Discussion Efficiency* (M = 5.48, SE = 0.31), than those not given a domain ontology (M = 4.58, SE= 0.26). This difference, 0.89 (95% CI, 0.06 to 1.72), was statistically significant t (26.44) = 2.21, p = 0.04. This means that the use of a domain ontology enhances the *Discussion Efficiency* during communication between group members, as we theorized. In other words, when using a domain ontology during the communication, group members are more results-oriented, they spend time on interactions effectively, their interactions are more meaningful, and they discuss issues thoroughly.

Third, on average, participants given a domain ontology developed better *Task Discussion Effectiveness* (M = 5.04, SE = 0.31), than those not given a domain ontology (M = 4.31, SE= 0.28). This difference, 0.72 (95% CI, -0.13 to 1.58), was not statistically significant t (27.14) = 1.73, p = 0.10. This means that the use of a domain ontology does not enhance *Task Discussion Effectiveness* during communication between group members. In other words, by using a domain ontology during a group task, team members do not participate more in the discussion, do not develop better discussion

content, do not exchange more information, and do not examine issues and ideas more effectively or clearly.

Finally, on average, participants given a domain ontology developed better *Communication Clarity* (M = 5.71, SE = 0.29), than those not given a domain ontology (M = 4.88, SE = 0.23). This difference, 0.84 (95% CI, 0.08 to 1.60), was statistically significant t (25.95) = 2.27, p = 0.03. This means that the use of a domain ontology enhances *Communication Clarity* during communication between group members, as we theorized. That is, by using a doming ontology during communication between group members, the exchanged messages between them are more clear.

In summary, our results support partially hypothesis H2. Moreover, the use of a domain ontology significantly facilitates better *Discussion Efficiency* and *Communication Clarity*. However, contrary to our view, *Communication Openness* and *Task Discussion Effectiveness* are not facilitated significantly through the use of a domain ontology. These results are to be considered in the analysis of *Shared Understanding Improvement* and *Developed Cross-Understanding* to follow.

5.3.3 Testing of Shared Understanding Improvement

Recall our third and fourth hypotheses are related to shared understanding improvement:

H3. Group members who use a domain ontology have greater shared understanding improvement than those who do not use a domain ontology.

H4. Communication quality in terms of communication openness, discussion efficiency, task discussion effectiveness, and communication clarity mediates the effect of domain ontology usage on the improvement of shared understanding.

Accordingly, for *Shared Understanding Improvement*, we performed two tests to assess H3 and H4. First, to test H3, we measured whether there is a significant difference in *Shared Understanding Improvement* between participants who use a domain ontology and those who do not use a domain ontology. Table 5-9 shows the mean scores (*M*), the number of samples, and the standard errors (*SE*) for *Shared Understanding Improvement*. Whilst, Table 5-10 shows the results of the independent-samples t-test.

Experimental GroupNMeanStd. Error MeanShared Understanding ImprovementOntology142.430.44Non-Ontology161.500.48

Table 5-9. Descriptive Statistics for Shared Understanding Improvement

						95% Confidence of the Diffe	e Interval
	t	df	Sig. (2-tailed)	Mean Difference	Std. Error Difference	Lower	Upper
Shared Understanding Improvement	1.40	28	0.17	0.93	0.66	-0.43	2.28

Table 5-10. Independent-Samples t-test Results for SU Improvement

According to the results in Table 5-9 and Table 5-10, on average, participants given a domain ontology had a greater *Shared Understanding Improvement* (M = 2.43, SE = 0.44), than those not given a domain ontology (M = 1.50, SE= 0.48). This difference, 0.93 (95% CI, -0.43 to 2.28), was not statistically significant t (28) = 1.40, p = 0.17; however, it did represent a medium-size effect, d = 0.62 (Cohen, 1992). According to this result, our H3 is not supported. This means that the use of a domain ontology does not improve shared understanding significantly during the group task. However, the results are still important in practice when shared understanding is required to be improved because of the medium-size effect (Cohen, 1992). Considering that the interaction group task was limited in time in our experimental setting, perhaps a longer activity or more interactions would be required in order to get a significant effect – e.g. see the work of He et al. (2007). Our fourth hypothesis, H4, deals with communication quality in terms of *Communication Openness, Discussion Efficiency, Task-discussion Effectiveness*, and *Communication Clarity* mediating the improvement of *shared understanding*. To test this hypothesis, we will use mediation analysis.

Mediation refers to a situation in which the relationship between a predictor variable and an outcome variable can be explained by their relationship with a third variable, the mediator (Field, 2013). Explicitly, our predictor variable is *Domain Ontology Usage*, our outcome variable is *Shared Understanding Improvement*, and our mediator is *Communication Quality*. In our case, *Communication Quality* – through the four different factors – mediates the effect of *Domain Ontology Usage* on *Shared Understanding Improvement*. To facilitate the analysis of mediation, we used the PROCESS⁵ (Hayes, 2013) custom dialog box. To estimate mediation, PROCESS computes the indirect effect and its significance or the confidence intervals for the indirect effect using bootstraps methods – see the schematic of a mediation model in Figure 5-2 (Field, 2013).

⁵ PROCESS is an add-on for SPSS and SAS for statistical mediation, moderation, and conditional process analysis: <u>http://processmacro.org/</u>



Figure 5-2. Diagram of a Mediation Model (Field, 2013)

We ran mediation analysis with PROCESS on SPSS considering *Domain Ontology Usage* as the predictor variable, the four factors of communication quality as mediators, and *Shared Understanding Improvement* as the outcome. The complete results of the analysis can be observed in Appendix G.1.

Table 5-11 summarizes the total, direct and indirect effects of *Domain Ontology Usage* on *Shared Understanding Improvement* presented in Appendix G.1. According to Table 5-11, *Communication Quality* – through the four different factors – does not significantly mediate the effect of *Domain Ontology Usage* on *Share Understanding Improvement*. First, there was not a significant effect of *Domain Ontology Usage* on *Shared Understanding Improvement*. First, there was not a significant effect of *Domain Ontology Usage* on *Shared Understanding Improvement* as mediated by *Communication Openness*, *b* = 0.00, 95% BCa CI (-1.12 to 0.77). Nor was there a significant effect via *Discussion Efficiency*, *b* = 0.22, 95% BCa CI (-1.06 to 1.94), *Task Discussion Effectiveness*, *b* = 0.41, 95% BCa Ci (-0.81 to 1.69), or *Communication Clarity*, *b* = 0.13, 95% BCa Ci (-1.81 to 0.33). Thus, we then analyzed if any of the factors added a negative effect to our results.

	Tota	al effect (of X on Y									
Effect		SE	t	р	LLCI	ULCI						
0	.93	0.68	1.37	0.18	-0.46	2.32						
	Direct effect of X on Y											
Effect		SE	t	р	LLCI	ULCI						
0	.43	0.88	0.49	0.63	-1.39	2.25						
Indirect effect of X on Y												
		Effect	Boot SE	BootLLCI	BootULCI							
Total		0.50	0.59	-0.53	1.90							
Communication Openness (CO)		0.00	0.44	-1.12	0.77							
Discussion Efficiency (DE)		0.22	0.72	-1.06	1.94							
Task Discussion Effectiveness (TI	DE)	0.41	0.68	-0.81	1.69							
Communication Clarity (CC)		-0.13	0.46	-1.81	0.33							

Table 5-11. Total, Direct and Indirect Effect on Shared Understanding Improvement (Four Factors)

From the previous analysis of Communication Quality (See section 5.3.2), we confirmed that *Communication Openness* and *Task Discussion Effectiveness* are not significantly facilitated through the use of a domain ontology. Thus, we only considered *Discussion Efficiency* and *Communication Clarity* as *Communication Quality* factors for the mediation analysis. The full results of the analysis can be observed in Appendix G.2.

Table 5-12 summarizes the total, direct and indirect effects of *domain ontology usage* on *Shared Understanding Improvement* with respect to *Discussion Efficiency* and *Communication Clarity*. According to these results, there was a significant indirect effect of *Domain Ontology Usage* on *Shared Understanding Improvement* through *Discussion Efficiency*, b = 0.60, 95% BCa Ci (0.22 to 1.67). However, there was not a significant indirect effect of *Domain Ontology Usage* on *Shared Understanding Improvement* through *Communication Clarity*, b = -0.01, 95% BCa Ci (-1.05 to 0.51). In other words, when a *domain ontology* is used by group members as an information tool during communication to improve *shared understanding*, this improvement is mediated by the extent of *Discussion Efficiency* of the group.

Table 5-12. Total, Direct and Indirect Effect on Shared Understanding Improvement (Two Factors)

Total effect of X on Y					
Effect	SE	t	р	LLCI	ULCI
0.93	0.68	1.37	0.18	-0.46	2.32
	Direct effect of X on Y				
Effect	SE	t	р	LLCI	ULCI
0.34	0.82	0.42	0.68	-1.34	2.03
Indirect effect of X on Y					
	Effect	Boot SE	BootLLCI	BootULCI	
Total	0.59	0.49	-0.17	1.85	
DE	0.60	0.41	0.22	1.67	
CC	-0.01	0.37	-1.05	0.51	

Consequently, we consider H4 to be partially supported in that *Discussion Efficiency* significantly mediates the improvement of *shared understanding* but the remaining three *Communication Quality* factors do not.

5.3.4 Test of Cross-Understanding

Similarly, to the previous analysis, we performed two tests to assess H5 and H6:

H5. Group members who use a domain ontology develop greater cross-understanding than those who do not use a domain ontology.

H6. Communication quality in terms of communication openness, discussion efficiency, taskdiscussion effectiveness, and communication clarity mediates the effect of domain ontology usage on the development of cross-understanding.

Accordingly, to test hypothesis H5, we measured whether there is a significant difference in *Developed Cross-Understanding* between participants who use a domain ontology and those who do not. Table 5-13 shows the mean scores (M), the number of samples (N), and the standard errors (SE) for *Developed Cross-Understanding*. Table 5-14 shows the results of the independent-samples t-test.

	1			•			8	
		Expei	rimental Grou	up N	Mean	Std. Err	or Mean	_
Developed Cross-Underst	anding	Ontol	ogy	14	10.64		0.59	
		Non-0	Ontology	16	7.94		0.86	_
Table 5-14. Independent	-Sampl	les t-test	Results for	Develope	ed Cros	s-Under	standing	
	t	df	Sig. (2-tailed)	Mean Differen	Sto ce Dif	l. Error fference	Lower	Upper
Developed Cross-Understanding	2.60	25.84	0.02	2.7	71	1.04	0.56	4.85

Table 5-13. Descriptive Statistics for Developed Cross-Understanding

According to the results in Table 5-13 and Table 5-14, on average, participants given a domain ontology had a greater *Developed Cross-Understanding* (M = 10.64, SE = 0.59), than those not given a domain ontology (M = 7.94, SE= 0.86). This difference, 2.71 (95% CI, 0.56 to 4.85), was statistically significant t (25.84) = 2.60, p = 0.02. Based on this result, H5 is fully supported. This means that the use of a domain ontology enhances the development of *cross-understanding* after team members communicate with each other. In other words, the ontology use leads to the exchange of more precise information about the domain, which allows group members to understand what the other group members know or think about the domain.

To test hypothesis H6, we ran a mediation analysis with PROCESS on SPSS considering *Domain-Ontology Usage* as the predictor, the four factors of *Communication Quality* as mediators, and *Developed Cross-Understanding* as the outcome. The complete results of the analysis are provided in Appendix G.3.

Table 5-15 summarizes the total, direct and indirect effects of *Domain Ontology Usage* on *Shared Understanding Improvement* presented in Appendix G.3. According to Table 5-15, *Communication Quality* – through the four different factors – does not mediate the effect of *Domain Ontology Usage* on *Developed Cross-Understanding*. There was no significant effect of *Domain Ontology Usage* on *Developed Cross-Understanding* through *Communication Openness*, b = 0.62, 95% BCa CI (-0.40 to 3.85), nor through *Discussion Efficiency*, b = -0.84, 95% BCa CI (-3.81 to 0.60), *Task Discussion Effectiveness*, b = 0.81, 95% BCa Ci (-0.67 to 4.17) or through *Communication Clarity*, b = -0.13,

95% BCa Ci (-2.39 to 1.38). Thus, we analyzed if any of the factors added a negative effect to our results as explained in the following.

Tot	al effect of	of X on Y			
Effect	SE	t	р	LLCI	ULCI
2.71	1.08	2.51	0.02	0.50	4.91
Dire	ect effect	of X on Y			
Effect	SE	Т	р	LLCI	ULCI
2.42	1.34	1.67	0.11	-0.52	5.00
Indir	ect effect	of X on Y			
	Effect	Boot SE	BootLLCI	BootULCI	
Total	0.46	0.90	-1.15	2.56	
Communication Openness (CO)	0.62	1.02	-0.40	3.85	
Discussion Efficiency (DE)	-0.84	1.09	-3.81	0.60	
Task Discussion Effectiveness (TDE)	0.81	1.16	-0.67	4.17	
Communication Clarity (CC)	-0.13	0.88	-2.39	1.38	

Table 5-15. Total, Direct and Indirect Effect on Developed Cross-Understanding

Performing a similar subsequent analysis as we did for *Shared Understanding Improvement*, we only considered the *Communication Quality* factors that were improved through the use of a domain ontology – i.e. *Discussion Efficiency* and *Communication Clarity*. Therefore, we performed the second mediation analysis only considering *Discussion Efficiency* and *Communication Clarity*. The complete results of this analysis are provided in Appendix G.4.

Table 5-16 summarizes the total, direct and indirect effects of Domain Ontology Usage on Developed Cross-Understanding. According to these results, there was a significant indirect effect of Domain Ontology Usage on Developed Cross-Understanding through Discussion Efficiency, b = 0.69, 95% BCa Ci (0.02 to 2.58). However, there was no significant indirect effect of Domain Ontology Usage on Developed Cross-Understanding through Communication Clarity, b = 0.25, 95% BCa Ci (-0.69 to 2.11). Consequently, H6 is partially supported in that Discussion Efficiency significantly mediates the development of cross-understanding but the remaining three Communication Quality factors do not. In other words, when a domain ontology is used by group members as an information tool during communication between them to develop cross-understanding, this development is mediated by the extent of Discussion Efficiency of the group.

Total effect of X on Y						
Effect	SE	t	р	LLCI	ULCI	
2.71	1.08	2.51	0.02	0.5	4.91	
	Direct effect of X on Y					
Effect	SE	t	р	LLCI	ULCI	
1.76	1.18	1.49	0.15	-0.67	4.19	
	Indirect effect of X on Y					
	Effect	Boot SE	BootLLCI	BootULCI		
Total	0.94	0.69	0.00	2.84		
DE	0.69	0.55	0.02	2.58		
CC	0.25	0.65	-0.69	2.11		

 Table 5-16. Total, Direct and Indirect Effect on Developed Cross-Understanding

To summarize, we now present a synopsis of the outcomes of hypotheses testing in Table 5-17.

Table 5-17.	Summarv	of	Hypothesis	Testing	Results
14010 0 17.	Summary	UI.	rypoincois	1 coung	itebuiteb

Hypothesis	Support
H1. Group members who use a domain ontology develop higher levels of shared understanding than those who do not use a domain ontology.	Full
H2. Group members who use a domain ontology engage in better Communication Quality in terms of Communication Openness, Discussion Efficiency, Task Discussion Effectiveness, and Communication Clarity than those who do not use a domain ontology.	Partially, only Discussion Efficiency and Communication Clarity
H3. Group members who use a domain ontology have greater shared understanding improvement than those who do not use a domain ontology.	No
H4. Communication Quality in terms of Communication Openness, Discussion Efficiency, Task-Discussion Effectiveness, and Communication Clarity mediates the effect of domain ontology usage on the improvement of share understanding.	Partially, only Discussion Efficiency
H5. Group members who use a domain ontology develop greater cross-understanding that those who do not use a domain ontology.	Full
H6. Communication quality in terms of Communication Openness, Discussion Efficiency, Task-Discussion Effectiveness, and Communication Clarity mediates the effect of domain ontology usage on the development of cross-understanding.	Partially, only Discussion Efficiency

5.4 CONCLUSION VALIDITY

Conclusion validity is the degree to which conclusions we reach about relationships in the data are reasonable (Trochim & Donnelly, 2008). Conclusion validity differs from internal validity (see Section 4.3.2) in that it is only concerned with whether there is a relationship and whether that relationship is a reasonable one or not, given the data (Trochim & Donnelly, 2008). Conversely, internal validity is the approximate truth about inferences regarding cause-effect or causal relationship (Trochim & Donnelly, 2008). Table 5-18 presents several threats to conclusion validity, their description, and the controls that we undertook to mitigate such threats.

	Threats	Description		Mitigation in This Study
1.	Low reliability of	Measures are not consistent or	٠	We measured reliability of variables.
	measures	cannot be repeated.		-
2.	Poor reliability of	Lack of standardized treatments in	٠	Four pilot studies were performed to check
	treatment	an experiment can lead to erroneous		our treatments were standardized.
	implementation	conclusions.		
3.	Low Statistical Power	Conclusions drawn from a study are	•	Sample sizes were very similar in the two
		based on insufficient statistical		treatments.
		power.	•	Participants were assigned randomly.
			٠	We implemented control variables.
4.	Violated Assumptions	Violation of assumptions required	•	We ensured statistical assumptions held for
	of Statistical Tests	by particular statistical techniques.		the t-test and linear regression.
5.	Fishing and the Error	Undertaking repeated testing for	•	No repeated tests were taken.
	Rate Problem	significant relationships.		-
6.	Restriction of Range	Lack of standardized treatments in	•	The used treatments have equivalence of
		an experiment can lead to erroneous		information.
		conclusions.		
7.	Extraneous Variance	Extraneous variance (distractions,	•	We used a computer lab, which was free of
	in the Experimental	noise) affecting participants in		external interruptions, to reduce this threat.
	Setting	experiments may cause inaccuracy		
		of statistical conclusions.		
8.	Random	Error increases due to participants'	٠	We used students in a database course. Their
	Heterogeneity of	heterogeneity.		heterogeneity was measured and controlled
	respondents			by our control variables.

Table 5-18. Threats to Conclusion Validity and their Mitigation (Trochim & Donnelly, 2008)

As per Table 5-18, there are 8 likely threats that could impact the conclusion validity of our experimental results. For each one of these threats we adopted strategies to decrease their likely occurrence. In the following, we describe these threats and the mitigations adopted:

- Low reliability of measures occurs when measures are not consistent or cannot be repeated (Trochim & Donnelly, 2008). To avoid this treat, we measured reliability of variables and ensured repeatability of the experiment.
- 2. *Poor reliability of treatment implementation* occurs when a lack of standardized treatments in an experiment leads to erroneous conclusions (Trochim & Donnelly, 2008). To avoid this threat, four pilots were performed to ensure that our treatment and control are standardized.
- 3. *Low Statistical Power* occurs when conclusions drawn from a study are based on insufficient statistical power (Trochim & Donnelly, 2008). To avoid this treat, sample sizes were similar in the treatment and the control. Also, participants were assigned randomly. Finally, we implemented control variables.
- 4. *Violated Assumptions of Statistical Tests* occurs by the violation of assumptions required by particular statistical techniques (Trochim & Donnelly, 2008). To avoid this threat, we ensured statistical assumptions held for the t-test and linear regression.
- 5. *Fishing and the Error Rate Problem* occurs when repeated testing for significant relationships is undertaken (Trochim & Donnelly, 2008). To avoid this threat, no repeated tests were carried out.

- 6. *Restriction of Range* occurs when a lack of standardized treatments in an experiment can lead to erroneous conclusions (Trochim & Donnelly, 2008). To avoid this threat, we used treatments that have equivalence of information.
- 7. *Extraneous Variance in the Experimental Setting* occurs when extraneous variance (distractions, noise) affecting participants in experiments may cause inaccuracy of statistical conclusions (Trochim & Donnelly, 2008). To reduce this threat, we used a computer lab, which was free of external interruptions.
- 8. *Random Heterogeneity of respondents* occurs when error increases due to participants' heterogeneity (Trochim & Donnelly, 2008). To avoid this threat, we used students in a database course. Their heterogeneity was measured and controlled by our control variables.

5.5 CHAPTER SUMMARY

This chapter provided the results of the empirical study performed to test our theoretical model of the effect of *Domain-Ontology Usage* on *shared understanding* (SU) and *cross-understanding* (CU). To achieve this objective, first, we screened the data to establish that the data is adequate for our statistical tests. Second, we ensured the reliability of our measurement instruments. Then we tested our six hypotheses through independent-samples t-tests and mediation analysis. The results of our tests allow us to fully or partially confirm all but one of our hypotheses. Finally, we analyzed the conclusion validity of our results and found our results to be satisfactory. In the next chapter, we will provide more insights into the meaning and importance of our findings.

6 CONCLUSIONS

This chapter concludes the thesis. First, we present an overview of the findings, which were the result of the empirical evaluation of our theoretical model. Second, we present the potential theoretical and practical contributions resulting from our findings. Third, we recognize the limitations of our study. Fourth, based on our limitations, we proposed new research opportunities that can be undertaken in the future. Finally, we present the overarching conclusions.

6.1 OVERVIEW OF FINDINGS

As evidenced in the literature review, many domain ontologies have been developed to facilitate the development of *shared understanding* (SU), however there is a lack of empirical evidence to support such claims. Accordingly, to address this gap in the body of knowledge in IS, the aim of this research was to theoretically and empirically investigate the effect of a domain ontology on *shared understanding* (SU) and *cross-understanding* (CU) in groups. Specifically, our main research question in this study was "*what is the effect of using a domain ontology on shared understanding and cross-understanding*?" In this study, we developed a theoretical model theorizing the effect of domain ontology usage on SU and CU. We then tested the model through experimental research. The findings from our empirical evaluation are described in Table 6-1.

Table 6-1.	Summary	of Results	of Thesis
------------	---------	------------	-----------

	Results
1.	Group members who use a domain ontology develop significantly higher levels of shared understanding than
	those who do not use a domain ontology.
2.	a) Group members who use a domain ontology have significantly better communication quality in terms of
	discussion efficiency and communication clarity than those who do not use a domain ontology.
	b) There is no significant difference in communication openness and task discussion effectiveness among group
	members who use a domain ontology and those who do not use a domain ontology.
3.	There is no significant improvement in shared understanding between group members who use a domain
	ontology and those who do not use a domain ontology.
4.	a) Discussion efficiency significantly mediates the effect of domain ontology usage on the improvement of
	shared understanding.
	b) Communication openness, communication clarity and task discussion effectiveness do not significantly
	mediate the effect of domain ontology usage on the improvement of shared understanding.
5.	Group members who use a domain ontology develop a significantly greater cross-understanding than those who
	do not use a domain ontology.
6.	a) Communication quality in terms of discussion efficiency significantly mediates the effect of domain ontology
	usage on the development of cross-understanding.
	b) Communication openness, communication clarity and task discussion effectiveness do not significantly
	mediate the effect of domain ontology usage on the development of cross-understanding.

As per Table 6-1, the group members who used a domain ontology developed significantly higher levels of *shared understanding* than those who did not use a domain ontology. This implies that if group members are provided a (good quality) domain ontology, it is possible to achieve a significantly

higher level of *shared understanding* development between group members before they interact with each other as compared to group members who do not have access to a domain ontology.

Second, group members who use a domain ontology have significantly better *Communication Quality* in terms of *Discussion Efficiency* and *Communication Clarity* than those who do not use a domain ontology. In other words, our results show that through the use of a good quality ontology it is possible to increase the efficiency and clarity of communication between group members. However, communication openness and task discussion effectiveness are not increased significantly through the use of an ontology. Hence, we consider that these factors are more affected by both group and task characteristics, and therefore, more research is necessary to clarify this issue.

Third, there is no significant improvement in shared understanding between team members who use a domain ontology and those who do not use a domain ontology. According to this result, the use of the ontology in a group interaction task is not enough to produce a significant improvement in shared understanding. However, the results are important in practice when shared understanding needs to be improved because of the medium-size effect (Cohen, 1992). Indeed, considering that the interaction group task was limited in time in our experimental setting, a longer activity or more interactions may be required to get a significant effect – e.g. see (He et al., 2007).

Fourth, *Discussion Efficiency* significantly mediates the effect of *Domain Ontology Usage* on the improvement of *shared understanding*. That is, although the use of a domain ontology improves *shared understanding*, this improvement also depends on how efficient the discussion between group members is. Contrarily to our hypotheses, other communication quality factors such as *Communication Openness, Communication Clarity and Task Discussion Effectiveness* did not mediate the effect of Domain Ontology on shared understanding. In other words, some communication quality factors are less relevant during shared understanding improvement. Thus, we consider that further research is required to understand what other quality factors mediate the effect of *Domain Ontology Usage* on the improvement of *shared understanding*.

Fifth, group members who used a domain ontology developed a significantly greater *cross-understanding* than those who did not use a domain ontology. This is an important finding because it indicates that the effect of *Domain Ontology Use* extends beyond just improving *shared understanding*. Although we recognize that further research is required to better understand how *cross-understanding* is developed, our results are significant in terms of practical applications. That is, the use of a domain ontology during an interaction task will lead to better development of *cross-*

understanding of the said domain among group members as compared to a group not using the domain ontology.

Finally, communication quality in terms of *Discussion Efficiency* significantly mediates the effect of domain ontology usage on the development of *cross-understanding*. That is, the development of *cross-understanding* is not only affected by the use of the domain ontology, but is also mediated by how efficient the communication is. Contrary to our hypotheses, *Communication Openness*, *Communication Clarity* and *Task Discussion Effectiveness* did not significantly mediate the effect of *Domain Ontology Usage* on the development of *cross-understanding*. Further research is required to understand what other communication quality factors may mediate the effect of *Domain Ontology Usage* on improvement of *cross-understanding*.

In summary, our results show that a domain ontology can be beneficial for *shared understanding* and *cross-understanding* in groups. First, domain ontology usage has an important effect on the *Development of Shared Understanding* and *Development of Cross-Understanding* but not *Shared Understanding Improvement*. Also, our results show that *communication quality* factors, specifically *Discussion Efficiency* and *Communication Clarity*, are improved through the use of a domain ontology. Finally, *Discussion Efficiency* is also important in mediating the effect of a domain ontology in *Shared Understanding Improvement* and *Cross-Understanding Development*.

6.2 CONTRIBUTIONS AND IMPLICATIONS

We argue that the results of this study contribute to both theory and practice. In the following subsections, we present the theoretical and practical contributions based on the findings of this study.

6.2.1 Theoretical Contributions

6.2.1.1 Theory of Domain Ontology Effect on Shared Understanding and Cross-Understanding

In IS and related fields, one of the often-cited benefits of conceptual models, e.g. domain ontologies, is that they facilitate *shared understanding* (SU) (Al-Debei & Fitzgerald, 2010; Innab et al., 2012; Lin et al., 2004; Uschold et al., 1998). However, such claims have lacked theoretical and empirical validation.

Through this study, we contribute to the theory of conceptual modelling by presenting a model of the effect of domain ontology usage on SU and *cross-understanding* (CU) in groups (Figure 6-1). To the best of our knowledge, this is the first theory of domain ontology usage in the body of knowledge of IS that considers the effect of a domain ontology on the group level. According to the results of our

model, domain ontology usage affects the development of SU. The model also shows that domain ontology usage affects communication quality and CU development. Furthermore, it shows that SU improvement and CU development are mediated by the communication quality factor *discussion efficiency*.



Figure 6-1. Theory of Domain Ontology Usage on Shared Understanding and Cross-Understanding

Another important contribution to theory is that at the group level, *communication quality* is a key construct on enabling SU and CU. Further, this construct can also be improved through the use of a domain ontology. As we mentioned before, communication quality is an important enabler of performance in groups in organizations (Bittner & Leimeister, 2014; Briggs, 2014; Field, 2013; Huber & Lewis, 2010; Salas et al., 2008). Thus, domain ontologies can also contribute to group performance by supporting SU and CU.

In this study, we also contribute to the theory by providing the first empirical evaluation of our theoretical model. As we mentioned in the previous chapter, our findings support fully H1 and H5, and partially H2, H4, and H6, which open a new spectrum of research opportunities that will be exposed later in this chapter in Section 6.4.

6.2.1.2 Operationalization and Empirical Testing of Cross-Understanding

Cross-understanding is a recent construct introduced by Huber and Lewis (2010) to explain contradictory results in empirical validations of team performance. Because of its recent introduction, to the best of our knowledge, this construct has not been operationalized and tested empirically. Although Huber and Lewis (2010) recommend two approaches, perceptual and behavioral, to operationalize CU, there are no specific or detailed instructions of how to perform such operationalization. In this study, we adopted the perceptual recommendation and through a coding process that considers the answers of two participants in a group, we estimate CU (See section 4.8.1).

The coding process shows that CU in each team member can be determined through comparing team members' answers on comprehension questions.

6.2.2 Practical Contributions

We consider that the most important contribution to practice is the enhanced understanding of the role that domain ontologies can play as information tools in novice groups. Previous studies show that visual ontologies can be useful for knowledge identification (a particular result of domain understanding at the individual level), e.g. Bera et al. (2011). However, as we mentioned in our literature review, there is a lack of evidence to support the benefits that domain ontologies provide in enabling SU, even less for CU. Thus, we consider that our study is the first to provide evidence of how domain ontologies can be used to support domain understanding at the group level, specifically on SU and CU. Indeed, our study shows that the use of domain ontologies as information tools during group tasks can enable SU and CU.

Our empirical validation shows how, through the use of the compliance management domain ontology CoMOn, team members can achieve a good understanding of the compliance management domain and also enhanced their SU and CU of the domain. The compliance management domain is a very complex domain that suffers from conflicting understanding of compliance management in organizations (Australasian Compliance Institute, 2013a; Open Compliance & Ethics Group, 2012) and hence it is difficult to achieve a shared understanding of it in group members. Moreover, although many organizations are using different tools that help control compliance, there is still a lack of understanding of how compliance management is related to, or impacts the organization's operations, in addition to a lack of common understanding of compliance in general within different functions of the same organization.

Thus, the introduction of CoMOn as an information tool within an organization can potentially assist in overcoming the differences and establish a common understanding. Domain ontologies such as CoMOn can be introduced in a variety of formats that engage employees in a group task, including but not limited to use in employee induction and training workshops and as a reference source for policy and process documentations.

6.3 LIMITATIONS

Although we believe our study makes a significant contribution, it also has limitations that are necessary to explain. The limitations of our study can be understood with reference to the traditional criteria for validity (Bera et al., 2011). Below we describe the main limitations of our study.

In terms of *internal validity*, a challenge could be raised in terms of the treatments that we used in our experiments. Specifically, we use a graphical domain ontology in our treatment group and a text based description of the domain in the control group. As these are two different presentation types of information, this could have created a difference in the results across groups that may not have been due to our treatment. While this might have been possible, we do not consider it to be a major concern for the following reason: we ensured that all the comprehension questions could be answered irrespective of the type of the treatment used and the information provided to each group was equivalent. Thus, we believe that general conclusions can be drawn safely from the experiments.

In terms of *construct validity*, we operationalized each construct in our study in a limited way. It would be ideal to manipulate each treatment in multiple ways (e.g. different domains, different representations of the ontology) to measure variables in various ways, and to study the effects of our treatments on our dependent variables in different tasks (Burton-Jones & Meso, 2008). However, we could not proceed in that way due to limitations of time and resources. Thus, our results are limited to the characteristics of our experimental design, and readers should be cautious in over-generalizing from our results (Bera et al., 2011; Burton-Jones & Meso, 2008).

In terms of *external validity*, our study could be challenged because we used laboratory experiments with a sample of university students rather than a sample of real practitioners or users. However, this practice is typical in the early stages of research (Trochim & Donnelly, 2008), and we maintained some realism by using a domain and an ontology that complied with our study necessities, while satisfying the need to have novice participants. That is, we believe that the students who participated in our experiments were very similar to the target population, in that they are individuals of a diverse background who are not very familiar with a domain and were learning from conceptual models.

Finally, in terms of *statistical conclusion validity*, the limited statistical power because of the modest sample size in the present study (N = 30) may have played a role in constraining the significance of the statistical comparisons conducted. Although a post hoc power analysis for t-tests shows the used sample size is sufficient, this not the case for the mediation analysis. First, the post hoc power analysis for t-tests revealed that on the basis of the mean, between-groups comparison effect size observed in the present study (d = 2.02), approximately 12 samples would have been sufficient to obtain statistical power at the recommended 0.80 level (Cohen, 1992). To the contrary, a study of the required sample size to detect a mediation effect to achieve statistical power of 0.80 level shows that at least 385 samples are needed when we consider our observed effects of the dependent variable on the mediators (a < 0.10) and the effect of the mediator on the independent variable (b < 0.59) (Fritz & MacKinnon, 2007).

6.4 FUTURE RESEARCH

In this study, we have taken the first step to study domain understanding of conceptual models at the group level. However, we consider that there are still aspects of the study that can be extended. First, our theoretical model considered only 4 factors of communication quality. We know that the communication process between human beings can be very complex (Dennis, Fuller, & Valacich, 2008), Thus it is important to know what other factors may be affecting communication processes and how they affect *shared understanding* (SU) and *cross-understanding* (CU).

Second, our results show that SU improvement was not significantly different with the use of the domain ontology to without. It is important to understand if such improvement can be improved over the time. Our group task was designed for only one short meeting between participants. Despite this, participants achieved a representative improvement on SU with the use of the domain ontology. We consider that SU improvement could be more significant through more team interactions over a longer period of time.

Third, we used an experiment in a lab setting to test our hypothesis. Although we can obtain high internal validity, external validity still remains low (Trochim & Donnelly, 2008). Through other types of studies, such as case studies, our findings can be contrasted and further developed. Thus, we encourage researchers to consider further research on this topic.

Finally, the theory of ontology effects on SU and CU is a foundation on which to understand domain understanding at the group level. Indeed, there are plenty of opportunities that need to be explored. For example, it would be important to know how SU and CU might be affected by both external and internal factors to the ontology. As we explained in Section 2.3, there are many factors that affect the individual understanding of group members, and hence SU and CU. However, we do not yet know how drastic the impact of those factors is on these shared cognition constructs. By determining the impact of these factors, it will be possible make recommendations of ontology design and development to get better results on SU and CU.

6.5 CONCLUDING REMARKS

In this study, we considered the importance that *shared understanding* (SU) and *cross-understanding* (CU) have on task performance in teams. Our literature review shows that there is a lack of mechanisms to improve such constructs in groups. Indeed, our literature review shows that domain ontologies might facilitate SU, but there is a lack of theoretical and empirical evidence to support such claims. Thus, we considered as an artifact of study domain ontologies as enabler of SU and CU.

Indeed, our study examined the effect of domain ontology usage on shared understanding and crossunderstanding in novice teams.

To achieve this aim, first, we developed a theoretical model that describes such effect. According to our theoretical model, domain ontology usage affects the development of *shared understanding*; as well as the development of *cross-understanding*. However, the effect of domain ontology usage on improvement of *shared understanding* and development of *cross-understanding* is mediated by *communication quality* (CQ) factors.

To test our theoretical model, we considered a specific domain and a particular kind of representation of a domain ontology. That is, we selected the compliance management domain and a graphical ontology of this domain (CoMOn). On the one hand, compliance management has been an important topic discussed within academia and industry for three main reasons: high-compliance management cost (Protiviti, 2012), drastic consequences of failing to comply (Australasian Compliance Institute, 2013b), and a weak and conflicting understanding of compliance management concepts in organizations (Australasian Compliance Institute, 2013a; Open Compliance & Ethics Group, 2012). Thus, we conclude that compliance management is a suitable setting in which to study the effectiveness of domain ontology in facilitating SU and CU. On the other hand, the compliance management ontology (CoMOn) is an informal ontology developed with the specific purpose to facilitate the development of a SU of compliance management (Syed Abdullah et al., 2013), and thus presents an ideal basis on which to conduct our test.

We tested our hypotheses through an experimental research design. We considered a two-group, posttest only, randomized experimental design (Shanks, 2008; Trochim & Donnelly, 2008). Accordingly, our treatment group used CoMOn and our control group used a portion of the standard ISO 19600:2014 (British Standards Institution, 2014), which provides guidelines for compliance management systems. Also, we considered as participants for our experiment students from a course of database fundamental offered in the School of Information Technology and Electrical Engineering at the University of Queensland. These students have zero or low level of conceptual modeling knowledge and compliance management, which helped us to recreate similar users to those we targeted – i.e., novice users.

Our results show that, first, group members who use a domain ontology develop significant higher levels of *shared understanding* than those who do not use a domain ontology. Second, group members who use a domain ontology have significantly better *communication quality* in terms of *Discussion Efficiency* and *Communication Clarity* than those who do not use a domain ontology. Third,

communication quality in terms of *Discussion Efficiency* significantly mediates the effect of *Domain Ontology Usage* on the improvement of *shared understanding*. Fourth, group members who use a domain ontology develop a significantly greater *cross-understanding* than those who do not use a domain ontology. Finally, *communication quality* in terms of discussion efficiency significantly mediates the effect of domain ontology usage on the development of *cross-understanding*.

Our theoretical model and the findings of its evaluation contribute to both theory and the practice of IS. Firstly, by expanding the body of knowledge of conceptual modelling at the group level by introducing the study of *shared understanding* and *cross-understanding*. Secondly, by presenting how domain ontologies can be used in practice to enhance *shared understanding* and *cross-understanding*, and consequently improve performance in groups. As we mentioned before, improving performance in groups is one of the most challenging tasks in organizations (Bittner & Leimeister, 2014).

REFERENCES

- Aguirre-Urreta, M. I. M., George M. (2008). Comparing conceptual modeling techniques: a critical review of the EER vs. OO empirical literature. *SIGMIS Database*, *39*(2), 9-32. doi:10.1145/1364636.1364640
- Al-Debei, M. M., & Fitzgerald, G. (2010). The design and engineering of mobile data services: developing an ontology based on business model thinking. *IFIP Advances in Information and Communication Technology*, 318, 28-51.
- Allen, G. N., & March, S. T. (2006). The Effects of State-Based and Event-Based Data Representation on User Performance in Query Formulation Tasks. *MIS Quarterly*, 30(2), 269-290. doi:10.2307/25148731
- Almeida, M. B. (2013). Revisiting ontologies: A necessary clarification. *Journal of the American* Society for Information Science and Technology, 64(8), 1682-1693. doi:10.1002/asi.22861
- Australasian Compliance Institute. (2013a). 2012 Compliance and Risk Management Benchmarking
StudyResults.Retrievedfromhttp://www.acigrc.com/www_aci/default.asp?menuid=24&artid=1538fromfromfrom
- Australasian Compliance Institute. (2013b). Legacy of non-compliance catches up with Cotton on. *GRC Professional, Summer 2013*.
- Bera, P., Burton-Jones, A., & Wand, Y. (2011). Guidelines For Designing Visual Ontologies to Support Knowledge Identification. *MIS Quarterly*, 35(4), 883-908.
- Bera, P., Burton-Jones, A., & Wand, Y. (2014). Research Note—How Semantics and Pragmatics Interact in Understanding Conceptual Models. *Information Systems Research*, 0(0), null. doi:doi:10.1287/isre.2014.0515
- Berners-Lee, T., Hendler, J., & Lassila, O. (2001). The Semantic Web. Scientific American, 35-43.
- Bierhals, R., Schuster, I., Kohler, P., & Badke-Schaub, P. (2007). Shared mental models—linking team cognition and performance. *CoDesign*, *3*(1), 75-94. doi:10.1080/15710880601170891
- Bittner, E. A. C., & Leimeister, J. M. (2014). Creating Shared Understanding in Heterogeneous Work Groups: Why It Matters and How to Achieve It. *Journal of Management Information Systems*, *31*(1), 111-144. doi:10.2753/MIS0742-1222310106
- Bland, J. M., & Altman, D. G. (1997). Statistics notes: Cronbach's alpha. *BMJ*, 314(7080), 572. doi:10.1136/bmj.314.7080.572
- Boella, G., Humphreys, L., Martin, M., Rossi, P., & Torre, L. (2012). Eunomos, a Legal Document and Knowledge Management System to Build Legal Services. In M. P. Palmirani, Ugo; Casanovas, Pompeu; Sartor, Giovanni (Ed.), AI Approaches to the Complexity of Legal Systems. Models and Ethical Challenges for Legal Systems, Legal Language and Legal Ontologies, Argumentation and Software Agents (Vol. 7639, pp. 131-146): Springer Berlin Heidelberg.

- Briggs, R. O. (2014). Special Section: Information Systems Support for Shared Understanding. Journal of Management Information Systems, 31(1), 107-110. doi:10.2753/MIS0742-1222310105
- British Standards Institution. (2014). BS ISO 19600:2014 Compliance Management Systems Guidelines (pp. 28). Switzerland: BSI Standards.
- Burton-Jones, A., Clarke, R., Lazarenko, K., & Weber, R. (2012). Is Use of Optional Attributes and Associations in Conceptual Modeling Always Problematic? Theory and Empirical Tests.
- Burton-Jones, A., & Meso, P. N. (2006). Conceptualizing Systems for Understanding: An Empirical Test of Decomposition Principles in Object-Oriented Analysis. *Information Systems Research*, 17(1), 38-60. doi:doi:10.1287/isre.1050.0079
- Burton-Jones, A., & Meso, P. N. (2008). The effects of decomposition quality and multiple forms of information on novices' understanding of a domain from a conceptual model. *Journal of the Association for Information Systems*, 9(12), 748-802.
- Burton-Jones, A., Wand, Y., & Weber, R. (2009). Guidelines for Empirical Evaluations of Conceptual Modeling Grammars. *Journal of the Association for Information Systems*, 10(6), 495-532. doi:<u>http://aisel.aisnet.org/jais/vol10/iss6/1</u>
- Burton-Jones, A., & Weber, R. (2014). Building Conceptual Modeling on the Foundation of Ontology. In H. Topi, T. Gonzalez, A. Tucker, & J. Diaz-Herrera (Eds.), *Computing Handbook: Information Systems and Information Technology* (Third ed., Vol. 2, pp. 15.11-15.24). Boca Raton, FL, U.S.A.: CRC Press.
- Cannon-Bowers, J. A., & Salas, E. (2001). Reflections on Shared Cognition. *Journal of Organizational Behavior*, 22(2), 195-202. doi:10.2307/3649591
- Cerans, K., Liepins, R., Sprogis, A., Ovcinnikova, J., & Barzdins, G. (2015). Domain-Specific OWL Ontology Visualization with OWLGrEd. In E. Simperl, B. Norton, D. Mladenic, E. Della Valle, I. Fundulaki, A. Passant, & R. Troncy (Eds.), *The Semantic Web: ESWC 2012 Satellite Events: ESWC 2012 Satellite Events, Heraklion, Crete, Greece, May 27-31, 2012. Revised Selected Papers* (pp. 419-424). Berlin, Heidelberg: Springer Berlin Heidelberg.
- Clarke, R., Burton-Jones, A., & Weber, R. (2013). *Improving the Semantics of Conceptual-Modeling Grammars: A New Perspective on an Old Problem.* Paper presented at the Thirty Fourth International Conference on Information Systems, Milan.
- Cohen, J. (1992). Statistical Power Analysis. *Current Directions in Psychological Science*, 1(3), 98-101.
- COMPAS. (2011). COMPAS: Compliance-driven Models, Languages, and Architectures for Services. Retrieved from <u>http://www.compas-ict.eu/</u>
- Compeau, D., Marcolin, B., Kelley, H., & Higgin, C. (2012). Research Commentary— Generalizability of Information Systems Research Using Student Subjects—A Reflection on Our Practices and Recommendations for Future Research. *Information Systems Research*, 23(4), 1093-1109. doi:doi:10.1287/isre.1120.0423
- Cook, T. D., & Campbell, D. T. (1979). *Quasi-Experimentation: Design & Analysis Issues for Field Settings*: Rand McNally & Company.
- Cooke, N. J., Gorman, J. C., Myers, C. W., & Duran, J. L. (2013). Interactive Team Cognition. *Cognitive Science*, 37(2), 255-285. doi:10.1111/cogs.12009
- Cooke, N. J., Salas, E., Cannon-Bowers, J. A., & Stout, R. J. (2000). Measuring Team Knowledge. *Human Factors: The Journal of the Human Factors and Ergonomics Society*, 42(1), 151-173. doi:10.1518/001872000779656561
- Corvera Charaf, M., Rosenkranz, C., & Holten, R. (2013). The emergence of shared understanding: applying functional pragmatics to study the requirements development process. *Information Systems Journal*, 23(2), 115-135. doi:10.1111/j.1365-2575.2012.00408.x
- Davison, R. (1997). An instrument for measuring meeting success. *Information & Management*, 32(4), 163-176. doi:http://dx.doi.org/10.1016/S0378-7206(97)00020-7
- De Nicola, A., Missikoff, M., & Navigli, R. (2009). A software engineering approach to ontology building. *Information Systems*, 34(2), 258-275. doi:http://dx.doi.org/10.1016/j.is.2008.07.002
- DeChurch, L. A., & Mesmer-Magnus, J. R. (2010a). The cognitive underpinnings of effective teamwork: A meta-analysis. *Journal of Applied Psychology*, 95(1), 32-53. doi:10.1037/a0017328
- DeChurch, L. A., & Mesmer-Magnus, J. R. (2010b). Measuring shared team mental models: A metaanalysis. *Group Dynamics: Theory, Research, and Practice, 14*(1), 1-14. doi:10.1037/a0017455
- Delir Haghighi, P., Burstein, F., Zaslavsky, A., & Arbon, P. (2013). Development and evaluation of ontology for intelligent decision support in medical emergency management for mass gatherings. *Decision Support Systems*, 54(2), 1192-1204. doi:http://dx.doi.org/10.1016/j.dss.2012.11.013
- Dennis, A. R., Fuller, R. M., & Valacich, J. S. (2008). Media, Tasks, and Communication Processes: A Theory of Media Synchronicity. *MIS Quarterly*, *32*(3), 575-600.
- El Kharbili, M. (2012). Business Process Regulatory Compliance Management Solution Frameworks: A Comparative Evaluation. Paper presented at the Asia-Pacific Conference on Conceptual Modelling (APCCM 2012), Melbourne, Australia.
- Elgammal, A., Turetken, O., Heuvel, W.-J., & Papazoglou, M. (2011). On the Formal Specification of Regulatory Compliance: A Comparative Analysis. In E. M. R. Maximilien, Gustavo; Yuan, Soe-Tsyr; Ludwig, Heiko; Fantinato, Marcelo (Ed.), *Service-Oriented Computing* (Vol. 6568, pp. 27-38): Springer Berlin Heidelberg.
- Fernández-López, M., Gómez-Pérez, A., & Juristo, N. (1997). *METHONTOLOGY: From Ontological Art Towards Ontological Engineering*. Paper presented at the AAAI-97 Spring Symposium Series, Stanford University, EEUU.
- Field, A. (2013). Discovering statistics using IBM SPSS statistics: Sage.
- Fiore, S. M., & Schooler, J. W. (2004). Process mapping and shared cognition: Teamwork and the development of shared problem models. In E. S. S. M. Fiore (Ed.), *Team cognition: Understanding the factors that drive process and performance* (pp. 133-152). Washington, DC, US: American Psychological Association.

- Fonseca, F. (2007). The double role of ontologies in information science research. *Journal of the American Society for Information Science and Technology*, 58(6), 786-793. doi:10.1002/asi.20565
- Foorthuis, R., & Bos, R. (2011). A Framework for Organizational Compliance Management Tactics. In C. P. Salinesi, Oscar (Ed.), Advanced Information Systems Engineering Workshops (Vol. 83, pp. 259-268): Springer Berlin Heidelberg.
- Fritz, M. S., & MacKinnon, D. P. (2007). Required Sample Size to Detect the Mediated Effect. *Psychological Science*, *18*(3), 233-239. doi:10.1111/j.1467-9280.2007.01882.x
- Gartner. (2012). Magic Quadrant for Enterprise Governance, Risk and Compliance Platforms2013(06/02/2013). Retrieved from http://www.gartner.com/technology/reprints.do?id=1-1CCZVJP&ct=121004&st=sg
- Gašević, D., Kaviani, N., & Milanović, M. (2009). Ontologies and Software Engineering. In S. S. Staab, Rudi (Ed.), *Handbook on Ontologies* (pp. 593-615): Springer Berlin Heidelberg.
- Gemino, A. W., Yair. (2004). A framework for empirical evaluation of conceptual modeling techniques. *Requirements Engineering*, 9(4), 248-260. doi:10.1007/s00766-004-0204-6
- Giaretta, P., & Guarino, N. (1995). Ontologies and knowledge bases towards a terminological clarification *Towards very large knowledge bases: knowledge building & knowledge sharing 1995* (pp. 25-32). Amsterdan: IOS Press.
- Glinz, M., & Fricker, S. A. (2014). On shared understanding in software engineering: an essay. *Computer Science - Research and Development, 30*(3), 363-376. doi:10.1007/s00450-014-0256-x
- Gómez-Pérez, A., Fernandez-Lopez, M., & Corcho, O. (2004). *Ontological Engineering* (First ed.). USA: Springer London.
- Grenon, P., & Smith, B. (2004). SNAP and SPAN: Towards Dynamic Spatial Ontology. *Spatial Cognition & Computation*, 4(1), 69-104. doi:10.1207/s15427633scc0401_5
- Grimm, S., Abecker, A., Völker, J., & Studer, R. (2011). Ontologies and the Semantic Web. In J. Domingue, D. Fensel, & J. A. Hendler (Eds.), *Handbook of Semantic Web Technologies* (pp. 507-579): Springer Berlin Heidelberg.
- Gruber, T. R. (1993). A translation approach to portable ontology specifications. *Knowledge* Acquisition, 5(2), 199-220. doi:<u>http://dx.doi.org/10.1006/knac.1993.1008</u>
- Gruber, T. R. (1995). Toward principles for the design of ontologies used for knowledge sharing. *International Journal of Human-Computer Studies*, 43(5–6), 907-928. doi:http://dx.doi.org/10.1006/ijhc.1995.1081
- Grüninger, M., & Fox, M. S. (1995). Methodology for the Design and Evaluation of Ontologies.
- Guarino, N. (1998). Formal Ontology and Information Systems Formal Ontology in Information Systems: Frontiers in Artificial Intelligence and Applications (pp. 3-15). Amsterdan, Netherlands: IOS Press.
- Guarino, N., Oberle, D., & Staab, S. (2009). What Is an Ontology? In S. S. Staab, Rudi (Ed.), *Handbook on Ontologies* (pp. 1-17): Springer Berlin Heidelberg.

- Guarino, N., & Welty, C. A. (2009). An Overview of OntoClean. In S. Staab & R. Studer (Eds.), *Handbook on Ontologies* (pp. 201-220): Springer Berlin Heidelberg.
- Guha, R. V., & Lenat, D. B. (1990). CYC A Midterm Report. AI magazine, 11(3), 32-59.
- Hayes, A. F. (2013). Introduction to mediation, moderation, and conditional process analysis: A regression-based approach: Guilford Press.
- He, J., Butler, B. S., & King, W. R. (2007). Team Cognition: Development and Evolution in Software Project Teams. *Journal of Management Information Systems*, 24(2), 261-292. doi:10.2753/MIS0742-1222240210
- Hepp, M. (2008). Ontologies: State of the Art, Business Potential, and Grand Challenges. In M. Hepp,P. De Leenheer, A. De Moor, & Y. Sure (Eds.), *Ontology Management* (Vol. 7, pp. 3-22): Springer US.
- Horrocks, I., & Patel-Schneider, P. F. (2011). KR and Reasoning on the Semantic Web: OWL. In J. Domingue, D. Fensel, & J. Hendler (Eds.), *Handbook of Semantic Web Technologies* (pp. 365-398): Springer Berlin Heidelberg.
- Huber, G. P., & Lewis, K. (2010). Cross-Understanding: Implications for Group Cognition and Performance. *Academy of Management Review*, 35(1), 6-26.
- Huber, G. P., & Lewis, K. (2011). Cross-Understanding and Shared Social Theories. Academy of Management Review, 36(2), 422-424.
- Innab, N., Kayed, A., & Sajeev, A. S. M. (2012). An ontology for software requirements modelling. Paper presented at the International Conference on Information Science and Technology (ICIST), Hubei, Republic of China.
- Jarrar, M., & Meersman, R. (2009). Ontology Engineering The DOGMA Approach. In T. C. Dillon, Elizabeth; Meersman, Robert; Sycara, Katia (Ed.), Advances in Web Semantics I (Vol. 4891, pp. 7-34): Springer Berlin Heidelberg.
- Jentsch, C., Beimborn, D., Jungnickl, C. P., & Renner, G. S. (2014). How to Measure Shared Understanding among Business and IT. Academy of Management Proceedings, 2014(1). doi:10.5465/ambpp.2014.138
- Johnson, T. E., Lee, Y., Lee, M., O'Connor, D. L., Khalil, M. K., & Huang, X. (2007). Measuring Sharedness of Team-Related Knowledge: Design and Validation of a Shared Mental Model Instrument. *Human Resource Development International*, 10(4), 437-454. doi:10.1080/13678860701723802
- Johnson, T. E., & O'Connor, D. L. (2008). Measuring team shared understanding using the analysisconstructed shared mental model methodology. *Performance Improvement Quarterly*, 21(3), 113-134. doi:10.1002/piq.20034
- Kacmar, K. M. (2003). The interactive effect of leader-member exchange and communication frequency on performance ratings. *Journal of Applied Psychology*, 88(4), 764-772. doi:10.1037/0021-9010.88.4.764
- Kahai, S. S., & Cooper, R. B. (2003). Exploring the Core Concepts of Media Richness Theory: The Impact of Cue Multiplicity and Feedback Immediacy on Decision Quality. *Journal of Management Information Systems*, 20(1), 263-299. doi:10.1080/07421222.2003.11045754

- Kayworth, T. R., & Leidner, D. E. (2002). Leadership Effectiveness in Global Virtual Teams. *Journal* of Management Information Systems, 18(3), 7-40. doi:10.1080/07421222.2002.11045697
- Khatri, V., Vessey, I., Ram, S., & Ramesh, V. (2006). Cognitive fit between conceptual schemas and internal problem representations: the case of geospatio-temporal conceptual schema comprehension. *Professional Communication, IEEE Transactions on, 49*(2), 109-127. doi:10.1109/TPC.2006.875091
- Khatri, V., Vessey, I., Ramesh, V., Clay, P., & Park, S.-J. (2006). Understanding Conceptual Schemas: Exploring the Role of Application and IS Domain Knowledge. *Information Systems Research*, *17*(1), 81-99. doi:10.1287/isre.1060.0081
- Kim, J. A. (2012). Understanding Knowledge Representation in the Knowledge Management Environment: Evaluation of Ontology Visualization Methods. *Knowledge Organization*, 39(3), 193-203.
- Kishore, R., & Sharman, R. (2004). Computational Ontologies and Information Systems I: Foundations. *Communications of the Association for Information Systems*, 14(8), 158-183.
- Ko, D.-G., Kirsch, L. J., & King, W. R. (2005). Antecedents of knowledge transfer from consultants to clients in enterprise system implementations. *MIS Quarterly*, 29(1), 59-85.
- Kotis, K., & Vouros, G. (2006). Human-centered ontology engineering: The HCOME methodology. *Knowledge and Information Systems, 10*(1), 109-131. doi:10.1007/s10115-005-0227-4
- Langan-Fox, J., Anglim, J., & Wilson, J. R. (2004). Mental models, team mental models, and performance: Process, development, and future directions. *Human Factors and Ergonomics in Manufacturing & Service Industries*, 14(4), 331-352. doi:10.1002/hfm.20004
- Langan-Fox, J., Wirth, A., Code, S., Langfield-Smith, K., & Wirth, A. (2001). Analyzing shared and team mental models. *International Journal of Industrial Ergonomics*, 28(2), 99-112. doi:http://dx.doi.org/10.1016/S0169-8141(01)00016-6
- Lenat, D. B., & Guha, R. V. (1989). Building large knowledge-based systems; representation and inference in the Cyc project: Addison-Wesley Longman Publishing Co., Inc.
- Levesque, L. L., Wilson, J. M., & Wholey, D. R. (2001). Cognitive divergence and shared mental models in software development project teams. *Journal of Organizational Behavior*, 22(2), 135-144. doi:10.1002/job.87
- Lim, B.-C., & Klein, K. J. (2006). Team mental models and team performance: a field study of the effects of team mental model similarity and accuracy. *Journal of Organizational Behavior*, 27(4), 403-418. doi:10.1002/job.387
- Lin, H. K., Harding, J. A., & Shahbaz, M. (2004). Manufacturing system engineering ontology for semantic interoperability across extended project teams. *International Journal of Production Research*, 42(24), 5099-5118. doi:10.1080/00207540412331281999
- Lowry, P. B., Romano, N. C., Jenkins, J. L., & Guthrie, R. W. (2009). The CMC Interactivity Model: How Interactivity Enhances Communication Quality and Process Satisfaction in Lean-Media Groups. *Journal of Management Information Systems*, 26(1), 155-196. doi:10.2753/MIS0742-1222260107

- Lutz, M., & Klien, E. (2006). Ontology-based retrieval of geographic information. *International Journal of Geographical Information Science*, 20(3), 233-260. doi:10.1080/13658810500287107
- Mathieu, J. E., Heffner, T. S., Goodwin, G. F., Salas, E., & Cannon-Bowers, J. A. (2000). The influence of shared mental models on team process and performance. *Journal of Applied Psychology*, 85(2), 273-283. doi:10.1037/0021-9010.85.2.273
- Mayer, R. E. (2005). Cognitive theory of multimedia learning. *The Cambridge handbook of multimedia learning*, 31-48.
- Mayer, R. E. (2009). *Multimedia Learning* (Second ed.). New York, USA: Cambridge University Press.
- Milton, S. K., & Smith, B. (2004). *Top-level ontology: The problem with naturalism*. Paper presented at the Formal ontology in information systems.
- Mohammed, S., & Dumville, B. C. (2001). Team mental models in a team knowledge framework: expanding theory and measurement across disciplinary boundaries. *Journal of Organizational Behavior*, 22(2), 89-106. doi:10.1002/job.86
- Mohammed, S., Ferzandi, L., & Hamilton, K. (2010). Metaphor No More: A 15-Year Review of the Team Mental Model Construct. *Journal of Management*. doi:10.1177/0149206309356804
- Mohammed, S., Klimoski, R., & Rentsch, J. R. (2000). The Measurement of Team Mental Models: We Have No Shared Schema. Organizational Research Methods, 3(2), 123-165. doi:10.1177/109442810032001
- Moody, D. (2007). What Makes a Good Diagram? Improving the Cognitive Effectiveness of Diagrams in IS Development. In W. Wojtkowski, W. G. Wojtkowski, J. Zupancic, G. Magyar, & G. Knapp (Eds.), Advances in Information Systems Development (pp. 481-492): Springer US.
- Moody, D. L. (2004). Cognitive Load Effects on End User Understanding of Conceptual Models: An Experimental Analysis. In A. D. Benczúr, János; Gottlob, Georg (Ed.), *Advances in Databases and Information Systems* (Vol. 3255, pp. 129-143): Springer Berlin Heidelberg.
- Mueller, J. S. (2012). Why individuals in larger teams perform worse. Organizational Behavior and
Human Decision Processes, 117(1), 111-124.
doi:http://dx.doi.org/10.1016/j.obhdp.2011.08.004
- Mulder, I., Swaak, J., & Kessels, J. (2002). Assessing group learning and shared understanding in technology-mediated interaction. *Educational Technology and Society*, 5(1), 35-47.
- Neches, R., Fikes, R. E., Finin, T., Thomas, G., Patil, R., Senator, T., & Swartout, W. R. (1991). Enabling technology for knowledge sharing. *AI magazine*, *12*(3), 36.
- Nelson, K. M., & Cooprider, J. G. (1996). The Contribution of Shared Knowledge to IS Group Performance. *MIS Quartely*, 20(4), 409-432.
- Neuman, W. L. (2011). *Social Research Methods: Quantitative and Qualitative Approaches* (Seventh ed.). The United States of America: Pearson Education Inc.

- Niaraki, A. S., & Kim, K. (2009). Ontology based personalized route planning system using a multicriteria decision making approach. *Expert Systems with Applications*, 36(2), 2250-2259. doi:10.1016/j.eswa.2007.12.053
- Noy, N. F., & McGuinness, D. L. (2001). Ontology Development 101: A Guide to Creating Your First Ontology. Retrieved from
- O'Reilly, C. A., & Roberts, K. H. (1977). Task group structure, communication, and effectiveness in three organizations. *Journal of Applied Psychology*, 62(6), 674-681. doi:10.1037/0021-9010.62.6.674
- Open Compliance & Ethics Group. (2012). 2012 GRC Maturity Survey. Retrieved from http://www.oceg.org/resource/2012-grc-maturity-survey-all-respondents-report
- Oxford University Press. (Ed.) (2015). Oxford University Press.
- Parsons, J. (2011). An experimental study of the effects of representing property precedence on the comprehension of conceptual schemas. *Journal of the Association for Information Systems*, *12*(6), 1.
- Protiviti. (2012). 2012 Sarbanes-Oxley Compliance Survey. Retrieved from <u>http://www.protiviti.com/en-US/Documents/Surveys/2012-SOX-Compliance-Survey-</u> <u>Protiviti.pdf</u>
- Ramezani, E., Fahland, D., & Aalst, W. P. (2012). Where Did I Misbehave? Diagnostic Information in Compliance Checking. In A. G. Barros, Avigdor; Kindler, Ekkart (Ed.), *Business Process Management* (Vol. 7481, pp. 262-278): Springer Berlin Heidelberg.
- Randolph-Seng, B., & Norris, J. I. (2011). Cross-Understanding in Groups: How to "Cross Over" Without "Dying". Academy of Management Review, 36(2), 420-422. doi:10.5465/amr.2010.0177
- Recker, J., Rosemann, M., Green, P., & Indulska, M. (2011). Do Ontological Deficiencies in Modeling Grammars Matter. *MIS Quarterly*, 35(1), 57-79.
- Recker, J. C., Rosemann, M., Indulska, M., & Green, P. (2009). Business process modeling: a comparative analysis. *Journal of the Association for Information Systems*, 10(4), 333-363.
- Resnick, L. B. (1991). Shared cognition: Thinking as social practice. In L. B. Resnick, J. M. Levine, & S. D. Teasley (Eds.), *Perspectives on socially shared cognition* (pp. 1-20). Washington, DC, US: American Psychological Association.
- Sadiq, S., & Governatori, G. (2010). Managing Regulatory Compliance in Business Processes. In J. R. vom Brocke, Michael (Ed.), *Handbook on Business Process Management 2* (pp. 159-175): Springer Berlin Heidelberg.
- Sadiq, S., Governatori, G., & Namiri, K. (2007). Modeling Control Objectives for Business Process Compliance. In G. D. Alonso, Peter; Rosemann, Michael (Ed.), *Business Process Management* (Vol. 4714, pp. 149-164): Springer Berlin Heidelberg.
- Saghafi, A., & Wand, Y. (2014, 6-9 Jan. 2014). Do Ontological Guidelines Improve Understandability of Conceptual Models? A Meta-analysis of Empirical Work. Paper presented at the 2014 47th Hawaii International Conference on System Sciences (HICSS), Waikoloa, HI.

- Salas, E., Cooke, N. J., & Rosen, M. A. (2008). On Teams, Teamwork, and Team Performance: Discoveries and Developments. *Human Factors: The Journal of the Human Factors and Ergonomics Society*, 50(3), 540-547. doi:10.1518/001872008x288457
- Sapkota, K., Aldea, A., Younas, M., Duce, D. A., & Banares-Alcantara, R. (2012, 25-25 Sept. 2012). Extracting meaningful entities from regulatory text: Towards automating regulatory compliance. Paper presented at the Requirements Engineering and Law (RELAW), 2012 Fifth International Workshop on.
- Saunders, M., Lewis, P., & Thornhill, A. (2009). *Research Methods for Business Students* (Fifth ed.): Pearson Education Milited.
- Schäfer, T., Fettke, P., & Loos, P. (2012). Towards an Integration of GRC and BPM Requirements Changes for Compliance Management Caused by Externally Induced Complexity Drivers. In F. B. Daniel, Kamel; Dustdar, Schahram (Ed.), *Business Process Management Workshops* (Vol. 100, pp. 344-355): Springer Berlin Heidelberg.
- Shaft, T. M., & Vessey, I. (2006). The Role of Cognitive Fit in the Relationship between Software Comprehension and Modification. *MIS Quarterly*, *30*(1), 29-55. doi:10.2307/25148716
- Shanks, G. T., Elizabeth; Nuredini, Jasmina; Tobin, Daniel; Weber, Ron. (2008). Representing partwhole relations in conceptual modeling: an empirical evaluation. *MIS Quarterly*, 32(3), 553-573.
- Sharman, R., Kishore, R., & Ramesh, R. (2004). Computational ontologies and information systems: II. Formal specification. *Communications of the Association for Information Systems*, 184, 205.
- Smith, B. (2008). Ontology *The Blackwell Guide to the Philosophy of Computing and Information* (pp. 153-166): Blackwell Publishing Ltd.
- Sowa, J. F. (2000). Ontology, metadata, and semiotics. In B. Ganter & G. W. Mineau (Eds.), *Conceptual Structures: Logical, Linguistic, and Computational Issues, Proceedings* (Vol. 1867, pp. 55-81). Berlin: Springer-Verlag Berlin.
- Staab, S., Studer, R., Schnurr, H. P., & Sure, Y. (2001). Knowledge processes and ontologies. *Intelligent Systems, IEEE, 16*(1), 26-34. doi:10.1109/5254.912382
- Standards Australia. (2006). AS 3806-2006 Australian Standard Compliance Programs. Sydney, Australia.
- Stein, D. S., Wanstreet, C. E., & Glazer, H. R. (2011). Knowledge Building Online: The Promise and the Process. In C. X. W. Victor (Ed.), *Encyclopedia of Information Communication Technologies and Adult Education Integration* (pp. 985-998). Hershey, PA, USA: IGI Global.
- Studer, R., Benjamins, V. R., & Fensel, D. (1998). Knowledge engineering: Principles and methods. *Data & Knowledge Engineering*, 25(1–2), 161-197. doi:<u>http://dx.doi.org/10.1016/S0169-023X(97)00056-6</u>
- Suárez-Figueroa, M. C., Gómez-Pérez, A., & Fernández-López, M. (2012). The NeOn Methodology for Ontology Engineering. In M. C. G.-P. Suárez-Figueroa, Asunción; Motta, Enrico; Gangemi, Aldo (Ed.), Ontology Engineering in a Networked World (pp. 9-34): Springer Berlin Heidelberg.

- Sugumaran, V., & Storey, V. C. (2006). The role of domain ontologies in database design: An ontology management and conceptual modeling environment. *ACM Transactions on Database Systems*, *31*(3), 1064-1094. doi:10.1145/1166074.1166083
- Swartout, B., Patil, R., Knight, K., & Russ, T. (1996). *Toward distributed use of large-scale ontologies*. Paper presented at the AMCIS Proceedings.
- Sweller, J. (1988). Cognitive load during problem solving: Effects on learning. *Cognitive Science*, 12(2), 257-285. doi:http://dx.doi.org/10.1016/0364-0213(88)90023-7
- Syed Abdullah, N., Sadiq, S., & Indulska, M. (2012a). *Compliance Management Ontology v1.0*. Retrieved from
- Syed Abdullah, N., Sadiq, S., & Indulska, M. (2012b). A Compliance Management Ontology: Developing Shared Understanding through Models. In J. F. Ralyté, Xavier; Brinkkemper, Sjaak; Wrycza, Stanislaw (Ed.), Advanced Information Systems Engineering (Vol. 7328, pp. 429-444): Springer Berlin Heidelberg.
- Syed Abdullah, N., Sadiq, S., & Indulska, M. (2013). A Study of Ontology Construction: The Case of a Compliance Management Ontology *Ontology-Based Applications for Enterprise Systems and Knowledge Management* (pp. 276-291): IGI Global.
- Tamm, T., Seddon, P. B., Shanks, G., & Reynolds, P. (2011). How Does Enterprise Architecture Add Value to Organisations? *Communications of the Association for Information Systems*, 28, 141-168.
- Tan, F. B., & Hunter, M. G. (2002). The Repertory Grid Technique: A Method for the Study of Cognition in Information Systems. *MIS Quarterly*, 26(1), 39-57.
- Tan, M. (1994). Establishing Mutual Understanding in Systems Design: An Empirical Study. *Journal* of Management Information Systems, 10(4), 159-182. doi:10.2307/40398098
- Tempich, C., Pinto, H. S., Sure, Y., & Staab, S. (2005). An Argumentation Ontology for DIstributed, Loosely-controlled and evolvInG Engineering processes of oNTologies (DILIGENT). In A. E. Gómez-Pérez, Jérôme (Ed.), *The Semantic Web: Research and Applications* (Vol. 3532, pp. 241-256): Springer Berlin Heidelberg.
- Thompson, L., & Fine, G. A. (1999). Socially Shared Cognition, Affect, and Behavior: A Review and Integration. *Personality and Social Psychology Review*, *3*(4), 278-302. doi:10.1207/s15327957pspr0304_1
- Trochim, W. M., & Donnelly, J. P. (2008). Research Methods Knowledge Base: Atomic Dog Publishing.
- Uschold, M., & Jasper, R. (1999). A Framework for Understanding and Classifying Ontology Applications. Paper presented at the IJCAI-99 Workshop on Ontologies and Problem-Solving Methods: Lessons Learned and Future Trends, Amsterdam, The Netherlands.
- Uschold, M., & King, M. (1995). *Towards a methodology for building ontologies*. Paper presented at the Workshop on basic ontological issues in knowledge sharing.
- Uschold, M., King, M., Moralee, S., & Zorgios, Y. (1998). The Enterprise Ontology. *The Knowledge Engineering Review*, 13(01), 31-89. doi:doi:null

- Van den Bossche, P., Gijselaers, W. H., Segers, M., & Kirschner, P. A. (2006). Social and Cognitive Factors Driving Teamwork in Collaborative Learning Environments: Team Learning Beliefs and Behaviors. *Small Group Research*, 37(5), 490-521. doi:10.1177/1046496406292938
- Vessey, I., & Galletta, D. (1991). Cognitive Fit: An Empirical Study of Information Acquisition. *Information Systems Research*, 2(1), 63-84. doi:10.2307/23010613
- Wand, Y., Monarchi, D. E., Parsons, J., & Woo, C. C. (1995). Theoretical foundations for conceptual modelling in information systems development. *Decision Support Systems*, 15(4), 285-304. doi:<u>http://dx.doi.org/10.1016/0167-9236(94)00043-6</u>
- Wand, Y., Storey, V., & Weber, R. (1999). An ontological analysis of the relationship construct in conceptual modeling. ACM Transactions on Database Systems, 24(4), 494-528. doi:10.1145/331983.331989
- Wand, Y., & Weber, R. (1995). On the deep structure of information systems. *Information Systems Journal*, 5(3), 203-223. doi:10.1111/j.1365-2575.1995.tb00108.x
- Wand, Y., & Weber, R. (2002). Research Commentary: Information Systems and Conceptual Modeling—A Research Agenda. *Information Systems Research*, 13(4), 363-376. doi:10.1287/isre.13.4.363.69
- Weber, R. (2002). Ontological Issues in Accounting Information Systems. In V. Arnold & S. Sutton (Eds.), *Researching Accounting as an Information Systems Discipline* (pp. 13-33). Sarasota, Florida: American Accounting Association.
- Webster, J., & Watson, R. T. (2002). Analyzing the Past to Prepare for the Future: Writing a Literature Review. *MIS Quarterly*, 26(2), 13-23.
- Wildman, J. L., Salas, E., & Scott, C. P. R. (2014). Measuring Cognition in Teams: A Cross-Domain Review. Human Factors: The Journal of the Human Factors and Ergonomics Society, 56(5), 911-941. doi:10.1177/0018720813515907
- Wu, S., & Keysar, B. (2007). The Effect of Information Overlap on Communication Effectiveness. Cognitive Science, 31(1), 169-181. doi:10.1080/03640210709336989
- Yang, D., Miao, R., Wu, H. W., & Zhou, Y. T. (2009). Product configuration knowledge modeling using ontology web language. *Expert Systems with Applications*, 36(3), 4399-4411. doi:10.1016/j.eswa.2008.05.026

APPENDIX A. RECRUITMENT MATERIALS

A.1 RECRUITMENT SLIDE AND LEAFLET

Would you like to participate in our study and receive \$50 cash?

- · Get involved in our study of conceptual models in which:
 - You will be paired with another student to discuss some provided information.
 - You will be asked about that information and your interaction with the other student.
 - There are not correct or incorrect answers.
- All participants who complete the task will receive \$50 cash.
- Confidentiality is guaranteed and you may withdraw from the study at any point in time.
- · You don't need to have any special knowledge.
- The task will last no more than 1.5 hours.
- To participate or to find out more about the study please go to:
 <u>https://doodle.com/poll/uesursnkiazx4bup</u>
 - Or the Facebook page: "UQ CM Study"
 - Or contact Henry Roa at h.roamarin@uq.edu.au.



Figure A-1. Recruitment Slide

A.2 APPOINTMENT MAIL EXAMPLE

□ ↑ ↓ - Conceptual modeling study - Meeting 团 □ File Meeting ♀ Tell me what you want to do	/×
Henry Roa Marin Conceptual modeling study	5-Mar-16
When Wednesday, March 9, 2016 9:00 AM-10:30 AM Location 78-631 (GPS Building)	
Dear This is a confirmation that your participation has been booked in your available time. Please, be on time	e.
Date: Wednesday, 09/03/2016 Room: 78-631 (GPS Building) Time: 09:00 – 10:30	
If you cannot assist , please notify to h.roamarin@uq.edu.au immediately.	
A complete detail of the study is attached for your information.	
Many thanks.	
Kind regards,	
Henry Roa	
ipant_Information_	•

Figure A-2. Appointment Mail Example

A.3 PARTICIPANT INFORMATION AND CONSENT FORM

School of Informat HEAD OF SCHOOL Professor Paul Strooper	ion Technology and Electrical Engineering	The University of Queensian Brisbane Qid 4072 Australi Telephone +61 7 3365 2097 Facsimile +61 7 3365 4999 Email enquirlee@thee.uq.edu.au Internet www.ttee.uq.edu.au
	Participant Information	
Title:	Evaluation of Usefulness and Effective Providing Individual and Group Understandi	ness of Ontologies or ng
Investigator:	Mr Henry Roa, PhD Candidate School of Information Technology and Elect	rical Engineering
Supervisors:	Prof Shazia Sadiq School of Information Technology and Elect	rical Engineering
	A/Prof Marta Indulska UQ Business School	
Location:	Brisbane	
Part 1 Wi	hat does my participation involve?	
Part 1 Wi	hat does my participation involve?	
Part 1 WI 1 Introduction You are invited determine the u providing unders	hat does my participation involve? to take part in this research project. In this re sefulness of a particular kind of concept mod tanding at the individual and group level.	esearch project we aim to el, called an ontology, fo
Part 1 Wi 1 Introduction You are invited determine the u providing unders This Participant explains what is	hat does my participation involve? to take part in this research project. In this re- sefulness of a particular kind of concept mod- tanding at the individual and group level. Information Sheet/Consent Form tells you abou involved, to help you decide if you want to take	esearch project we aim to el, called an ontology, fo it the research project and part.
Part 1 WI 1 Introduction You are invited determine the u providing unders This Participant explains what is Please read this understand or wa	hat does my participation involve? to take part in this research project. In this re sefulness of a particular kind of concept modi tanding at the individual and group level. Information Sheet/Consent Form tells you abou involved, to help you decide if you want to take part s information carefully. Ask questions about ant to know more about.	esearch project we aim to el, called an ontology, fo it the research project and part. anything that you do no
Part 1 WI 1 Introduction You are invited determine the u providing unders This Participant explains what is Please read this understand or was Participation in ti to.	hat does my participation involve? to take part in this research project. In this re- sefulness of a particular kind of concept mod- tanding at the individual and group level. Information Sheet/Consent Form tells you abou- involved, to help you decide if you want to take p s information carefully. Ask questions about ant to know more about. his research is voluntary. If you do not wish to the	esearch project we aim to el, called an ontology, fo it the research project and part. anything that you do no take part, you do not have
Part 1 WI 1 Introduction You are invited determine the u providing unders This Participant explains what is Please read this understand or was Participation in ti to. If you decide yoo consent section.	hat does my participation involve? to take part in this research project. In this re sefulness of a particular kind of concept mod tanding at the individual and group level. Information Sheet/Consent Form tells you abou involved, to help you decide if you want to take p s information carefully. Ask questions about ant to know more about. his research is voluntary. If you do not wish to the u want to take part in the research project, you By signing it you are telling us that you:	esearch project we aim to el, called an ontology, fo it the research project and part. anything that you do no take part, you do not have u will be asked to sign the
Part 1 WI 1 Introduction You are invited determine the u providing unders This Participant explains what is Please read this understand or was Participation in the to. If you decide yo consent section. Understar Consent to Consent to Consent to Consent to	hat does my participation involve? hat does my participation involve? to take part in this research project. In this re sefulness of a particular kind of concept mod- tanding at the individual and group level. Information Sheet/Consent Form tells you about involved, to help you decide if you want to take point s information carefully. Ask questions about ant to know more about. his research is voluntary. If you do not wish to the u want to take part in the research project, you By signing it you are telling us that you: nd what you have read o take part in the research project o be involved in the procedures described o be audio recorded	esearch project we aim to el, called an ontology, fo it the research project and part. anything that you do no take part, you do not have a will be asked to sign the

Figure A-3. Participant Information and Consent form 1-5

2	What is the purpose of this research project?
The ont par cor cor	e purpose of this research project is to understand the usefulness and effectiveness of ologies on providing understanding, at the individual and group level. Ontologies are a ticular kind of conceptual models that represent a portion of specific domain through cepts and relations. We are looking for the main characteristics which allow these ceptual models to transfer knowledge about a particular domain effectively.
3	What does participation in this research project involve?
A s bas	hort introduction into the research will be conducted in lay terms. This will involve a ic introduction into what ontologies are and how they are used.
The loc witi par	e study will take place at The University of Queensland at a quiet, neutral, private ation (such as a conference room) where you will not be distracted. You will be paired in another student participant in order to discuss a particular topic related with a ticular domain, through the use of the ontology as a tool of discussion.
Pai you col	ts of the study will be audio recorded through the use of an iPad so the full value o ir comments can be used. You will be notified when this takes place. Other data will be ected through some questionnaires.
Yo agr	Ir involvement in the study will not exceed two hours of your time, and will occur at an eed upon time at your convenience.
4	What are the possible benefits to you?
Yoi Yoi ava	i will receive \$50 cash as thanks for your participation and reimbursement of your time may also learn about the results of the study (More information regarding this option is ilable in section 7).
lf par	rou withdraw from the experiment, you will be paid at a pro-rata rate for your ticipation (e.g. if you decide to withdraw after 1 hour, you will receive \$25).
5	What are the possible risks?
The bec the the res wit	There are no risks foreseen beyond those of normal living. In the unlikely event that you ome upset or distressed as a result of your participation in the research, please refer to last paragraph for contact details to speak with an officer of the University. You have right to stop any time you wish during the session, simply by informing either the earcher or his supervisor. You do not need to give any reasons for not participating or drawing from the study and your data will be excluded from the research.
6	Do I have to take part in this research project?

Figure A-4. Participant Information and Consent form 2-5

f vou decid	le to withdraw, please notify the researcher as soon as possible
/our docie	ion whathar to take part or part or to take part and then withdraw, will not affect
your relation	inship with the researchers, your studies, or with The University of Queensland.
7 How w	ill I be informed of the final results of this research project?
lf you are project. W Please che interested	interested, we can send you a copy of the final outcomes of this research e expect to start producing peer-reviewed papers in about a year from now eck the box in the signature section and provide an email address if you are n receiving a write-up of the results.
Part 2	How is the research project being conducted?
8 What	vill happen to information about me?
When info rather thar ID code to	rmation is collected from you, it will be collected with a Participant ID code with your name. No personal information is stored which links the Participan the participant.
Audio reco and any c identify you project. Ad any hard c	rding of this study has the potential to identify you by voice. This information ther information obtained in connection with this research project that car a, will remain confidential, and will only be used for the purpose of this research ditionally, the information will be kept on UQ password protected computers and opies will be stored in locked filing cabinets in the researcher's office.
The only p be used f publication	eople who will see the raw data will be the study researchers. The raw data wil or the present study alone with analyzed data being used in subsequent s/presentations.
In any sub way that yo	sequent publication and/or presentation, information will be provided in such a ou cannot be identified.
9 Canla	access research information kept about me?
In accorda laws, you f about you. would like f	nce with relevant Australian and/or Queensland privacy laws, and other relevant have the right to access the information collected and stored by the researchers Please contact one of the researchers named at the end of this document if you to access your information.
Please not after the p commence contribution	e, however, that the information collected about you will not be identifiable as articipant ID has been assigned. The participant ID will be assigned upon the ment of the experimental tasks. Access to information about your specific as after this point will not be possible.
10 Is this	research project approved?
The othice	aspects of this research project have been approved by the Ethics Committee
of the Sch Committee	of the University of Queensland.

Figure A-5. Participant Information and Consent form 3-5

11 Wh	o can I contact?				
If you w problem feelings 1186 or researc you are	ant any further info s which may be re of distress), you c at: <u>h.roamarin@uc</u> n team members, (welcome to offer fi	ormation conce lated to your in an contact the <u>gedu.au</u> . You either verbally o eedback at any	rning this proje volvement in tl principal resea will also be invi or in writing, at r time later.	ct or if you have ne project (for e: rcher, Henry Ro ted to offer feed the end of the s	any kample, a, on: 3365 back to the tudy and
Profess School shazia@ 3365 19	or Shazia Sadiq of Information Tech <u>≬itee.uq.edu.au</u> 99	nnology and El	ectrical Engine	ering	
A/Prof. UQ Bus <u>m.indul</u> 3346 8(Marta Indulska iness School <u>ka@business.uq.</u> ¢ 34	edu.au			
This stu Queens you are 336 51 Univers	dy adheres to the land and the Natio free to discuss yo 186 or at: h.roama ity not involved in t	e Guidelines oi onal Statemeni our participatio nrin@uq.edu.au he study, you i	f the ethical re t on Ethical Co n in this study I), if you would nay contact the	view process o onduct in Huma with project st d like to speak e Ethics Coordin	f The University of n Research. Whili aff (contactable of to an officer of th ator on 3365 3924

Figure A-6. Participant Information and Consent form 4-5

	Consent Form
	Consent Form
Title:	Evaluation of Usefulness and Effectiveness of Ontologies on Providing Individual and Grou Understanding
Investigator:	Mr Henry Roa School of Information Technology and Electrical Engineering
Supervisors:	Prof Shazia Sadiq School of Information Technology and Electrical Engineering
	A/Prof Marta Indulska
Location:	Brisbane
	Part in a
Declaration by	/ Participant:
I nave read the	r rancopant information Sneet or someone has read it to me in a language that i understand.
i understand tr	e purposes, procedures and risks of this research project as described within it.
i nave nad an	opportunity to ask questions and I am satisfied with the answers I have received.
I freely agree t understand that	o participate in this research project, as described within the Participant Information Sheet. It I can withdraw from the study at any point for any reason.
l understand ti responses to ti	nat during my participation of the study, an audio recording device will be used to record m ne experimental tasks.
I give permiss understand tha	ion to the researchers to use the collected information for the purpose of this research. It such information will remain confidential.
l understand th	at I will be given a signed copy of this document to keep.
I have been int	formed that I can contact the researcher if I want feedback on this study.
Participant's n	ame (printed)
Signature:	Date:
l would like to	receive a copy of a write-up of the results of this study.
🛛 Yes. Mye	mail address is 🛛 No
	y researcher:
Declaration b	a verbal explanation of the research project, its procedures and risks and I believe that th
Declaration b I have given a participant has	understood that explanation.
Declaration b I have given a participant has Researcher's r	understood that explanation.

Figure A-7. Participant Information and Consent form 5-5

APPENDIX B. ETHICAL CLEARANCE DOCUMENTS

	ETHICS CO	OMMITTEE	
	OUTCOME	OF REVIEW	
APPLICATION DETAI	LS:		
Reference:	EC201406ROA		
Name:	Henry N Roa		
Student Number:	43008254	15%	
Project litle:	and Group Understanding	nd Effectiveness of Ontolog	ies on Providing Individual
Principal Advisor:	Prof Shazia Sadiq, A/Prof N	1arta Indulska	
Date Received:			
Expedited Review:	No		
Date Reviewed:	26 August 2014		
OUTCOME OF REVIE	W:		
Approved	(Approval is for 12 Months)	Approved From:	Approved To: 25/8/2014
(subject to filing of		26/8/2014	
minor amendments)			
Request Amendment	it		
COMMENTS:			
My apologies for the d	elay on finalising processing of	this application -I am on S	SP and Emma's messages
fell between the cracks	5.		-
There are just a few re	latively small things to amend,	which I have outlined in th	e attached documents.
There are just a few re	latively small things to amend, tion on how you approach part	which I have outlined in th icipants. (1) Need lecturer	e attached documents. 'gatekeeper" approval
There are just a few re 1. Need more informat (keep documentation o	latively small things to amend, tion on how you approach part of that) and (2) lecturer should	which I have outlined in th icipants. (1) Need lecturer not provide you with stude	e attached documents. 'gatekeeper" approval ents' email addresses, but
There are just a few re 1. Need more informat (keep documentation of lecturer can forward at	latively small things to amend, tion on how you approach part of that) and (2) lecturer should n email from you to the studen	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be	e attached documents. 'gatekeeper" approval ents' email addresses, but given your email address
There are just a few re 1. Need more informat (keep documentation of lecturer can forward and so they can reply to yoo	latively small things to amend, tion on how you approach part of that) and (2) lecturer should n email from you to the studen u. II changes because your surfice	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address te potentially identifiable
There are just a few re 1. Need more informat (keep documentation of lecturer can forward ar so they can reply to yo 2. Just need a few sma However your protect	latively small things to amend, tion on how you approach part of that) and (2) lecturer should n email from you to the studen u. Il changes because your audio (ons of the audio data are accer	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan stable.	e attached documents. 'gatekeeper" approval Ints' email addresses, but given your email address ts potentially identifiable.
There are just a few re 1. Need more informat (keep documentation of lecturer can forward ar so they can reply to yo 2. Just need a few sma However your protect 3. See also a few furth	latively small things to amend, tion on how you approach part of that) and (2) lecturer should n email from you to the studen u. Il changes because your audio ons of the audio data are accep er small nits in the application f	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan stable. form and on the informatio	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet.
There are just a few re 1. Need more informat (keep documentation of lecturer can forward an so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few further	latively small things to amend, tion on how you approach part of that) and (2) lecturer should n email from you to the studen u. II changes because your audio (ons of the audio data are accepter er small nits in the application f	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan otable. form and on the informatio	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet.
There are just a few re 1. Need more informat (keep documentation of lecturer can forward ar so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few furth Please make the chang	latively small things to amend, tion on how you approach part of that) and (2) lecturer should nemail from you to the studen u. Il changes because your audio (ons of the audio data are accepter er small nits in the application for the our scored. Then we will car	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan otable. form and on the informatio re indicated in the two docu	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet. uments) and return the provid version of this
There are just a few re 1. Need more informat (keep documentation of lecturer can forward an so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few furthe Please make the chang documents to Emma for outcome document.	latively small things to amend, tion on how you approach part of that) and (2) lecturer should nemail from you to the studen u. II changes because your audio o ons of the audio data are accept er small nits in the application f ges (or accept the changes I hav or our records. Then we will ser	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan otable. form and on the informatio re indicated in the two docu nd you an unconditionally a	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet. uments) and return the pproved version of this
There are just a few re 1. Need more informat (keep documentation of lecturer can forward an so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few further Please make the chang documents to Emma for outcome document.	latively small things to amend, tion on how you approach part of that) and (2) lecturer should nemail from you to the studen u. II changes because your audio o ons of the audio data are accept er small nits in the application f ges (or accept the changes I hav or our records. Then we will ser	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan otable. form and on the informatio re indicated in the two doct ad you an unconditionally a	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet. uments) and return the ipproved version of this
There are just a few re 1. Need more informat (keep documentation of lecturer can forward an so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few further Please make the chang documents to Emma for outcome document. Penny Sanderson	latively small things to amend, tion on how you approach part of that) and (2) lecturer should nemail from you to the studen u. II changes because your audio o ons of the audio data are accept er small nits in the application f ges (or accept the changes I hav or our records. Then we will ser	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan otable. form and on the informatio re indicated in the two doct ad you an unconditionally a	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet. uments) and return the pproved version of this
There are just a few re 1. Need more informat (keep documentation of lecturer can forward an so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few further Please make the chang documents to Emma for outcome document. Penny Sanderson	latively small things to amend, tion on how you approach part of that) and (2) lecturer should nemail from you to the studen u. II changes because your audio o ons of the audio data are accept er small nits in the application f ges (or accept the changes I hav or our records. Then we will ser	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan otable. form and on the informatio re indicated in the two doct ad you an unconditionally a	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email address ts potentially identifiable. n and consent sheet. uments) and return the ipproved version of this
There are just a few re 1. Need more informat (keep documentation of lecturer can forward an so they can reply to yo 2. Just need a few sma However your protecti 3. See also a few furthor Please make the chang documents to Emma for outcome document. Penny Sanderson Signature:	latively small things to amend, tion on how you approach part of that) and (2) lecturer should nemail from you to the studen u. Il changes because your audio of ons of the audio data are accept er small nits in the application f ges (or accept the changes I hav or our records. Then we will ser	which I have outlined in th icipants. (1) Need lecturer ' not provide you with stude ts, and students should be data makes your participan stable. form and on the informatio re indicated in the two doc nd you an unconditionally a Date:	e attached documents. 'gatekeeper" approval ints' email addresses, but given your email addresses ts potentially identifiable. n and consent sheet. uments) and return the pproved version of this 26 August 2014

Figure B-1. Approved Ethical Clearance

SCHOOL OF INFORMATION TECHNOLOGY AND ELECTRICAL ENGINEERING

	OUTCOME	OF REVIEW	
APPLICATION DETAIL	LS:		
Reference:	EC201406ROA		
Name:	Henry N Roa		
Student Number:	43008254		
Project Title:	Evaluation of Usefulness ar and Group Understanding	nd Effectiveness of Onto	logies on Providing Individua
Principal Advisor:	Prof Shazia Sadiq, A/Prof M	larta Indulska	
Date Received:			
Expedited Review:	No		
Date Reviewed:	26 August 2014		
OUTCOME OF REVIE	W: (Approval is for 12 Months)	Approved From:	Approved To: 25/8/201
(subject to filing of		26/8/2014	
minor amendments)			
Request Amendmen	t		
The American Conference and	- DECEDC/MADE/		
Require Submission COMMENTS: My apologies for the defell between the cracks	elay on finalising processing of	this application —I am or	n SSP and Emma's messages
Require Submission COMMENTS: My apologies for the di fell between the cracks There are just a few rel Need more informat (keep documentation of lecturer can forward ar so they can reply to yoo J. Just need a few smal However your protectio S. See also a few further Please make the chang documents to Emma fo outcome document.	to BSSERC/MREC elay on finalising processing of 5. latively small things to amend, tion on how you approach parti of that) and (2) lecturer should in email from you to the studen u. Il changes because your audio of ons of the audio data are accep er small nits in the application f ges (or accept the changes I hav or our records. Then we will ser	this application –I am or which I have outlined in icipants. (1) Need lecture not provide you with stu ts, and students should I data makes your particip table. form and on the informa re indicated in the two d d you an unconditional	n SSP and Emma's messages the attached documents. er "gatekeeper" approval idents' email addresses, but be given your email address vants potentially identifiable. tion and consent sheet. ocuments) and return the by approved version of this
Require Submission COMMENTS: My apologies for the di fell between the cracks There are just a few rel Need more informat (keep documentation c lecturer can forward ar so they can reply to yoo Just need a few small However your protecti See also a few further Please make the chang documents to Emma fc outcome document. Penny Sanderson	elay on finalising processing of atively small things to amend, ion on how you approach parti of that) and (2) lecturer should n email from you to the studen u. Il changes because your audio of ons of the audio data are accep er small nits in the application f yes (or accept the changes I hav or our records. Then we will ser	this application –I am or which I have outlined in icipants. (1) Need lectur not provide you with stu ts, and students should I data makes your particip otable. form and on the informa re indicated in the two d id you an unconditional	n SSP and Emma's messages the attached documents. er "gatekeeper" approval idents' email addresses, but be given your email address vants potentially identifiable. tion and consent sheet. ocuments) and return the y approved version of this
Require Submission COMMENTS: My apologies for the di fell between the cracks There are just a few ref Need more informat (keep documentation of lecturer can forward ar so they can reply to yoo 2. Just need a few smal However your protecti 3. See also a few furthe Please make the chang documents to Emma fo outcome document. Penny Sanderson Signature:	elay on finalising processing of atively small things to amend, ion on how you approach parti of that) and (2) lecturer should nemail from you to the student u. II changes because your audio of ons of the audio data are accept er small nits in the application f yes (or accept the changes I hav or our records. Then we will ser	this application –I am or which I have outlined in icipants. (1) Need lectur not provide you with stu ts, and students should I data makes your particip stable. form and on the informa re indicated in the two d ad you an unconditionall Dat	n SSP and Emma's messages the attached documents. er "gatekeeper" approval idents' email addresses, but be given your email addresses pants potentially identifiable. tion and consent sheet. ocuments) and return the y approved version of this exe: 26 August 2014
Require Submission COMMENTS: My apologies for the di fell between the cracks There are just a few ref Need more informat (keep documentation of lecturer can forward ar so they can reply to yo 2. Just need a few smal However your protecti 3. See also a few furthe Please make the chang documents to Emma fo outcome document. Penny Sanderson Signature:	elay on finalising processing of i. latively small things to amend, iion on how you approach parti of that) and (2) lecturer should in email from you to the studen u. II changes because your audio of ons of the audio data are accep er small nits in the application f yes (or accept the changes I hav or our records. Then we will ser Chair, ITEE Ethics Commit	this application –I am or which I have outlined in icipants. (1) Need lectur not provide you with stu ts, and students should I data makes your particip otable. form and on the informa re indicated in the two d ad you an unconditional Dat	n SSP and Emma's messages the attached documents. er "gatekeeper" approval idents' email addresses, but be given your email address vants potentially identifiable. tion and consent sheet. ocuments) and return the y approved version of this e: 26 August 2014

Figure B-2. Approval of Ethical Clearance Amendment

	ETHICS CO	OMMITTEE	
	OUTCOME	OF REVIEW	
APPLICATION DETAI	LS:		
Reference:	EC201509ROA_B		
Name:	Henry Roa		
student Number:	43008254	1-11-11-11-11-11-11-11-11-11-11-11-11-1	
Project Title:	and Group Understanding	nd Effectiveness of Ontologi	es on Providing Individual
Principal Advisor:	Prof Shazia Sadiq		
-	A/Prof Marta Indulska		
Date Pereived	08-02-2016		
Fynedited Deview:	No		
Date Reviewed	17-02-2016		
	(Approval is far 12 Marsha)	Approved From:	Approved To:
K Approved	(Approval is for 12 months)	Approved From.	Approved 10.
V	•	01-03-2010	23-01-2017
A Request Amenamer	it .		
Require Submission	to BSSERC/MREC		
COMMENTS:			
Your ethics application	n #EC201509ROA_B has been ap	pproved, provided the follow	ving changes are made:
 Include media 	a in the Bastinian to Informati		T in the encoderant
 Include wordling 	ng in your Participant information	on sneet and at items 3 and s from the study, they will b	5 in the amendment
for their partie	in that if a participant withdraws	ithdraw after 1 hour, they will b	e paiù al a pro-rata rate vill rocoivo \$25)
for their partic	apation (e.g. in they decide to w	indiawarter 1 nour, they v	viir receive \$257
Please submit final ver	rsions of the forms to ITEE ethic	s office – <u>rao@itee.uq.edu.</u>	au
			_
the second is far also do			
Approval is for the dur	ation		
From: 01-03-2016			
10.23-01-2017			
Prot Penelope Sanders	5011		
	0.0	Date:	29/02/2016
signature: Rem	down		
Signature: Rem	dge hand		

Figure B-3. Approval of Ethical Clearance Amendment

APPENDIX C. EXPERIMENTAL TRAINING MATERIALS

C.1 TRAINING SECTION FOR TREATMENT GROUP

In this task, you will be introduced to a particular approach for representing information about a domain or topic. The key concepts used in the approach are described below.

A class (represented by a rectangle) is the name of a concept in a particular domain. For example, *Student*, *University* and *Course*, are classes of the Student Registration domain. Figure 1 shows the representation of the *Student* class.



Figure C-1. Example of Student class

A class is described by attributes called properties that describe the class itself. These properties are mentioned within the class. For example, *studentName* and *dateOfBirth* are the properties of the class *Student* – see Figure 2.



Figure C-2. Properties that describe the student class

Relationships relate one class to another class and are represented through arrows, for example the relationship *takes* connects the *Student* class with the *Course* class – see Figure 3. Figure 3 also shows another relationship, *isTakenBy*, which expresses that 'Course is taken by Student'.



Figure C-3. Relationships

A subclass describes a specialization of the concept of class. A white arrowhead indicates the subclass belongs to its parent class. The arrow originates from the subclass and points towards its parent class. For example, in Figure 4, *Student* is a subclass of *Person*.



Figure C-4. Student Subclass

The properties of a class are inherited by its subclass. For example, in Figure 4, *Student* inherits the *dateOfBirth* property belonging to *Person*.

The relationships of a class are also inherited by its subclass, but not explicitly represented on the model. For example, in Figure 4 a *person* uses *transportation*, thus a *student* also uses *transportation* because *Student* is a subclass of *Person*.

To demonstrate how information of a domain is represented using the above approach, we will use a specific example. Consider the situation that a student holds a library membership (Figure 5). *Student*, *Membership*, and *Library* are shown as three classes and their relationships are indicated by arrows between these classes. A student can borrow books; thus the *Student* class is related to the *Book* class by a relationship *borrows*. Also, a book can be borrowed by students, thus the *Book* class is related to the *Student* class by a relationship *isBorrowedBy*.



Figure C-5. Library subscription domain

All the classes have internal properties. For example, a library has a name and is located in a particular address. Another example is that a book has its own title, access number, and author.

A graduate student is a student who has taken some graduate courses. Thus, *Graduate Student* is a subclass of the *Student* class. Because it is a subclass, all the properties of the *Student* class are inherited by the *Graduate Student* class as shown on Figure 5. Similarly, all the relationships belonging to the *Student* class are inherited by the *Graduate Student* subclass. For example, a graduate student borrows book in the same way that a student borrows book.

C.2 PRACTICE QUESTIONS

In this experiment, you will be asked to answer true/false statements and provide an explanation of your answer by viewing some diagrams. These are practice statements to follow to familiarize you with the type of questions.

Using the information provided answer the following statements and write your answers on the form.

1. Every final itinerary has a reservation.

- □ True
- □ False

 \Box I don't know

Which part of the documentation or what information allowed you to come to your answer:

2. A service provider is involved in preparing initial itineraries.

- □ True
- □ False
- □ I don't know

Which part of the documentation or what information allowed you to come to your answer:

3. Any person can make a reservation.

- □ True
- □ False

□ I don't know

Which part of the documentation or what information allowed you to come to your answer:

4. Preparing final itineraries involves a service provider.

□ True

□ False

□ I don't know

Which part of the documentation or what information allowed you to come to your answer:

5. Preparing final itineraries involves a travel agent.

□ True

□ False

□ I don't know

Which part of the documentation or what information allowed you to come to your answer:

6. Every initial itinerary has a reservation.

□ True

□ False

 \Box I don't know

Which part of the documentation or what information allowed you to come to your answer:

- 7. Every itinerary has a departure date and return date.
- □ True
- □ False
- \Box I don't know

Which part of the documentation or what information allowed you to come to your answer:

- 8. Customers can participate in the elaboration of the initial itinerary.
- □ True
- □ False
- \Box I don't know

Which part of the documentation or what information allowed you to come to your answer:

C.3 TRAVEL DOMAIN ONTOLOGY



Figure C-6. Travel Domain Ontology

C.4 TRAVEL DOMAIN DESCRIPTION

"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."

C.5 GROUP TASK SECTION

In this section, first, you will complete a task individually and then, you will interact with your partner. Your interaction will be audio recorded to help with analysis. Please complete the following tasks.

Imagine you work as a compliance management expert in an organization. Your manager wants
to have an idea of what issues, tasks or activities the company should consider in order to improve
compliance management. <u>Based on the information that you have</u>, list six of the most important
issues that you think the organization should consider in order to improve compliance
management. Write your answers in the following table along with an explanation of your
answers.

Issue to consider	Explanation
A.	
B.	
C.	
D.	
E.	
F.	

2. Participant A reads and explains his/her answer to his/her partner. Participant B completes the following table. If there is any disagreement on the answers, participants discuss the disagreement and write down any agreement on the table.

Partner's issue	Do you agree with your partner's issue?	Do you agree with your partner's explanation?	What was the agreement after the discussion?
1.	Yes □ No □	Yes □ No □ Partially □	
2.	Yes □ No □	Yes □ No □ Partially □	
3.	Yes □ No □	Yes □ No □ Partially □	
4.	Yes □ No □	Yes □ No □ Partially □	
5.	Yes □ No □	Yes □ No □ Partially □	
6.	Yes □ No □	Yes □ No □ Partially □	

- 3. Change the roles and complete the previous activity.
- 4. Return this sheet to the moderator.

APPENDIX D. DATA COLLECTION INSTRUMENT

D.1 INSTRUMENT FOR THE TREATMENT GROUP

OF QUEENSLAND
EXPERIMENT
During this experiment, you will be asked to work on different tasks and questions grouped into five sections:
 Pre-test questions Practice questions Individual task Group task Final questions
Please read the instructions carefully and complete the required tasks.
0% 100% Continue

Figure D-1. Screenshot of the Experiment Introduction

THE UNIVERSITY OF QUEENSLAND	
SECTION 1: PRE-TEST QUESTIONS	
Complete this section individually.	
Participant Profile	
Fill in the blanks or select the most appropriate answer.	
1. Age:	
16 – 25	
25 - 35	
Above 35	
2 Conder	
- Formalo	
3. Type of student:	
 Undergraduate 	
Coursework postgraduate	
Research postgraduate	
4. Area of study:	
Agribusiness, Agriculture, Environment and Science	
Business, Economics and Law	
Engineering, Architecture and Planning, and Information Technology	
Health	
Humanities, Education, Psychology and Music	
5 Year of study in your current degree:	
First vear	
Second year	
Third year	
Fourth year	
Fifth year	
6. Native language:	
English	
Other	
0% 100%	
	Submit

Figure D-2. Screenshots of the Pre-Test Questions

THE UNIN OF QUEEN	/ERSITY NSLAND							
Conceptual Modeling Knowledge								
Select the option that best represents you.								
1. What is the extent of your knowledge of data modeling concepts such as entities, classes, relationships, and properties?								
None	Very low	Low	Average	Moderate	High	Very high		
0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	•		
2. Over the last two years, to what extent do you have experience in using data modeling concepts such as entities, classes, relationships, and properties?								
Never	Hardly ever	Rarely	Sometimes	Often	Frequently	Usually		
0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc		
		0%		100%		Submit		

Figure D-3. Screenshot of the Conceptual Modeling Knowledge Questions



Figure D-4. Screenshot of the Training Section



Figure D-5. Screenshot of the Practice Questions Section



Figure D-6. Screenshot of the Practice Question 1



Figure D-7. Screenshot of the Practice Question 2





Figure D-8. Screenshot of the Practice Question 3



Figure D-9. Screenshot of the Practice Question 4


Figure D-10. Screenshot of the Practice Question 5

THE UNIVERSITY OF QUEENSLAND
6. Every initial itinerary has a reservation.
● True
False
I don't know
Which part of the documentation or what information allowed you to come to your answer:



Figure D-11. Screenshot of the Practice Question 6



Figure D-12. Screenshot of the Practice Question 7



Figure D-13. Screenshot of the Practice Question 8



SECTION 3: INDIVIDUAL TASK

Compliance Management

In this section, the moderator will provide you with information about the compliance management domain that describes how organizations can meet their obligations.

For each of these statements, explain whether they are true or false and provide your justification for your answer on the form.

Please respond as best as you can by using the provided information and without using your prior knowledge.





THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 Organizations implement a compliance management system to enable compliance management. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
0% 100% Submit

Figure D-15. Screenshot of the Compliance Management Question 1

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 2. Organizations engage in compliance management to aid risk management. True False 	
I don't know Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit



Figure D-16. Screenshot of the Compliance Management Question 2

Figure D-17. Screenshot of the Compliance Management Question 3

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 4. Organizations engage in compliance management to ensure they comply with obligations. True 	
 Faise I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit



Figure D-18. Screenshot of the Compliance Management Question 4

Figure D-19. Screenshot of the Compliance Management Question 5

OF QUEENSLAND	
 6. A compliance commitment (or voluntary obligation) of an organization is an example of an obligation type. True False 	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit

Figure D-20. Screenshot of the Compliance Management Question 6

OF QUEENSLAND	
 7. Organizations should identify, analyze and evaluate risks to perform a risk assessment. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit

Figure D-21. Screenshot of the Compliance Management Question 7

OF QUEENSLAND	
 8. Monetary resources (or financial resources) and human resources (non-monetary resources) a required for compliance management in organizations. True False 	are
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit

Figure D-22. Screenshot of the Compliance Management Question 8

OF QUEENSLAND	
 9. Organizations through their compliance management systems ensure the competency and training of organization staff. True 	
False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0% 100% Submit	t

Figure D-23. Screenshot of the Compliance Management Question 9

THE UNIVERSITY OF QUEENSLAND	
10. Compliance management systems register and report documented information Information True	ation of organizations.
False	
I don't know	
Which part of the documentation or what information allowed you to come to y	our answer:
0%	Submi



Figure D-24. Screenshot of the Compliance Management Question 10

Figure D-25. Screenshot of the Compliance Management Question 11

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
12. A process (or business process) has related risks.	
 True 	
○ False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit



Figure D-26. Screenshot of the Compliance Management Question 12

Figure D-27. Screenshot of the Compliance Management Question 13

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 14. The treatment of a risk does not include a description of the applied treatment nor the risk. True False 	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit



OF QUEENSLAND
SECTION 4: GROUP TASK
In this section, first, you will complete a task individually and then, you will interact with your partner. Your interaction will be audio recorded to help with analysis.
Please follow the moderator instructions and then, go back to this survey.
0%
Submit

Figure D-29. Screenshot of the Group Task Section



SECTION 5: FINAL QUESTIONS

Compliance Management

For each of these statements, explain whether they are true or false by using the information provided by the moderator. Write your explanations on the form. Please respond as best as you can by using the provided information and not with your prior knowledge.

Also, respond what would be the answers of your partner if you think you know those answers..



Submit

Figure D-30. Screenshot of the Final Question Section	
	-

THE UNIVERSITY OF QUEENSLAND
1. To enable compliance management, organizations implement a compliance management system.
True
False
I don't know
Which part of the documentation or what information allowed you to come to your answer:
What do you think would be the answer of your partner?
My partner would answer "True"
My partner would answer "False"
My partner wouldn't know the answer
I don't know what my partner would answer
0%
Submit

Figure D-31. Screenshot of the Compliance Management Question 1

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 2. To aid risk management, organizations engage in compliance management. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer	
I don't know what my partner would answer 0%	Submit



THE UNIVERSITY OF QUEENSLAND	
 3. To enable compliance management, organizations need to provide sufficient resources. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer	
I don't know what my partner would answer 0% 100%	Submit

Figure D-33. Screenshot of the Compliance Management Question 3

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 4. To ensure organizations comply with obligations, they engage in compliance management. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



THE UNIVERSITY OF QUEENSLAND AVSTRALIA	
 5. An example of an obligation type is a compliance requirement (or mandatory requirement). True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	1
What do you think would be the answer of your partner?	
 My partner would answer "False" My partner would answer "False" 	
 My partner wouldn't know the answer 	
I don't know what my partner would answer	
0%	Submit

Figure D-35. Screenshot of the Compliance Management Question 5

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 6. An example of an obligation type is a compliance commitment (or voluntary obligation). True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



OF QUEENSLAND	
 7. To perform a risk assessment, organizations should identify, analyze and evaluate risks. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	1
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "Eales" 	
 My partner would answer My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit

Figure D-37. Screenshot of the Compliance Management Question 7

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 8. Compliance management in organizations requires monetary resources (or financial resources) and human resources (non-monetary resources). True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:]
	**
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	omit



The University Of Queensland	
AUSTRALIA	
The competency and training of employees are ensured by the organization through their compliance management systems.	
True	
False	
○ I don't know	
Which part of the documentation or what information allowed you to some to your answer:	
	//
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0% 100%	
	Cubmit
	Supmit

Figure D-39. Screenshot of the Compliance Management Question 9

OF QUEENSLAND	
 10. Documented information of organizations is registered and reported by compliance manag systems. True False I don't know 	ement
Which part of the documentation or what information allowed you to come to your answer:	4
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



OF QUEENSLAND	
11. To check compliance in organizations, compliance management systems establish all the necessary controls.	
True	
False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
	1.
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	
	Submit
	Submit

Figure D-41. Screenshot of the Compliance Management Question 11

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 12. A process (or business process) includes related risks. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner? My partner would answer "True"	
 My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
	Submit



THE UNIVERSITY OF QUEENSLAND	
 13. The compliance culture of an organization specifies behavioral norms. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False"	
 My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit

Figure D-43. Screenshot of the Compliance Management Question 13

OF QUEENSLAND	
 14. The treatment of a risk does not consider a description of the applied treatment nor the ris True False I don't know 	k.
Which part of the documentation or what information allowed you to come to your answer:	æ
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



THE UNIVE OF QUEENS	ERSITY SLAND					
Communicat	ion Openness	;				
Select the opt	ion that best re	epresents you.				
1. It was easy	to communica	te openly with	my partner.			
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
•		•	0		0	
2. Communica	ation was very	open.				
Strongly	,	Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
•	\bigcirc	•	\odot	\bigcirc	\odot	
3. When we c	ommunicated	with each othe	er in this group, t	here was a gre	eat deal of und	derstanding.
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
0		•				
4. It was easy	to ask advice	from my partn	er.			
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
0	0	0	0	0	0	0
				1008/		
		0%		100%		
						Submit

Figure D-45. Screenshot of the Communication Openness Questions

THE UNIVE OF QUEENS	ERSITY SLAND					
Discussion E	fficiency					
Select the opt	ion that best re	epresents you.				
1. To what ext	ent would you	agree that this	s group discussio	on was result o	riented?	
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
0	\bigcirc	\bigcirc	\bigcirc	\bigcirc		\bigcirc
2. The time sp	ent in the grou	p discussion v	was efficiently us	sed.		
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
0		\bigcirc	\odot	\bigcirc	\bigcirc	\bigcirc
3. Issues raise	ed in the group	interaction we	ere discussed th	oroughly.		
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
0		\bigcirc	\odot	\odot		\odot
		0%		100%		
		U /o		100%		
						Submit

Figure D-46. Screenshot of the Discussion Efficiency Questions

THE UNIV OF QUEEN	/ERSITY NSLAND					
Discussion	Effectiveness					
Select the op	tion that best re	presents you.				
1. The discus	ssions were effe	ctive.				
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
0						
2. The conte	xt of the discuss	ions was care	fully developed.			
Strongly	Disagree	Somewhat	Neither agree	Somewhat	Agree	Strongly
disagree	Disagree	uisagi ee		ayree	Agree	agree
	0	0	0	0	0	0
3 Jesuios wo	ro not oxaminod	offectively				
Strongly	e not examined	Somowhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
0	0	0	0	0	0	0
1 Participat	ion in the discus	cions was oar	scietontly distribu	utod		
4. Farticipat		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
0	0	0	0	0	0	•
5. Ideas in t	he discussions v	vere critically	examined.			
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
•	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
6. The amo	unt of information	n exchanged v	was sufficient.			
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
						\odot
		0%		100%		
						Submit

Figure D-47. Screenshot of the Discussion Effectiveness Questions

THE UNIV OF QUEEN	/ERSITY ISLAND					
Communica	tion Clarity					
Select the op	tion that best re	presents you.				
1. How clear	were the messa	ges of your pa	artner? Explain	your answer.		
Not at all	Very unclear	Lowly clear	Slightly clear	Moderately clear	Very clear	Highly clear
0		0	0	\bigcirc		0
2. How often	were the messa	iges of your pa	artner unclear?	Explain your	answer.	
Never	Hardly ever	Rarely	Sometimes	Usually	Almost always	Always
•			\bigcirc			
		0%		100%		Submit

Figure D-48. Screenshot of the Communication Clarity Questions

THE UNIVE	RSITY
OF QUEENS	LAND
	We thank you for your time spent taking this survey. Your response has been recorded.

Figure D-49. Screenshot of the End of the Experiment

D.2 INSTRUMENT FOR THE CONTROL GROUP

OF QUEENSLAND AUSTRALIA
EXPERIMENT
During this experiment, you will be asked to work on different tasks and questions grouped into five sections:
 Pre-test questions Practice questions Individual task Group task Final questions
Please read the instructions carefully and complete the required tasks.
0% 100%
Continue

Figure D-50. Screenshot of the Experiment Introduction

SECTION 1: PRE-TEST QUESTIONS

Complete this section individually.

Participant Profile

Fill in the blanks or select the most appropriate answer.

1. Age:
16 – 25
25 – 35
Above 35
2. Gender:
Male
Female
3. Type of student:

- Undergraduate
- Coursework postgraduate
- Research postgraduate

4. Area of study:

Agribusiness, Agriculture, Environment and Science	
Business, Economics and Law	
Engineering, Architecture and Planning, and Information Technology	
Health	
Humanities, Education, Psychology and Music	
5. Year of study in your current degree:	
First year	
Second year	
Third year	
Fourth year	
Fifth year	
6. Native language:	
English	
Other	
0% 100%	
	Submit

Figure D-51. Screenshots of the Pre-Test Questions



Conceptual Modeling Knowledge

Select the option that best represents you.

1. What is the extent of your knowledge of data modeling concepts such as entities, classes, relationships, and properties?

None	Very low	Low	Average	Moderate	High	Very high
\bigcirc						

2. Over the last two years, to what extent do you have experience in using data modeling concepts such as entities, classes, relationships, and properties?

Never	Hardly ever	Rarely	Sometimes	Often	Frequently	Usually
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
		0%		100%		
						Submit

Figure D-52. Screenshot of the Conceptual Modeling Knowledge Questions



	Management Ki	nowledge				
Select the op	tion that best rep	resent you.				
1. What is the	e extent of your k	nowledge in	compliance mar	nagement con	cepts?	
None	Very low	Low	Average	Moderate	High	Very high
	0					
2. Over the la management	ast two years, to v t concepts?	what extent o	do you have exp	erience with a	pplying complia	nce
Never	Hardly ever	Rarely	Sometimes	Often	Frequently	Usually
0	\odot	\bigcirc		\bigcirc	\bigcirc	\bigcirc
3. What is the	e extent of your k	nowledge of	risk manageme	nt concepts?		
None	Very low	Low	Average	Moderate	High	Very high
	0	\bigcirc	0	\bigcirc	0	0
4. Over the la concepts? Never	ast two years, to v Hardly ever	what extent o Rarely	do you have exp Sometimes	erience with a Often	pplying risk mai Frequently	Usually
5. What is the None	e extent of your k Very low	nowledge o	f Business Proce Average	ess Managem Moderate	ent concepts? High	Very high
					•	
6. Over the la Management	ast two years, to t concepts?	what extent	do you have exp	perience with a	applying Busine	ess Process
Never	Hardly ever	Rarely	Sometimes	Often	Frequently	Usually
		0	\odot			
		0%		100%		

Figure D-53. Screenshot of the Compliance Management Knowledge Questions



SECTION 2: PRACTICE QUESTIONS

1. PRACTICE QUESTIONS

In this experiment, you will be asked to answer true/false statements and provide an explanation of your answer by viewing some information. You will have practice statements to familiarize you with the type of questions.

The following text is a description of a travel domain, which provides information about the organization of a travel plan. Using the information below to answer the following statements and write your answers on the form.

"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."







Figure D-55. Screenshot of the Practice Question 1

THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 2. A service provider is involved in preparing initial itineraries. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."
0% 100% Continue

Figure D-56. Screenshot of the Practice Question 2

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 3. Any person can make a reservation. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings airline tickets and entertainment tickets. A customer has travel preferences including departure and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called in itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called fin itinerary. The reservation is a contract between the customer and the service provider. The trav agent does reservations when the final itinerary is complete and the customer is satisfied with t itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."	s, time nitial al el he
0%	Continue

Figure D-57. Screenshot of the Practice Question 3

THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 4. Preparing final itineraries involves a service provider. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."
Continue



THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 5. Preparing final itineraries involves a travel agent. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."
0% 100% Continue

Figure D-59. Screenshot of the Practice Question 5

THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 6. Every initial itinerary has a reservation. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."
0% 100% Continue



THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 7. Every itinerary has a departure date and return date. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."
0% 100% Continue

Figure D-61. Screenshot of the Practice Question 7

THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 8. Customers can participate in the elaboration of the initial itinerary. True False I don't know
Which part of the documentation or what information allowed you to come to your answer:
"A travel agent helps customers to organize travel plans. Travel agents organize hotel bookings, airline tickets and entertainment tickets. A customer has travel preferences including departure time and return date. A travel agent proposes a travel itinerary to a person. Such itinerary is called initial itinerary. A travel itinerary is complete when the reservation is made. Such itinerary is called final itinerary. The reservation is a contract between the customer and the service provider. The travel agent does reservations when the final itinerary is complete and the customer is satisfied with the itinerary. The customer's preference is composed of the preference values for hotel, airline and entertainment."
0% 100% Continue



OF QUEENSLAND
SECTION 3: INDIVIDUAL TASK
Compliance Management
In this section, the moderator will provide you with information about the compliance management domain that describes how organizations can meet their obligations.
For each of these statements, explain whether they are true or false and provide your justification for your answer on the form.
Please respond as best as you can by using the provided information and without using your prior knowledge.
0%



AUSTRALIA			
1. Organizations imple	ment a compliance managem	ient system to enable co	mpliance management.
True			
False			
I don't know			
Which part of the docu	mentation or what informatior	n allowed you to come to	your answer:



OF QUEENSLAND	
 2. Organizations engage in compliance management to aid risk management. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit

Figure D-65. Screenshot of the Compliance Management Question 2

THE UNIVERSITY OF QUEENSLAND		
 Organizations need to prov True False 	ide sufficient resources to enable compliance management.	
I don't know		
Which part of the documentat	ion or what information allowed you to come to your answer:	
	0%	Submit



THE UNIVERSITY OF QUEENSLAND	
 4. Organizations engage in compliance management to ensure they comply with obligations. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	Z
0%	Submit



THE UNIVERSITY OF QUEENSLAND	
 5. A compliance requirement (or mandatory obligation) of an organization is an example of a obligation type. True 	n
False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	



OF QUEENSLAND	
 6. A compliance commitment (or voluntary obligation) of an organization is an example of an obligation type. True 	
False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	Å
0%	Submit

Figure D-69. Screenshot of the Compliance Management Question 6

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 7. Organizations should identify, analyze and evaluate risks to perform a risk assessment. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 8. Monetary resources (or financial resources) and human resources (non-monetary resources) a required for compliance management in organizations. True False 	ire
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
	11
0%	
5	Submit

Figure D-71. Screenshot of the Compliance Management Question 8
THE UNIVERSITY OF QUEENSLAND AUSTRALIA
 9. Organizations through their compliance management systems ensure the competency and training of organization staff. True
False
I don't know
Which part of the documentation or what information allowed you to come to your answer:
0% 100% Submi



OF QUEENSLAND	
 10. Compliance management systems register and report documented information of organiza True False I don't know 	itions.
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit

Figure D-73. Screenshot of the Compliance Management Question 10

OF QUEENSLAND	
 11. Compliance management systems establish all controls necessary to check compliance organizations. True 	in
False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	



OF QUEENSLAND	
12. A process (or business process) has related risks. ◎ True	
False	
I don't know	
Which part of the documentation or what information allowed you to come to your answer:	
0%	Submit

Figure D-75. Screenshot of the Compliance Management Question 12

OF QUEENSLAND	
 13. The compliance culture of an organization specifies behavioral norms. True False 	
 I don't know Which part of the documentation or what information allowed you to come to your answe 	r:
	į.
0%	Subn



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 14. The treatment of a risk does not include a description of the applied treatment nor the risk. ■ True ■ True 	
 False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
0%	
	Submit

Figure D-77. Screenshot of the Compliance Management Question 14





OF QUEENSLAND	
SECTION 5: FINAL QUESTIONS	
Compliance Management	
For each of these statements, explain whether they are true or false by using the information provided by the moderator. Write your explanations on the form. Please respond as best as you can by using the provided information and not with your prior knowledge.	
Also, respond what would be the answers of your partner if you think you know those answers	
0%	
	Submit

Figure D-79. Screenshot of the Final Question Section

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 To enable compliance management, organizations implement a compliance management sy True False I don't know 	rstem.
Which part of the documentation or what information allowed you to come to your answer:	10
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False"	
 My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 2. To aid risk management, organizations engage in compliance management. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	
	Submit

Figure D-81. Screenshot of the Compliance Management Question 2

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 3. To enable compliance management, organizations need to provide sufficient resources. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	Submit



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 4. To ensure organizations comply with obligations, they engage in compliance management. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	1
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	Submit

Figure D-83. Screenshot of the Compliance Management Question 4

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 5. An example of an obligation type is a compliance requirement (or mandatory requirement). True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	6
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	
	Submit



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 6. An example of an obligation type is a compliance commitment (or voluntary obligation). True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	1
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	Submit

Figure D-85. Screenshot of the Compliance Management Question 6

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 7. To perform a risk assessment, organizations should identify, analyze and evaluate risks. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	Submit



OF QUEENSLAND	
 8. Compliance management in organizations requires monetary resources (or financial resource and human resources (non-monetary resources). True False I don't know 	ces)
Which part of the documentation or what information allowed you to come to your answer:	
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer	
I don't know what my partner would answer 0%	Submit

Figure D-87. Screenshot of the Compliance Management Question 8

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 9. The competency and training of employees are ensured by the organization through their compliance management systems. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	1
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



OF QUEENSLAND	
 10. Documented information of organizations is registered and reported by compliance manage systems. True False I don't know 	ement
Which part of the documentation or what information allowed you to come to your answer:	
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit

Figure D-89. Screenshot of the Compliance Management Question 10

THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 11. To check compliance in organizations, compliance management systems establish all the necessary controls. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 12. A process (or business process) includes related risks. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	li.
What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False"	
 My partner wouldn't know the answer I don't know what my partner would answer 	
0%	Submit

Figure D-91. Screenshot of the Compliance Management Question 12

OF QUEENSLAND	
 13. The compliance culture of an organization specifies behavioral norms. True False I don't know 	
Which part of the documentation or what information allowed you to come to your answer:	li
What do you think would be the answer of your partner?	
My partner would answer "True"	
My partner would answer "False"	
My partner wouldn't know the answer	
I don't know what my partner would answer	
0%	Submit



THE UNIVERSITY OF QUEENSLAND AUSTRALIA	
 14. The treatment of a risk does not consider a description of the applied treatment nor the risk True False I don't know 	ς.
Which part of the documentation or what information allowed you to come to your answer:	<i>i</i>
 What do you think would be the answer of your partner? My partner would answer "True" My partner would answer "False" My partner wouldn't know the answer I don't know what my partner would approver 	
	Submit

Figure D-93. Screenshot of the Compliance Management Question 14



Communication Openness

Select the option that best represents you.

1. It was easy t	o communica	te openly with	mv partner.			
Strongly disagree	Disagree	Somewhat disagree	Neither agree or disagree	Somewhat agree	Agree	Strongly agree
0	0	0	0	0	0	0
2. Communicat	tion was very o	open.				
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
3. When we co	ommunicated v	with each othe	er in this group, t	here was a gre	at deal of und	lerstanding.
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
4. It was easy t	o ask advice f	rom my partne	er.			
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	or disagree	agree	Agree	agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
		0%		100%		
		0.0				
						Submit

Figure D-94. Screenshot of the Communication Openness Questions



Discussion Efficiency

Select the option that best represents you.

1. To what extent would you agree that this group discussion was result oriented?						
Strongly	Disagraa	Somewhat	Neither agree	Somewhat	Agroo	Strongly
uisayiee	Disagree	uisayiee	nor usagree	ayree	Agree	ayree
\bigcirc	\bigcirc	\bigcirc		\bigcirc		0
2. The time spe	ent in the grou	p discussion v	vas efficiently us	sed.		
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
3. Issues raise	d in the group	interaction we	ere discussed the	oroughly.		
Strongly		Somewhat	Neither agree	Somewhat		Strongly
disagree	Disagree	disagree	nor disagree	agree	Agree	agree
	\bigcirc		\bigcirc	\bigcirc	\bigcirc	\bigcirc
		0%		100%		
						Submit

Figure D-95. Screenshot of the Discussion Efficiency Questions



Discussion Effectiveness

Select the option that best represents you.

1. The discussions were effective.

Strongly disagree	Disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Agree	Strongly agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

2. The context of the discussions was carefully developed.

Strongly disagree	Disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Agree	Strongly agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

3. Issues were not examined effectively.

Strongly disagree	Disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Agree	Strongly agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc		\bigcirc

4. Participation in the discussions was consistently distributed.

Strongly disagree	Disagree	Somewhat disagree	Neither agree nor disagree	Somewhat agree	Agree	Strongly agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\odot	

5. Ideas in the Strongly disagree	discussions w Disagree	vere critically e Somewhat disagree	xamined. Neither agree nor disagree	Somewhat agree	Agree	Strongly agree
\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	•
6. The amount Strongly disagree	t of information Disagree	exchanged w Somewhat disagree	as sufficient. Neither agree nor disagree	Somewhat agree	Agree	Strongly agree
		0			0	
		0%		100%		
						Submit

Figure D-96. Screenshot of the Discussion Effectiveness Questions

THE UNIV OF QUEEN	/ERSITY ISLAND					
Communica	tion Clarity					
Select the op	tion that best re	presents you.				
1. How clear Not at all	were the messa Very unclear	ges of your pa Lowly clear	artner? Explain y Slightly clear	your answer. Moderately clear	Very clear	Highly clear
2. How often	were the messa	iges of your pa	artner unclear?	Explain your	answer.	
Never	Hardly ever	Rarely	Sometimes	Usually	Almost always	Always
0	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
		0%		100%		Submit

Figure D-97. Screenshot of the Communication Clarity Questions

UNIVERSITY QUEENSLAND FRALIA
We thank you for your time spent taking this survey. Your response has been recorded.
0%

Figure D-98. Screenshot of the End of the Experiment

APPENDIX E. ANALYSIS OF OUTLIERS





Figure E-1. Dependent Variable Boxplot

APPENDIX F. TEST OF ASSUMPTIONS

F.1 NORMALITY CHECK

	Group	Statistic	df	Sig.
Developed SU	Non-ontology Usage	0.81	16	0.00
	Ontology Usage	0.94	14	0.38
Improved SU	Non-ontology Usage	0.82	16	0.00
	Ontology Usage	0.91	14	0.16
Developed CU	Non-ontology Usage	0.88	16	0.04
	Ontology Usage	0.94	14	0.45
Communication Openness	Non-ontology Usage	0.92	16	0.15
	Ontology Usage	0.96	14	0.64
Discussion Efficiency	Non-ontology Usage	0.92	16	0.19
	Ontology Usage	0.93	14	0.34
Task Discussion Effectiveness	Non-ontology Usage	0.90	16	0.08
	Ontology Usage	0.95	14	0.53
Communication Clarity	Non-ontology Usage	0.91	16	0.11
	Ontology Usage	0.91	14	0.14
SU Improvement	Non-ontology Usage	0.90	16	0.69
	Ontology Usage	0.94	14	0.38

Table F-1. Shapiro-Wilk Test Results

	Experimental Group		Statistic	Std. Error	Z-score
Developed SU	Non-Ontology	Skewness	0.14	0.56	0.24
		Kurtosis	-1.83	1.09	-1.68
	Ontology	Skewness	-0.15	0.60	-0.25
		Kurtosis	-0.91	1.15	-0.79
Improved SU	Non-Ontology	Skewness	-0.83	0.56	-1.46
		Kurtosis	-0.85	1.09	-0.78
	Ontology	Skewness	0.00	0.60	0.00
		Kurtosis	-1.26	1.15	-1.09
Developed CU	Non-Ontology	Skewness	-0.75	0.56	-1.33
		Kurtosis	-0.75	1.09	-0.69
	Ontology	Skewness	-0.12	0.60	-0.20
		Kurtosis	-0.45	1.15	-0.39
Communication Openness	Non-Ontology	Skewness	-0.10	0.56	-0.17
		Kurtosis	-1.50	1.09	-1.37
	Ontology	Skewness	-0.60	0.60	-1.00
		Kurtosis	0.19	1.15	0.16
Discussion Efficiency	Non-Ontology	Skewness	0.32	0.56	0.57
		Kurtosis	-0.43	1.09	-0.39
	Ontology	Skewness	-0.43	0.60	-0.72
		Kurtosis	-0.88	1.15	-0.76
Task Discussion Effectiveness	Non-Ontology	Skewness	0.32	0.56	0.56
		Kurtosis	-1.37	1.09	-1.25
	Ontology	Skewness	0.06	0.60	0.10
		Kurtosis	-1.10	1.15	-0.95
Communication Clarity	Non-Ontology	Skewness	-0.96	0.56	-1.71
		Kurtosis	1.26	1.09	1.15
	Ontology	Skewness	-1.17	0.60	-1.96
		Kurtosis	1.74	1.15	1.51

Table F-2. Z-scores for Skewness and Kurtosis

F.2 LINEARITY ANALYSIS





Figure F-1. Linearity Analysis for Dependent Variables

APPENDIX G. MEDIATION ANALYSIS

G.1 MEDIATION OF SHARED UNDERSTANDING IMPROVEMENT: FOUR FACTORS

* * * * * * * * * * * * *	** PROCESS Pr	ocedure d	for SPSS Rele	ease 2.16 *	* * * * * * * * * * *	* * * * * *
Wri	itten by Andr	ew F. Hay	yes, Ph.D.	www.af	hayes.com	
Documenta	ation availab	le in Hay	yes (2013). w	ww.guilfor	d.com/p/hay	es3
<pre>************************************</pre>	**************************************	*****	*****	*****	****	****
M4 = CC						
Sample size 30						
* * * * * * * * * * * * *	* * * * * * * * * * * * *	******	* * * * * * * * * * * * *	******	* * * * * * * * * * *	* * * * * *
Outcome: CO						
Model Summary R .2535	R-sq .0643	MSE 1.1894	F 1.7776	df1 1.0000	df2 28.0000	p .1932
Model						
constant Group	coeff 4.8750 .5536	se .2728 .4152	t 17.8673 1.3333	p .0000 .1932	LLCI 4.3161 2970	ULCI 5.4339 1.4041
*****	* * * * * * * * * * * * *	******	* * * * * * * * * * * * *	*****	* * * * * * * * * * * *	* * * * * * *
Outcome: DE						
Model Summary	/		_	1.54	160	
.3847	R-sq .1480	MSE 1.2239	4.4686	1.0000	28.0000	р .0436
Model						
constant Group	coeff 4.5833 .8929	se .2713 .4224	t 16.8967 2.1139	p .0000 .0436	LLCI 4.0277 .0276	ULCI 5.1390 1.7581
***************** Outcome: TDE	* * * * * * * * * * * * *	******	* * * * * * * * * * * * *	*****	* * * * * * * * * * *	* * * * * *
Model Summary R .3105	R-sq .0964	MSE 1.3076	F 2.7697	df1 1.0000	df2 28.0000	p .1072
Model						
constant Group	coeff 4.3125 .7232	se .2897 .4346	t 14.8867 1.6642	p .0000 .1072	LLCI 3.7191 1670	ULCI 4.9059 1.6134
* * * * * * * * * * * * *	* * * * * * * * * * * * *	******	* * * * * * * * * * * * *	******	* * * * * * * * * * *	* * * * * *
Outcome: CC						
Model Summary R	/ R-sa	MSE	न	df1	df2	n
.3941	.1553	1.0217	4.7035	1.0000	28.0000	.0387
Model					TTOT	
constant Group	4.8750 .8393	se .2427 .3870	t 20.0890 2.1688	р .0000 .0387	4.3779 .0466	5.3721 1.6320

*******	* * * * * * * * *	* * * * * * *	* * * * * * * * * *	* * * * * * * * * * * *	********	********	* * * * * * *
Outcome:	SUImp						
Model Sum	mary						
	R	R-sq	MSE	F	df1	df2	р
.50	29 .	2529	3.0463	2.2431	5.0000	24.0000	.0828
Model							
	coef	f	se	+	n	LLCT	ULCT
constant	-1 319	7 2	8907	- 4565	6521	-7 2860	4 6466
CO	1.010	0	6053	0066	9978	-1 2453	1 2533
00	249	0	0129	.0000	7975	-1 6352	2 1220
DE	.240	9	.9120	.2720	. / 0 / J	-1.0352	2.1329
TDE	.564	6	.9981	. 5657	.5/69	-1.4954	2.6246
CC	159	0	.4858	3273	.7463	-1.1617	.8437
Group	.429	3	.8804	.4876	.6302	-1.3878	2.2464
*****	* * * * * * * * *	*****	* TOTAL E	FFECT MODEL	*******	*****	* * * * * * *
Outcome:	SUImp						
Model Sum	mary						
	R	R-sq	MSE	F	df1	df2	р
.25	65 .	0658	3.2653	1.8805	1.0000	28.0000	.1812
Model							
	coef	f	se	t	р	LLCI	ULCI
constant	1.500	0	.4989	3.0067	.0055	.4780	2.5220
Group	.928	6	.6771	1.3713	.1812	4585	2.3157
* * * * * * * * *	* * * * * * * *	TOTAL,	DIRECT, A	ND INDIRECT	EFFECTS **	* * * * * * * * * * * *	* * * * * *
Total eff	ect of X	on Y					
Effe	ct	SE	t	q	LLCI	ULCI	
. 92	86 .	6771	1.3713	.1812	4585	2.3157	
• • • •	••••	0,71	1.0/10		.1000	2.010	
Direct of	fect of V	on V					
DITECC EI	TECC OI A		+	â	TTCT	III CT	
EIIE	02	3E 0004	1070	c 202	1 2070	2 2464	
.42	93 .	8804	.48/0	.0302	-1.38/8	2.2404	
- 11 .							
Indirect	effect of	X on Y					
	Effect	Boot	SE Boot	LLCI Boott	JLCI		
TOTAL	.4993	.59	36	5329 1.8	3954		
CO	.0022	.43	52 -1.	.1172	695		
DE	.2222	.72	46 -1.	0604 1.9	9427		
TDE	.4083	.68	17	8181 1.6	5939		
CC	1335	.46	09 -1.	8098 .3	326		
* * * * * * * * *	* * * * * * * * *	** ANAL	YSIS NOTE	S AND WARNIN	IGS ******	* * * * * * * * * * * *	* * * * * * *
Number of 10000	bootstra	p sampl	es for bi	as corrected	l bootstra <u>r</u>	o confidence	intervals:
Level of 95.00	confidenc	e for a	ll confid	ence interva	als in out <u>r</u>	out:	

NOTE: All standard errors for continuous outcome models are based on the HC3 estimator

----- END MATRIX -----

G.2 MEDIATION OF SHARED UNDERSTANDING IMPROVEMENT: TWO FACTORS

* * * * * * * * * * * * *	*** PROCESS 1	Procedure f	for SPSS Rele	ease 2.16 *	*******	* * * * * * *
W1 Document	ritten by And tation availa	drew F. Hay able in Hay	yes, Ph.D. yes (2013).	www.af www.guilfor	Thayes.com cd.com/p/hay	res3
* * * * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * *
Model = 4 Y = SUIn X = Grou M1 = DE M2 = CC	np Ip					
Sample size 30						
**************************************	* * * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	*****	* * * * * *
Model Summar	CV					
.3847	R-sq .1480	MSE 1.2239	F 4.4686	df1 1.0000	df2 28.0000	p .0436
Model						
constant Group	coeff 4.5833 .8929	se .2713 .4224	t 16.8967 2.1139	p .0000 .0436	LLCI 4.0277 .0276	ULCI 5.1390 1.7581
**************** Outcome: CC	* * * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * *	*****	* * * * * *
Model Summan R .3941	ry R-sq .1553	MSE 1.0217	F 4.7035	df1 1.0000	df2 28.0000	p .0387
Model	coeff	50	+	n	LLCT	III.CT
constant Group	4.8750	.2427 .3870	20.0890 2.1688	.0000 .0387	4.3779	5.3721 1.6320
***************** Outcome: SUI	************ Imp	* * * * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * *	*****	* * * * * * *
Model Summan	сy					
R .4701	R-sq .2210	MSE 2.9323	F 2.0007	df1 3.0000	df2 26.0000	p .1386
Model						
constant DE CC Group	coeff -1.5025 .6737 0175 .3417	se 2.3453 .3590 .3821 .8198	t 6406 1.8769 0458 .4169	p .5274 .0718 .9638 .6802	LLCI -6.3235 0642 8030 -1.3434	ULCI 3.3186 1.4116 .7680 2.0269
*********	* * * * * * * * * * * * *	*** TOTAL E	EFFECT MODEL	*******	*****	* * * * * * *
Outcome: SUI	Imp					
Model Summar R .2565	ry R-sq .0658	MSE 3.2653	F 1.8805	df1 1.0000	df2 28.0000	p .1812
Model						
	coeff	se	t	p	LLCI	ULCI
constant Group	1.5000 .9286	.4989 .6771	1.3713	.0055 .1812	.4780 4585	2.5220 2.3157
* * * * * * * * * * * * *	***** TOTAL,	, DIRECT, A	AND INDIRECT	EFFECTS **	*****	* * * * * *
Total effect	c of X on Y					
Effect .9286	SE .6771	t 1.3713	р .1812	LLCI 4585	ULCI 2.3157	

Direct effect of X on Y

Eff∈	ect	SE	t	р	LLCI	ULCI	
.34	17 .	8198	.4169	.6802	-1.3434	2.0269	
Indirect	effect of	X on Y					
	Effect	Boot SE	BootLLCI	BootULC	I		
TOTAL	.5868	.4875	1654	1.854	4		
DE	.6015	.4059	.0221	1.670	13		
CC	0147	.3703	-1.0469	.512	6		
********** Number of 10000	bootstra	** ANALYSI p samples	S NOTES AND) WARNINGS) *******)ootstrap	confidence in	**** tervals:
Level of 95.00	confidence	e for all	confidence	intervals	; in outp	out:	
NOTE: All	standard	errors fo	r continuou	is outcome	models	are based on t	he HC3 estimator
EN	ID MATRIX						

G.3 MEDIATION OF DEVELOPED CROSS-UNDERSTANDING: FOUR FACTORS

****	* * * * * * * *	* PROCESS P	rocedure i	for SPSS Rel	ease 2.16 *	* * * * * * * * * * *	* * * * * * *
Do	Wri ocument <i>a</i>	tten by And tion availa	rew F. Hay ble in Hay	yes, Ph.D. yes (2013).	www.af www.guilfor	hayes.com d.com/p/hay	ves3
***** Model Y X M1 M2 M3 M4	******** = 4 = DCU = Group = CO = DE = TDE = CC	***********	*****	*****	******	*****	****
Sample	e size 30						
***** Outcor	******* ne: CO	*****	* * * * * * * * * *	* * * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * *	* * * * * *
Model	Summary R .2535	R-sq .0643	MSE 1.1894	F 1.7776	df1 1.0000	df2 28.0000	p .1932
Model consta Group	ant	coeff 4.8750 .5536	se .2728 .4152	t 17.8673 1.3333	p .0000 .1932	LLCI 4.3161 2970	ULCI 5.4339 1.4041
***** Outcor	******** ne: DE	*****	* * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * *
Model	Summary R .3847	R-sq .1480	MSE 1.2239	F 4.4686	df1 1.0000	df2 28.0000	p .0436
Model consta Group	ant	coeff 4.5833 .8929	se .2713 .4224	t 16.8967 2.1139	p .0000 .0436	LLCI 4.0277 .0276	ULCI 5.1390 1.7581
***** Outcor	******** me: TDE	******	* * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	*******	****
Model	Summary R .3105	R-sq .0964	MSE 1.3076	F 2.7697	df1 1.0000	df2 28.0000	p .1072
Model consta Group	ant	coeff 4.3125 .7232	se .2897 .4346	t 14.8867 1.6642	p .0000 .1072	LLCI 3.7191 1670	ULCI 4.9059 1.6134
*****	*******	********	* * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * *	* * * * * * *
Outcor	ne: CC						
Model	Summary R .3941	R-sq .1553	MSE 1.0217	F 4.7035	df1 1.0000	df2 28.0000	p .0387
Model consta Group	ant	coeff 4.8750 .8393	se .2427 .3870	t 20.0890 2.1688	р .0000 .0387	LLCI 4.3779 .0466	ULCI 5.3721 1.6320
***** Outcor	******** ne: DCU	* * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * *	****
Model	Summary	7		_	1.51	100	
	к .6186	к-sq .3826	MSE 7.5836	'± 2.8668	ail 5.0000	ai2 24.0000	р .0362

Model ModelcoeffsetpLLCIULCIconstant2.69212.9186.9224.3655-3.33188.7160CO1.12671.1684.9643.3445-1.28493.5382DE-.9352.8998-1.0394.3090-2.7924.9219TDE1.1125.95221.1683.2542-.85293.0779CC-.1555.8487-.1833.8561-1.90721.5961Group2.24271.34071.6728.1074-.52455.0098 Outcome: DCU Model Summary R-sq MSE F df1 df2 p .1854 8.5768 6.3071 1.0000 28.0000 .0181 R .4306 Model p LLCI .0000 6.1210 .0181 4001
 coeff
 se
 t
 p

 constant
 7.9375
 .8868
 8.9509
 .0000

 Group
 2.7054
 1.0772
 2.5114
 .0181
 ULCT 9.7540 .4987 4.9120 Total effect of X on Y
 Effect
 SE
 t
 p
 LLCI
 ULCI

 2.7054
 1.0772
 2.5114
 .0181
 .4987
 4.9120
 Direct effect of X on Y Effect SE t p LLCI ULCI 1.6728 .1074 -.5245 5.0098 1.3407 2.2427 Indirect effect of X on Y Effect Boot SE BootLLCI BootULCI .4627 .9009 -1.1527 2.5600 TOTAL 3.8536 CO .6237 1.0198 -.3991 .5980 DE -.8350 1.0944 -3.8054 1.1641 TDE .8046 -.6761 4.1691 1.3795 -2.3939 CC -.1305 .8805 Number of bootstrap samples for bias corrected bootstrap confidence intervals: 10000 Level of confidence for all confidence intervals in output: 95.00 NOTE: All standard errors for continuous outcome models are based on the HC3 estimator ----- END MATRIX -----

G.4 MEDIATION OF DEVELOPED CROSS-UNDERSTANDING: TWO FACTORS

*********	** PROCESS 1	Procedure f	or SPSS Rele	ease 2.16 *	******	* * * * * * *
Wr	itten by And	drew F. Hay	es, Ph.D.	www.af	hayes.com	
Documenta	ation availa	able in Hay	es (2013). T	www.gullior	cd.com/p/nay	ess
Model = 4 Y = DCU X = Group M1 = DE M2 = CC	**********	*****	*****	* * * * * * * * * * *	*****	*****
Sample size 30						
* * * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	******	* * * * * * *
Outcome: DE						
Model Summary R	R-sq	MSE	F	df1	df2	р
.3847	.1480	1.2239	4.4686	1.0000	28.0000	.0436
Model						
constant Group	coeff 4.5833 .8929	se .2713 .4224	t 16.8967 2.1139	p .0000 .0436	LLCI 4.0277 .0276	ULCI 5.1390 1.7581
*****	* * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	*****	* * * * * * *
Outcome: CC						
Model Summary R	R-sq	MSE	F	df1	df2	р
.3941	.1553	1.0217	4.7035	1.0000	28.0000	.0387
Model						
constant Group	coeff 4.8750 .8393	se .2427 .3870	t 20.0890 2.1688	р .0000 .0387	LLCI 4.3779 .0466	ULCI 5.3721 1.6320
* * * * * * * * * * * * * *	******	* * * * * * * * * * *	* * * * * * * * * * * *	* * * * * * * * * * *	*****	* * * * * * *
Outcome: DCU						
Model Summary	P-sq	MCF	F	d f 1	df2	n
.5375	.2889	8.0625	3.0944	3.0000	26.0000	.0443
Model						
constant	coeff 2.9204	se 3.3061	t .8833	р . 3851	LLCI -3.8755	ULCI 9.7164
DE	.7729	.4663	1.6576	.1094	1856	1.7313
СС	.3025	.7058	.4286	.6717	-1.1482	1.7533
Group	1./614	1.1832	1.4886	.1486	6709	4.1937
**************************************	* * * * * * * * * * * * *	*** TOTAL E	FFECT MODEL	* * * * * * * * * *	*******	* * * * * * *
000000000000000000000000000000000000000						
Model Summary R	R-sq	MSE	F	df1	df2	р
.4306	.1854	8.5768	6.3071	1.0000	28.0000	.0181
Model						
constant	coeff 7.9375	se .8868	t 8.9509	р .0000	LLCI 6.1210	ULCI 9.7540
Group	2.7054	1.0772	2.5114	.0181	.4987	4.9120
*****	**** TOTAL,	, DIRECT, A	ND INDIRECT	EFFECTS **	*****	* * * * * * *
Total effect	of X on Y					
Effect 2.7054	SE 1.0772	t 2.5114	p .0181	LLCI .4987	ULCI 4.9120	
Direct effect	c of X on Y					
Effect	SE	t	р	LLCI	ULCI	